

Title	IRREDUCIBLE DECOMPOSITIONS OF NON-TYPEI REPRESENTATIONS
Author(s)	河上, 哲
Citation	大阪大学, 1982, 博士論文
Version Type	VoR
URL	https://hdl.handle.net/11094/1666
rights	
Note	

The University of Osaka Institutional Knowledge Archive : OUKA

https://ir.library.osaka-u.ac.jp/

The University of Osaka

IRREDUCIBLE DECOMPOSITIONS OF NON-TYPEI REPRESENTATIONS

Satoshi KAWAKAMI

Faculty of Engineering Science Osaka University

1982

Errata Sheet

Page	Line	Error	Correction
iii	2	variaty	variety
iv	5	tecniques	techniques
iv	6	representatios	representations
iv	9	trasformation	transformation
iv	17	represeatations	representations
9	9	repreaentations	representations
13	13	indivisual	individual
15	4	$U^{\chi} = Ind_{K}^{G}\chi$	$U^{\chi} = Ind_{H}^{G}\chi$, $V^{\eta} = Ind_{K}^{G}\eta$
43	6	indivisual	individual
65	6	for for all	for all
75	12	There	These
78	17	H_{χ} -nvariant	H_{χ} -invariant
86	20	immedictely	immediately
86	26	Proposition 2.4.6.	Lemma 4.2.4.
134	15	(1976)	(1977)
134	16	certain	non-regular
135	1	variaty	variety

To my wife

Keiko

ACKNOWLEDGEMENT

It is my pleasure to thank Professors Masamichi Takesaki, Fujihiro Araki, and Hiroshi Takai for their warm encouragements and valuable suggestions.

Special thanks go to Professor Calvin C. Moore for his invitation and warm hospitality extended to me at University of California in Berkeley, and to Professors Nobuhiko Tatsuuma and Takeshi Hirai for their frequent and stimulating discussions at Kyoto University.

My deep gratitudes go to Doctor Yoshikazu Katayama and Mr. Tsuyoshi Kajiwara for their stimulating and fruitful discussions, and to Professor Shunsuke Funakoshi for his guidance and valuable suggestions on the present research.

But above all, my most sincere gratitude goes to Professor Osamu Takenouchi, my adviser, whose constant encouragements, enthusiasm for the present material, and instructive suggestions have been most invaluable.

Abstract

In this thesis, we study the variaty of irreducible decompositions of non-type I representations. It is known that irreducible decompositions into direct integrals of non-type I representations are not unique in general. Long before examples are known which demonstrate this fact and these are about regular representations of non-type I groups.

In this thesis, we show many other new examples of this non-uniqueness phenomenon of decompositions not only for regular representations but also for some factor representations and will make it clear that this phenomenon has some close relation with ergodic measures and cohomology groups of transformation groups. To do this, we investigate cohomology theory for double transformation groups and generalize the Mackey's theory of induced representations. Furthermore, we describe explicitly the interrelations between decompositions and maximal abelian subalgebras from a view point of operator algebras. As an application of our theory, we obtain some new families of irreducible representations of certain non-type I groups.

- iii -

Table of contents				
a la serie de la serie de La serie de la s La serie de la s	Pago			
Trtroduction	age			
Introduction				
Preliminaries and notations	9			
Chapter I. Decompositions of regular representations	11			
1.1. Fundamental tecniques				
by induced representatios	12			
1.2. Discrete groups	13			
1.3. Semi-direct product groups	20			
Chapter II. Cohomology of trasformation groups				
2.1. Elementary properties	28			
2.2. Double transformation groups	32			
2.3. Cohomology subordinate to measures	35			
2.4. Weak cohomology	39			
2.5. Examples and some calculations	43			
2.6. An application to decompositions				
of regular representations	52			
Chapter III. Generalized induced represeatations	56			
3.1. A generalization of induced representations	57			
3.2. Irreducible representations				
of semi-direct product groups	61			
3.3. Applications and examples	71			
Chapter IV. Decompositions				
of some factor representations	75			
4.1. Decompositions of π^{χ}	77			
4.2. Properties of π^{χ} and $U^{(\chi,A,\eta)}$	84			
4.3. Applications and examples	89			
- iv -				

Chapt	ter V	. Representations of C*-crossed products	93
	5.1.	Abelian subalgebras in $\pi^0(A)$ '	94
	5.2.	Decompositions of π^0	102
	5.3.	Further discussions	
		on the decompositions of π^0	110
	5.4.	Applications and examples	117
Bibl	iograp	phy	133
			· . 2

In this thesis, we study the variety of irreducible decompositions of non-type I representations. It is known that irreducible decompositions into direct integrals of non-type I representations are not unique in general. Long before, examples are known which demonstrate this fact, and these are about regular representations of non-type I groups. In this thesis, we show many other new examples of this nonuniqueness phenomenon of decompositions not only for regular representations but also for some factor representations. We then construct a theory which makes it clear that this phenomenon has some close relation with ergodic measures and cohomology groups of transformation groups.

This thesis consists of five chapters. In chapter I, under rather systematic versions, we arrange examples obtained before by several authors. In chapters II and III, we investigate cohomology theory for double transformation groups and generalize the Mackey's theory of induced representations. These will culminate in the description of our theory of irreducible decompositions of non-type I representations in chapter IV. In chapter V, we extend these results from a view point of operator algebras. Now we will explain briefly the contents of each chapter.

In chapter I, we study the phenomenon that regular representations of some non-type I groups may be decomposed into direct integrals of irreducible representations in completely different ways.

In 1951, H.Yoshizawa first pointed it out about the

- l -

free group on two generators [54]. In the same year, G.W. Mackey independently showed similar results about some discrete semi-direct product groups by applying the method of induced representations [29]. In 1974, M.Saito studied the classification of cojugacy classes of Cartan subgroups of $SL(2, \mathbb{Z})$ and found that the regular representation of $SL(2, \mathbb{Z})$ was possible to be decomposed into irreducible components in infinitely many completely differnt ways [46]. Furthermore, A.A.Kirillov reported in [28] that the phenomenon occurs even in the case of the Mautner group which is not discrete but a simply connected solvable Lie group.

In this chapter, we first introduce these four examples under rather systematic versions and offer new other examples. In section 1.1, we describe fundamental techniques which are used to give different decompositions of regular representations, by using Mackey's method of induced representations [30]. In section 1.2, we state the criteria of irreducibility and equivalency of monomial representations of discrete groups according to [29]. Using these results, we review the non-uniqueness phenomenon about individual discrete groups. In section 1.3, we generalize the results about the Mautner group obtained by A.A.Kirillov [28] and consider the occurrence of the phenomenon in search about semi-direct product groups, which include Mackey's examples in some sense [24].

In chapter II, we explain the notion of cohomology of transformation groups [20]. This notion appeared in the Mackey's works [33], [34], [35] and its study has been

- 2 -

developed by several authors. K.Schmidt has studied it related with ergodic theory [45]. Another way of the development was pursued by A.Guichardet [17] and C.C.Moore [37], [38] who considered this cohomology as the one cohomology of locally compact groups. Further, there is a way followed by G.W.Mackey and A.Ramsay. They have investigated it as a family of similarity classes of homomorphisms of a measure groupoid or a virtual group [34], [43], [44].

In section 2.1, we describe elementary properties of the cohomology of topological transformation groups. Propositions 2.1.1 and 2.1.2 are fundamental and may follow from the results in some of the works by C.C.Moore and A.Ramsay. However, we will add the proofs for completeness.

In section 2.2, we introduce double transformation groups and their cohomology. These replace certain nonsmooth topological transformation groups and their cohomology, and play a principal role in our considerations.

In section 2.3, we state the cohomology subordinate to measures. We often use this cohomology in later arguments.

In section 2.4, we study the notion of weak cohomology. This notion is important as an index showing the variety of decompositions of representations, which is one of our main subjects [27].

In section 2.5, subgroups of cohomology groups and weak cohomology groups are found in some concrete cases.

In section 2.6, we argue again decompositions of the regular representation of semi-direct product groups related with cohomology, as an application of this chapter [24].

In chapter III, we investigate generalized induced

- 3 -

representations for double transformation groups, related with cohomology and we construct families of non-Mackey representations of certain non-regular semi-direct product groups as a generalization of Mackey's method [20]. Applying this construction to the Mautner group, we obtain a new parametrized family of non-Mackey representations. The representations found by L.Baggett form a part of this family.

In 1978, L.Baggett found a family of non-Mackey irreducible representations of the Mautner group via the decompositions of a generalizd tensor product of some concrete representations [4]. In order to elucidate the mechanism of his family, we develop a theory of generalized induced representations in this chapter. In 1976, A.Ramsay turned the Mackey's theory into a representation theory of measure groupoids [44] and obtained a generalization of induced representations. Our notion is close to his but there are some differences. These differences will be seen to be crucial in the decomposition theory in later chapters. It is known that, for a connected and simply connected solvable Lie group G, there exists an algebraic solvable Lie group \widetilde{G} which contains G such that $[\tilde{G},\tilde{G}] = [G,G] = N$ and \tilde{G} acts on \hat{N} (the dual of N) smoothly. L. Pukanszky made an extensive use of this fact in [41], [42]. We impose a similar assumption (*) for nonregular semi-direct product groups, which will be used effectively as a substitute of the fact above-mentioned.

In section 3.1, for a double transformation group, we define unitary representations in relation to cohomology,

- 4 -

which appears as a generalization of the Mackey's induced representations [30], [33].

In section 3.2, following the construction in section 3.1, we have families of non-Mackey representations of nonregular semi-direct product groups satisfying our condition (*). In Theorem 3.2.6, we show when such representations are mutually equivalent, and in Theorem 3.2.7, we give a criterion of the irreducibility. In Proposition 3.2.9, we mention a property which characterizes these representations. The results obtained are akin to the results in [33] or [44] but ours are more precise according to the strong conditions imposed. Moreover, the techniques employed by L.Baggett [4] will be better understood from our points of view.

In section 3.3, we apply our general results to the discrete Mautner group and the Mautner group.

In chapter IV, we consider the irreducible decompositions of type II factor representations of some non-regular semi-direct product groups [21]. Taking a certain factor representation of such a group, we show it can be decomposed in many different ways into direct integrals of irreducible representations, while the diagonal algebras are spatially isomorphic with each other. The explicit form of the diagonal algebra is also given.

The theory of irreducible decompositions is based on the following general result of F.I. Mautner [36]. Let G be a locally compact group and π be a unitary representation of G on a separable Hilbert sapce $\frac{1}{2}$. Suppose that \Re is an abelian von Neumann subalgebra in $\pi(G)$ '. Then, there exists

- 5 -

a standard measure space (Y, ν) such that π is algebraically isomorphic with $L^{\infty}(Y, \nu)$ and π is decomposed as follows.

$$\pi \cong \int_{Y}^{\Phi} \pi^{\eta} d\nu(\eta) ,$$

Moreover, π^{η} is irreducible for ν -almost all $\eta \in Y$ if and only if Π is maximal abelian in $\pi(G)$ '.

In chapter IV, we consider the irreducible decompositions of type II factor representations of some non-regular semidirect product group. In Theorem 4.1.3, a certain representation π^{X} of such a group G will be decomposed in an explicit way to a direct integral of irreducible representations, each component having a definite form. The corresponding maximal abelian von Neumann subalgebra in $\pi(G)$ ' is also described in a concrete form.

It is known that the non-type I'ness of a locally compact group or a C^{*}-algebra is closely related to the nonsmoothness of topological transformation groups [12], [15], [16]. In non-smooth topological transformation groups, there are various kinds of quasi-orbits. Furthermore, the cohomology group for each non-transitive quasi-orbit seems to be huge, at least it is known to be non-trivial under some conditions [38]. The non-uniqueness of decompositions of a non-type I representation seems to depend deeply on these two facts. The results in [14] and [50] are certainly connected with the former, viz. the existence of various quasi-orbits and the examples in [28], [29], and section 1.3 also seem to be so intrincically. The present chapter is an attempt to describe the relation of the non-uniqueness of

- 6 -

decompositions with the latter phenomenon viz. the nontriviality of cohomology groups.

The decomposition in Theorem 4.1.3 is done by using a cocycle of the cohomology theory studied in chapter II, and it is shown in Proposition 4.2.1 that two decompositions are completely different when only the used cocycles are not weakly cohomologous, even the diagonal algebras are spatially isomorphic with each other. Thus we may get a large number of different decompositions of a given representation into irreducible components. To illustrate various possibilities, we will give two examples in section 4.3.

Chapter V is devoted to study representations of certain C^{*}-crossed products [26]. From a view point of operator algebras, we will extend the results investigated in the previous chapters.

For two closed subgroups H and K of a locally compact abelian group G, we get a C*-crossed product $A = C_0(G/H) \times_{\gamma} K$. We investigate decompositions of a certain representation π^0 of A. In section 5.1, we study two families of abelian von Neumann subalgebras $\{\pi^a\}$ and $\{\mathcal{B}^b\}$ in the commuting algebra $\pi^0(A)', \ a^a$ being associated with the automorphism α^a of $\pi^0(A)', \ where a \in Z(K; G; H), \ and \ a^b$ with β^b , $b \in Z(H^{\perp}; \hat{G}; K^{\perp})$. We will have also some necessary and sufficient conditions of the maximality of π^a and \mathfrak{B}^b in $\pi^0(A)'$. In section 5.2, we study decompositions of π^0 corresponding to \mathfrak{A}^a . To do this, we study generalized induced representations of \mathcal{C}^* crossed products following to chapter III. In section 5.3, using the concept of Heisenberg representations, we write

- 7 -

down decompositions of π^0 corresponding to the above σ^a and σ^b in explicit forms. In section 5.4, we show some examples and an application to unitary representations of a certain locally compact group.

In the thesis, a representation means a continuous unitary representation of a locally compact group or a bounded *-representation of a C*-algebra. For a representation π of a locally compact group G, π is called to be of type I (resp. type II, type III, and non-type I) if the von Neumann algebra generated by $\pi(G)$ is of type I (resp. type II, type III, and non-type I). Similar definitions are done for a representation of a C*-algebra.

Through the thesis, when we use a terminology of a locally compact group or a locally compact space, they are assumed to satisfy the second axiom of countability. Furthermore, a Hilbert space and a C*-algebra are also assumed to be separable.

We often treat some semi-direct product groups given as follows. Let N and K be locally compact abelian groups. K acts on N as an automorphism group of N and the action is denoted by $N \ni z \longrightarrow k \cdot z \in N$ for each $k \in K$. Let G be a locally compact group which is $N \times K$ as a topological space and whose multiplication is given by

 $(z, k)(z', k') = (z + k \cdot z', k + k')$

for z, $z' \in \mathbb{N}$ and k, $k' \in K$. This group G is called a semidirect product of N with K and denoted by $N \times_S K$. We note that G is a unimodular group. We identify the subgroup $\{(z,0); z \in \mathbb{N}\}$ of G with N and the subgroup $\{(0,k); k \in K\}$ of G with K. We often consider the topological transformation

- 9 -

group $(K; \hat{N})$, canonically obtained for $G = N \times_{s} K$, where \hat{N} is the dual group of N and the action of K on the space \hat{N} is given by, for each keK and XeN,

$$\langle z, k \cdot \chi \rangle = \langle k \cdot z, \chi \rangle$$

for zEN.

We denote each abelian group of integers, rational numbers, real numbers, and complex numbers by \mathbb{Z} , \mathbb{Q} , \mathbb{R} , and \mathfrak{C} respectively. The positve parts of \mathbb{Z} , \mathbb{Q} , and \mathbb{R} are written by \mathbb{N} , \mathbb{Q}^* , and \mathbb{R}^+ respectively. One-dimensional torus group, namely, the abelian group of complex numbers with absolute value 1 is denoted by \mathbb{T} .

Other terminologies and notations mainly follow from J.Dixmier's books [8],[9], G.W.Mackey's works [31],[35] and the monograph [2] by L.Auslander and C.C.Moore.

Chapter I. Decompositions of regular representations

In this chapter, we study the phenomenon that regular representations of some non-type I groups may be decomposed into direct integrals of irreducible representations in completely different ways.

In 1951, H.Yoshizawa first pointed it out about the free group on two generators [54]. In the same year, G.W. Mackey independently showed similar results about some discrete semi-direct product groups by applying his method of induced representations [29]. In 1974, M.Saito studied the classification of cojugacy classes of Cartan subgroups of $SL(2, \mathbb{Z})$ and found that the regular representation of $SL(2, \mathbb{Z})$ was possible to be decomposed into irreducible components in infinitely many completely differnt ways [46]. Furthermore, A.A.Kirillov reported in [28] that the phenomenon occurs even in the case of the Mautner group which is not discrete but a simply connected solvable Lie group.

In this chapter, we introduce these four examples under rather systematic versions and offer new other examples. In section 1.1, we describe fundamental techniques which are used to give different decompositions of regular representations, by using the Mackey's method of induced represenations [30]. In section 1.2, we state the criteria of irreducibility and equivalency of monomial representations of discrete groups according to [29]. Using these results, we review the non-uniqueness phenomenon about individual discrete groups. In section 1.3, we generalize the results

- 11 -

about the Mautner group obtained by A.A.Kirillov [28] and consider the phenomenon about semi-direct product groups, which also includes Mackey's examples in some sense.

1.1. Fundamental techniques by induced representations

Let G be a locally compact group. Take a closed abelian subgroup H of G and denote the dual group of H by \hat{H} . For a unitary character χ of H, we get a unitary representation U^{χ} of G by U^{χ} = $Ind_{H}^{G}\chi$ which is the induced representation of χ from H to G, developed by G.W.Mackey [29] [30]. Let λ be the right regular representation of G and 1 be the trivial representation of {e}. Then, by general considerations about induced representations, we see that

 $\lambda \cong \int_{\widehat{H}}^{\bigoplus} U^{\chi} d\mu(\chi)$

where μ is a Haar measure of $\widehat{H}.$ Indeed,

$$\begin{aligned} & \cong \operatorname{Ind}_{\{e\}}^{G} \\ & \cong \operatorname{Ind}_{H}^{G} \operatorname{Ind}_{\{e\}}^{H} \\ & \cong \operatorname{Ind}_{H}^{G} \int_{\widehat{H}}^{\bigoplus} \chi d\mu(\chi) \\ & \cong \int_{\widehat{H}}^{\bigoplus} \operatorname{Ind}_{H}^{G} \chi d\mu(\chi) \\ & = \int_{\widehat{H}}^{\bigoplus} U^{\chi} d\mu(\chi). \end{aligned}$$

(by stage theorem)

(by Fourier transform)

(by Theorem 10.1 in [30])

Next, Take another closed abelian subgroup K of G and denote by V^{η} the unitary representation of G given by $V^{\eta} =$ $\operatorname{Ind}_{K}^{G} \eta$ for $\eta \in \widehat{K}$. Then, by similar arguments, we see that $\lambda \cong \int_{\widehat{K}}^{\bigoplus} V^{\eta} d\nu(\eta)$ where v is a Haar measure of \hat{K} .

Therefore, for two closed abelian subgroups H and K of G, we get two decompositions of the right regular representation λ of G as

$$\lambda \cong \int_{\widehat{H}}^{\bigoplus} U^{\chi} d\mu(\chi) \cong \int_{\widehat{K}}^{\bigoplus} V^{\eta} d\nu(\eta).$$
 (1.1.1)

Here we must consider the following problems (A) and (B).

Problem

(A) Irreducibility: When the representations U^{χ} and v^{η} ($\chi \widehat{\epsilon H},~\eta \widehat{\epsilon K})$ are irreducible ?

(B) Inequivalency: Which are unitarily equivalent or inequivalent among U^{χ} and V^{η} ($\chi \in \hat{H}$, $\eta \in \hat{K}$)?

In the next sections, we study these problems (A) and (B) in indivisual cases and show that the formula (l.l.l) gives indeed different decompositions into irreducible representations under some situations imposed.

1.2. Discrete groups

For discrete groups, the results obtained by G.W.Mackey [29] are valuable as criteria for (A) and (B).

Proposition 1.2.1. (Theorem 6' and 7' in [29]) let G be a locally compact group.

(A) Let H be an open subgroup of G and U^{χ} denote the representation $\text{Ind}_{H}^{G_{\chi}}$ of G for a unitary character χ of H. Then U^{χ} is irreducible if and only if for each gth one of

____ 1 O

the following statements is true.

(i) $\chi \neq g \cdot \chi$ on $g^{-1}Hg \cap H$ where $(g \cdot \chi)(h) = \chi(ghg^{-1})$ for $h \in g^{-1}Hg \cap H$.

(ii) $[g^{-1}Hg : g^{-1}Hg\Lambda H] = \infty$ or $[H : g^{-1}Hg\Lambda H] = \infty$, where [:] means the index.

(B) Let H_i (i=1,2) be two open subgroups of G and U^{χ_i} be the representations $\operatorname{Ind}_{H_i}^G \chi_i$ of G for unitary characters χ_i of H_i (i=1,2). Assume that U^{χ_i} (i=1,2) are irreducible. Then U^{χ_i} is unitarily equivalent to U^{χ_2} if and only if there exists geG such that

(i) $\chi_2 = g_{\cdot\chi_1}$ on $g^{-1}H_1g\Lambda H_2$, (ii) $[g^{-1}H_1g : g^{-1}H_g\Lambda H_2] < \infty$ and $[H_2 : g^{-1}H_1g\Lambda H_2] < \infty$.

Example I (G.W.Mackey)

Let G be a semi-direct product group $N \times_S K$ where N and K are infinite "discrete" abelian-groups and K acts on N as an automorphism group of N.

Taking two closed subgroups N and K, we have two decompositions of the right regular representation λ of G according to (1.1.1) as follows.

$$\cong \int_{\widehat{N}}^{\bigoplus} U^{\chi} d\mu(\chi) \cong \int_{\widehat{K}}^{\bigoplus} V^{\eta} d\nu(\chi)$$
 (1.2.1)

where $U^{\chi} = \operatorname{Ind}_{N}^{G} \chi$ for $\chi \in \widehat{N}$, $V^{\eta} = \operatorname{Ind}_{K}^{G}$ for $\eta \in \widehat{K}$, and μ (resp. ν) is a Haar measure of \widehat{N} (resp. \widehat{K}).

G.W.Mackey considered the following condition(a) about the action of K on N.

Condition(a) All non-trivial orbits of N under the action of K are infinite.

- 14 -

Proposition 1.2.2. (Lemmal, 2, 3 in §3 of [29])

(A)(i) U^{χ} is irreducible if and only if $k \cdot \chi \neq \chi$ for $k \in K$ distinct from the unit 0.

(ii) Under the condition(a), V^{η} is irreducible for each $\eta \in K$.

(B)(i) For $\chi_1^{}$, $\chi_2^{} \in \mathbb{N}^{}$, $U^{\chi_1^{}}$ is unitarily equivalent to $U^{\chi_2^{}}$ if and only if there exists $k \in K$ such that $\chi_2^{}=k \cdot \chi_1^{}$.

(ii) Under the condition(a), for η_1 , $\eta_2 \in \mathbb{N}^{n_1}$, v^{η_1} is unitarily equivalent to v^{η_2} if and only if $\eta_1 = \eta_2$.

(iii) U^{χ} is never unitarily equivalent to V^{η} for arbitray choice of $\chi \in \hat{N}$ and $\eta \in \hat{K}$.

Proof. This follows directly from Proposition 1.2.1. For the detail, see [29]. [Q.E.D.]

Let Q^* denote the multiplicative abelian group of all positive rational numbers. Then Q^* acts on Q by the multiplication as an automorphism group of Q and we have a semidirect product group $G = Q \times_S Q^*$. This group G satisfy the condition(a) and the decompositions (1.2.1) means that the regular representation of G may be decomposed into irreducible parts in two entirely different ways.

Example II (H.Yoshizawa)

Let F_2 denote the free group on two generators a and b. Let us now choose as H the abelian subgroup of F_2 generated by a and as K the abelian subgroup of F_2 generated by b. Then, both H and K are isomorphic with \mathbb{Z} of integers so that their dual groups H and K are isomorphic with the onedimensional torus group \mathbb{T} . By general consideratios in

- 15 -

section 1.1, the right regular representation λ of F_2 is decomposed as

$$\lambda \cong \int_{\mathbb{T}}^{\oplus} U^{\chi} d\mu(\chi) \cong \int_{\mathbb{T}}^{\oplus} V^{\eta} d\nu(\eta)$$

where μ is the normalized Haar measure of \mathbb{T} and $U^{\chi} = \operatorname{Ind}_{K}^{G} \chi$ for $\chi \in \widehat{H}$ and $\eta \in \widehat{K}$ parametrized with each element of \mathbb{T} .

Proposition 1.2.3. ([54])

(A) U^{χ} and V^{η} are irreducible.

(B)(i) For $X_1, X_2 \in \hat{H}, U^{X_1}$ is unitarily equivalent to U^{X_2} if and only if $X_1 = X_2$.

(ii) For η_1 , $\eta_2 \in \hat{K}$, V^{η_1} is unitarily equivalent to V^{η_2} if and only if $\eta_1 = \eta_2$.

(iii) U^{χ} is never unitarily equivalent to V^{η} for arbitrary choice of $\chi \in \mathcal{H}$ and $\eta \in \mathcal{K}$.

Proof. We give the proof according to Proposition 1.2.1 different from H.Yoshizawa's method.

(A) For each $g \notin H$, $g^{-1}Hg \cap H = \{e\}$ so that $[H : g^{-1}Hg \cap H] = \infty$. Therefore, by (A) in Proposition 1.2.1, U^{χ} is irreducible. Similarly, V^{η} is irreducible.

(B) Suppose U^{χ_1} is unitarily equivalent to U^{χ_2} . Then, by (B) in Proposition 1.2.1, there should exist $g\epsilon G$ such that $[H : g^{-1}Hg\Lambda H] < \infty$ and $\chi_2 = g \cdot \chi_1$ on $g^{-1}Hg\Lambda H$. Since $g^{-1}Hg H = \{e\}$ for any $g \notin H$, the condition $[H : g^{-1}Hg\Lambda H] < \infty$ implies that $g\epsilon H$ and so $\chi_2 = \chi_1$ on H. The coverse is clear.

(ii) is shown similarly as (i).

(iii) follows immediately from the fact that $g^{-1}Hg\Lambda K = \{e\}$ for any geG. [Q.E.D.]

Therefore, we see that the regular representation of F_2 is also decomposed into irreducible parts in two entirely different ways.

Example III (M.Siato)

Let G be a locally compact group. M.Saito considered a family A of open subgroups of G satisfying the following condition (b).

Condition (b)

(i) For H_1 , $H_2 \in A$ and $g \in G$, if $[H_2 : g^{-1}H_1g \cap H_2] < \infty$, then $H_2 \subset g^{-1}H_1g$.

(ii) For $H \in A$ and $g \in G$, $g H g^{-1} \subset H$ yields $g \in N(H)$ where N(H) is the normalizer of H.

For a unitary character X of H in A, the representation $\operatorname{Ind}_{H}^{G}X$ of G is denoted by U^X. He stated the irreducibility and equivalency of such representations by his original method [46]. Here we show them by using Proposition 1.2.1 but under some restrictions.

Let A^+ denote the subfamily of satisfying that N(H)=H for $H \in A^+$. Then, we get the following.

Proposition 1.2.4.

(A) For $H \in A^+$, U^X is irreducible for every unitary character X of H.

(B) For unitary characters X_i of H_i in A^+ (i=1,2), U^{X_1} is unitarily equivalent to U^{X_2} if and only if there exists geG such that $H_2 = g^{-1}H_1g$ and $X_2 = g \cdot X_1$ on H_2 .

- 17 -

proof. (A) Suppose $[H : H\Lambda g^{-1}Hg] < \infty$ for geG. Then, by (i) of the condition(b), $H < g^{-1}Hg$ and so $gHg^{-1} < H$. By (ii) of the condition (b), g must be in N(H). Since N(H)=H, we get geH. Therefore, for any $g \notin H$, $[H : H\Lambda g^{-1}Hg] = \infty$ holds. This implies that U^{χ} is irreducible by (A) of Proposition 1.2.1.

(B) Suppose $U^{\chi_{i}}$ is unitarily equivalent to $U^{\chi_{2}}$. Then, by (B) of Proposition 1.2.1, there exists $g \in G$ such that (i) $\chi_{2} = g \cdot \chi_{1}$ on $g^{-1}H_{1}g \cap H_{2}$ and (ii) $[g^{-1}H_{2}g : g^{-1}H_{1}g \cap H_{2}] < \infty$ and $[H_{2}:g^{-1}H_{1}g \cap H_{2}] < \infty$. The first condition in (ii) is equivalent to the condition $[H_{2} : H_{1} \cap g H_{2}g^{-1}] < \infty$, which implies that $H_{1} < g H_{2}g^{-1}$ by (i) of the condition (b). Similarly, by the second condition in (ii), we get $H_{2} < g^{-1}H_{1}g$. Therfore, we see that $H_{2} = g^{-1}H_{1}g$. [Q.E.D.]

M.Saito [46] studied the conjugacy classes of Cartan subgroups of SL(2, \mathbb{Z}) and showed that the family of Cartan subgroups satisfies the condition (b). Furtheremore, he pointed out that a Cartan subgroup H satysfying N(H)=H is an abelian group isomorphic with $\mathbb{Z}^{x\{\pm 1\}}$ and that the conjugacy classes of such groups are infinite.

For such a group H, according to section 1.1, we get a decomposition of the regular representation λ of SL(2, Z) as

$$\lambda \cong \int_{\widehat{H}}^{\Phi} U^{\chi} d\mu(\chi).$$

Therefore, by Proposition 1.2.4, it is clear that the regular representation of $SL(2, \mathbb{Z})$ may be decomposed into irreducible constituents in infinitely many entirely different ways.

- 18 -

Example IV

Observing the above three examples, we can also give infinitely many entirely different decompositions of the regular representation of the free group F_2 on two generators as a generalization of Example II.

We consider a family \mathcal{B}^{\dagger} of subgroups H of F₂ satisfying that

(i) H is generated by one element of F_2 .

(ii) N(H) = H.

Then, we get the following.

Proposition 1.2.5. \mathcal{B}^+ satisfies the condition (b).

For the proof of Proposition 1.2.5, we need the following lemma.

Lemma 1.2.6. For g_1 , $g_2 \in F_2$, if $g_1^m = g_2^n$ for some $m, n \in \mathbb{Z}$, then there exists $k \in F_2$ and $m', n' \in \mathbb{Z}$ such that $g_1 = k^{m'}$ and $g_2 = k^{n'}$.

Proof. Suppose $g_1g_2 \neq g_2g_1$. Then, the equation $g_1^m = g_2^n$ would give a relation in F_2 , which contradicts with the freeness of F_2 . Therefore, g_1 and g_2 generate an abelian subgroup of F_2 . Since an abelian subgroup of F_2 is generated by one element, there exists $k \in F_2$ such that $g_1 = k^m$ and $g_2 = k^n$ ' for some m', $n' \in \mathbb{Z}$. [Q.E.D.]

Proof of Proposition 1.2.5.

At first we show that \mathcal{B}^+ satisfies (i) of the condition (b). For H_1 , $H_2 \in \mathcal{B}^+$ and $g \in F_2$, suppose $[H_2 : g^{-1}H_1g \wedge H_2] < \infty$. Then, $g^{-1}H_1g\Lambda H_2 \neq \{e\}$. Hence, there exist non-zero integers m, n such that $g^{-1}h_1^mg = h_2^n$ where h_1 are generators of $H_1(i=1,2)$. Therefore, we get $(g^{-1}h_1g)^m = h_2^n$. By Lemma 1.2.6, there exists $k \in F_2$ such that $g^{-1}h_1g = k^m'$ and $h_2 = k^m'$ for some m', $n' \in \mathbb{Z}$. By the condition $N(H_2) = H_2$, we see that $k = h_2$ or h_2^{-1} . Then, we get $g^{-1}H_1gcH_2$.

Next we check (ii) of the condition (b). For $H \in \mathcal{B}^+$ and $g \in F_2$, suppose $gHg^{-1}CH$. Let h be a generator of H. Then, $ghg^{-1} = h^n$ for some integer n. By the condition N(H) = H, n must be ±1. Hence, we see that $gHg^{-1} = H$, which implies $g \in N(H)$. [Q.E.D.]

Each subgroup H in \mathcal{B}^+ is an abelian group isomorphic with \mathbb{Z} and it is easily verified that the cojugacy classes of \mathcal{B}^+ are infinite. Therefore, by similar arguments as in Example III, we know that the regular representation of F_2 is decomposed into irreducible components in infinitely many entirely different ways.

1.3. Semi-direct product groups

Let G be a semi-direct product group $N \times_S K$ of N with K where N and K are both locally compact abelian groups and K acts on N as an automorphism group. In this section, we consider irreducible decompositions of the regular representation λ of G.

Applying the general consideration in section 1.1, we get

$$\lambda \cong \int_{\widehat{N}}^{\bigoplus} U^{\chi} d\mu(\chi) \cong \int_{\widehat{K}}^{\bigoplus} V^{\eta} d\nu(\eta)$$

where $U^{\chi} = \operatorname{Ind}_{N}^{G} \chi$ $(\chi \in \widehat{N}), V^{\eta} = \operatorname{Ind}_{K}^{G} \eta$ $(\eta \in \widehat{K}), \text{ and } \mu$ (resp. ν) is a Haar measure of \widehat{N} (resp. \widehat{K}). For the topological transformation group $(K; \widehat{N})$, let H_{χ} denote the stability group of K at $\chi \in \widehat{N}$. Then, we get the following criteria.

Proposition 1.3.1.

(A)(i) U^{χ} is irreducible if and only if $H_{\chi} = \{0\}$.

(ii) V^{η} is irreducible if and only if the Haar measure μ of N is ergodic under the action of K.

(B)(i) If μ is a non-transitive measure, U^{χ} is never unitarily equivalent to V^{η} for arbitrary choice of $\chi \in \mathbb{N}^{n}$ and $n \in \mathbb{K}$.

Proof. The proof will be given later in a more general situation (see Theorem 1.3.3). [Q.E.D.]

In the case that N and K are discrete abelian groups, we can give a sufficient condition of ergodicity of the measure μ on $\stackrel{\frown}{N}$ under K.

Lemma 1.3.2. Let N and K be discrete abelian groups. If the action of K on N satisfies the condition (a) in section 1.2, the Haar measure of $\overset{\frown}{N}$ is ergodic under the action of K.

Proof. Since N is a discrete abelian group, the dual \widehat{N} of N is a compact group. We denote by μ the normalized Haar measure of \widehat{N} . Let $\delta_{_{Z}}$ be a continuous function on \widehat{N}

- 21 -

given by $\delta_{z}(\chi) = \langle z, \chi \rangle (\chi \in \widehat{N})$ for each zen. Then, the family $\{\delta_{z}; z \in \mathbb{N}\}$ is a orthonormal basis of the Hilbert space $L^{2}(\widehat{N}, \mu)$ of all square integrable functions on \widehat{N} with respect to μ . Let $L^{\infty}(\widehat{N}, \mu)$ be the space of all essentially bounded measurable functions with respect to μ . For f in $L^{\infty}(\widehat{N}, \mu)$, we define the action of K by $(k \cdot f)(\chi) = f(k \cdot \chi)$.

Suppose that f in $L^{\infty}(\hat{N}, \mu)$ satisfies that k f = f (µ-a.a.). Then, we show that f = constant (µ-a.a.). This will establish the ergodicity of the measure µ under the action of K. Since µ is a finite measure on \hat{N} , f belongs to $L^{2}(\hat{N}, \mu)$ and k f = f in L^{2} -norm. Let the Fourier expansion of f be

$$f = \sum_{Z \in \mathbb{N}} a_Z \delta_Z$$
 in L²-norm

where

$$a_z \in \mathbb{C}$$
 and $\sum_{z=N} |a_z|^2 < \infty$ (1.3.1)

By simple calculations, we get, for each $k \in K$,

$$k \cdot f = \sum_{z \in N} a_{(-k) \cdot z} \delta_z$$
 in L^2 -norm.

By the assumption that $k \cdot f = f$ in L^2 -norm and the uniqueness of the coefficients of the Forier expansion, we see that for each $k \in K$,

$$a_z = a_{(-k) \cdot z}$$
 (1.3.2)

Suppose $a_z \neq 0$ for $z \neq 0$. Then, the equality (1.3.2) and the condition (a) stand in contraction with the fact (1.3.1). Therefore, a_z must be 0 for $z \neq 0$ and so we get

$$f = a_0 \delta_0$$
 in L^2 -norm,

- 22 -

which yields

$$f = a_0$$
 (µ-a.a.). [Q.E.D.]

By Lemma 1.3.2, we see that the criterion of irreducibility of V^{η} ($\eta \in \hat{K}$) in Proposition 1.3.1 covers the discrete cases described in section 1.2 and that Mackey's example $G = Q \times_{S} Q^{*}$ satisfies the conditions in Proposition 1.3.1.

Now, we consider more general cases that H_χ is not necessarily trivial or μ is not necessarily ergodic.

When $H_{\chi} \neq \{0\}$, for each $\omega \epsilon \hat{H}_{\chi}$, we put

$$L_{(z,h)}^{(\chi,\omega)} = \langle z, \chi \rangle \langle h, \omega \rangle$$

for $(z,h) \in N_{s} H_{\chi} = G_{\chi}$. Then, $L^{(\chi,\omega)}$ is a unitary character of $G_{\chi} = N_{s} H_{\chi}$ and we get a unitary representation $U^{(\chi,\omega)}$ of G by

$$U^{(\chi,\omega)} = \text{Ind } {}^{G}_{G_{\chi}}L^{(\chi,\omega)}.$$

By general considerations of induced representations [30], we see that U^{χ} is decomposed as

$$\mathbf{U}^{\chi} \cong \int_{\widehat{\mathbf{H}}_{\chi}}^{\oplus} \mathbf{U}^{(\chi,\omega)} d\tau_{\chi}(\omega)$$

where τ_{χ} is a Haar measure of \hat{H}_{χ} .

When μ is not ergodic under K, we decompose into ergodic measures as

$$\mu = \int_{Z}^{\bigoplus} \mu_{\zeta} \, d\sigma \, (\zeta)$$

- 23 -

where (Z,σ) is a standard measure space and μ_{ζ} ($\zeta \in Z$) are quasi-invariant ergodic measures on \hat{N} under the action of K. According to this decomposition of μ , we get a decompsition of the Hilbert space $L^2(\hat{N},\mu)$ as

$$L^{2}(\widehat{N},\mu) \cong \int_{Z}^{\bigoplus} L^{2}(\widehat{N},\mu_{\zeta})d\sigma(\zeta)$$

and a decomposition of the representation V^{η} of G as

$$v^{\eta} \cong \int_{Z}^{\bigoplus} v^{(\eta,\zeta)} d\sigma(\zeta).$$

We can give an explicit form of the component representation $V^{(n,\zeta)}$ on $L^2(\hat{N},\mu_{\zeta})$ as follows. Take a \mathbb{R}^+ -valued Borel function $\rho_{\zeta}(k,\chi)$ on $K \times \hat{N}$ satisfying for each keK,

$$\rho_{\zeta}(k,\chi) = \frac{d(k \cdot \mu_{\zeta})}{d\mu_{\zeta}}(\chi) \qquad (\mu_{\zeta}-a.a.\chi)$$

where $(k \cdot \mu_{\zeta})(E) = \mu_{\zeta}(k \cdot E)$ for each Borel set E in \hat{N} . Then, for each $\xi(\chi) \in L^2(\hat{N}, \mu_{\zeta})$,

$$(V_{(z,k)}^{(\eta,\zeta)}\xi)(\chi) = \rho_{\zeta}(k,\chi) \langle z,\chi \rangle \langle k,\eta \rangle \xi(k \cdot \chi)$$

for $(z,k) \in \mathbb{N}_{s}^{\times} \mathbb{K} = \mathbb{G}$.

Here, we get irreducible decompositions of the regular representation λ of a semi-direct product group G = N×_SK of locally compact abelian groups N and K as follows.

Theorem 1.3.3. The regular representation λ of G is decomposed as

$$\lambda \cong \int_{N}^{\oplus} \int_{\hat{H}_{\chi}}^{\oplus} U^{(\chi,\omega)} d\tau_{\chi}(\omega) d\mu(\chi)$$
$$\cong \int_{K}^{\oplus} \int_{Z}^{\oplus} V^{(\eta,\zeta)} d\sigma(\zeta) d\nu(\eta).$$

- 24 -

(A) $U^{(\chi,\omega)}$ and $V^{(\eta,\zeta)}$ are irreducible representations of G.

(B) When μ_{ζ} is a non-transitive measure, $V^{(n,\zeta)}$ is never unitarily equivalent to $U^{(\chi,\omega)}$ for arbitrary choice of $\eta \in \hat{K}$, $\chi \in \hat{N}$, and $\omega \in \hat{H}_{\chi}$.

Proof. The irreducibility of $U^{(\chi,\omega)}$ follows from general considerations of induced representations [30]. We show the irreducibility of $V^{(\eta,\zeta)}$. Suppose $WV_{(z,k)}^{(\eta,\zeta)} = V_{(z,k)}^{(\eta,\zeta)}W$ for some bounded operator W on $L^2(\hat{N}, \mu_{\zeta})$. Since the set of operators $V_{(z,0)}^{(\eta,\zeta)}$ (z \in N) generates the maximal abelian von Neumann algebra $L^{\infty}(\hat{N}, \mu_{\zeta})$ of all multiplication operators in $\mathcal{L}(L^2(\hat{N}, \mu_{\zeta}))$, the algebra of all bounded operators on $L^2(\hat{N}, \mu_{\zeta})$, the equality $WV_{(z,0)}^{(\eta,\zeta)} = V_{(z,0)}^{(\eta,\zeta)}W$ for each $z \in$ N implies that W must be a multiplication operator $\rho(f)$ ($f \in L^{\infty}(\hat{N}, \mu_{\zeta})$). By simple calculations, we see that

$$V(\eta,\zeta)^* \rho(f) V(\eta,\zeta) = \rho(k \cdot f)$$

where $(k \cdot f)(\chi) = f(k \cdot \chi)$. On the other hand, by the assumption, for each $k \in K$,

$$V_{(0,k)}^{(n,\zeta)*}\rho(f)V_{(0,k)}^{(n,\zeta)} = \rho(f).$$

Therfore, we get, for each $k \in K$,

$$k \cdot f = f$$
 (μ_{ζ} -a.a.).

Since μ_{χ} is an ergodic measure on \hat{N} , we see that

$$f = constant$$
 (μ_{ζ} -a.a.).

Hence W must be a constant operator on $L^2(\hat{N}, \mu_{\tau})$. This

implies that $V^{(n,\zeta)}$ is an irreducible representation of G. (B) Let $_{N} | U^{(\chi,\omega)}$ and $_{N} | V^{(n,\zeta)}$ denote the representations of the subgroup N of G, given by the restrictions to N of $U^{(\chi,\omega)}$ and $V^{(n,\zeta)}$ respectively. These representations of N are decomposed as follows.

$$\begin{array}{c} & | U^{(\chi,\omega)} \cong \int_{N}^{\bigoplus} \gamma \, d\nu_{\chi}(\gamma) , \\ & \\ & N | V^{(\eta,\zeta)} \cong \int_{N}^{\bigoplus} \gamma d\mu_{\zeta}(\gamma) , \end{array}$$

where ν_{χ} is the canonical transitive quasi-invariant measure concentrated on $\operatorname{Orb}_{K}\chi$ on \widehat{N} . By the assumption that the measure μ_{ζ} is non-transitive, μ_{ζ} is never equivalent to ν_{χ} so that $\sqrt{V^{(\eta,\zeta)}}$ is never unitarily equivalent to $\sqrt{U^{(\chi,\omega)}}$ for any $\eta\in\widehat{K}$, $\chi\in\widehat{N}$, and $\omega\in\widehat{H}_{\chi}$ [32]. Therfore, we get the desired conclusion. [Q.E.D.]

Remark 1.3.4. The Mautner group is given as a semidirect product group $C^2 \times_{s} R$ of two dimensional vector group C^2 over C with R. Applying our result to this group, we get the example obtained by A.A.Kirillov ([28]). Furthermore, our result is applicable to the discrete Mautner group, the discrete Heisenberg group, and the Dixmier group.

Remark 1.3.5. We can give other irreducible decompositions of the regular representation of $G = N \times_S K$ different from those in Theorem 1.3.3. These are given with related to cohomology groups, which will be described later (see section 2.6).

- 26 -

Chapter II. Cohomology of transformation groups

In this chapter, we explain the notion of cohomology of transformation groups. This notion appeared in the Mackey's works [33], [34], [35] and its study has been developed by several authors. K.Schmidt has studied it related with ergodic theory [45]. Another way of the development was pursued by A.Guichardet [17] and C.C.Moore [37], [38] who considered this cohomology as the one cohomology of locally compact groups. Further, there is a way followed by G.W.Mackey and A.Ramsay. They have investigated it as a family of similarity classes of homomorphisms of a measure groupoid or a virtual group [34], [43], [44].

In section 2.1, we describe elementary properties of cohomology of topological transformation groups. Propositions 2.1.1 and 2.1.2 are fundamental and may follow from the results in some of the works by C.C.Moore and A.Ramsay. However, we add the proofs for completeness.

In section 2.2, we introduce double transformation groups and their cohomology. These replace certain nonsmooth topological transformation groups and their cohomology, and play a principal role in our considerations.

In section 2.3, we state the cohomology subordinate to measures. We often use this cohomology in later arguments.

In section 2.4, we study the notion of weak cohomology. This notion is important as an index showing the variety of decompositions of representations, which is one of our main subjects. In section 2.5, subgroups of cohomology groups and weak cohomology groups are found in some concrete cases. It is known that $H^{C}_{\mu}(\mathbb{Z};\mathbb{T})$ and $H^{C}_{\nu}(\mathbb{R};\mathbb{T}^{2})$ are isomorphic as cohomology groups as a general statement in [43]. We give here a concrete imbedding of $H^{C}_{\mu}(\mathbb{Z};\mathbb{T})$ to $H^{C}_{\nu}(\mathbb{R};\mathbb{T}^{2})$. This makes us possible to get some concrete cocycles easily.

In section 2.6, we argue again decompositions of the regular representation of a semi-direct product group related with cohomology, as an application of this chapter.

2.1. Elementary properties

Let (G;X) be a topological transformation group. The action of the group G on the topological space X is denoted by $(g,x) \longrightarrow g \cdot x$, where $x \longrightarrow g \cdot x$ is a homeomorphism of X, and we suppose it satisfies $g_2 \cdot (g_1 \cdot x) = (g_1 g_2) \cdot x$. Let Ω be a von Neumann algebra on a separable Hilbert space. Then we can define the cohomology of (G;X) as follows (see [28], [35], [45]).

Let \mathfrak{A}^{u} denote the set of unitary oparators of \mathfrak{A} equipped with the Borel structure generated by the weak oparator topology. G and X have the canonical Borel structures induced by their topologies. A Borel function C of G×X into \mathfrak{A}^{u} is said to be an \mathfrak{A}^{u} -valued cocycle, if it satisfies the condition

 $C(g_1g_2,x) = C(g_1,x)C(g_2,g_1,x)$ for all g_1 , $g_2 \in G$ and all $x \in X$.

- 28 -
Such cocycles C_1 and C_2 are said to be cohomologous when there exists an σ^u -valued Borel function A on X such that

 $C_2(g,x) = A(x)^* C_1(g,x)A(g \cdot x)$ for all geG and all xeX.

It is clear that the above relation of "cohomologous" is an equivalence relation. If a cocycle C is cohomologous to the one which equals identically the identity oparator of \mathfrak{A} , C is said to be an \mathfrak{A}^{u} -valued coboundary. We denote the set of all \mathfrak{A}^{u} -valued cocycles of (G;X) by $Z^{\mathfrak{A}}(G;X)$, and the set of all \mathfrak{A}^{u} -valued coboundaries of (G;X) by $B^{\mathfrak{A}}(G;X)$. Moreover, let us denote $H^{\mathfrak{A}}(G;X)$ the quotient set of $Z^{\mathfrak{A}}(G;X)$ by the above equivalence relation. This is called the \mathfrak{A}^{u} -valued cohomology set of (G;X), so is $C_{1}C_{2}$ (pointwise product) as well as C_{1}^{-1} , so that $Z^{\mathfrak{A}}(G;X)$ has an abelian group structure and moreover $B^{\mathfrak{A}}(G;X)$ is a subgroup of $Z^{\mathfrak{A}}(G;X)$. In this case, $H^{\mathfrak{A}}(G;X)$ is regarded as the quotient group of $Z^{\mathfrak{A}}(G;X)$ by $B^{\mathfrak{A}}(G;X)$, and it is called an \mathfrak{A}^{u} -valued cohomology group of (G;X).

A topological transformation group (G;X) is said to be smooth if every orbit is locally closed in X (see [12]. [16]), and effective if each stability group is trivial.

Proposition 2.1.1. If a topological transformation group (G;X) is smooth and effective, then $Z^{\Omega}(G;X) = B^{\Omega}(G;X)$ i.e. $H^{\Omega}(G;X)$ is trivial.

Proof. Since (G;X) is smooth, there exists a Borel cross section c from the orbit space G\X to X (see [12]).

- 29 -

Then the map $\Psi : (g,y) \longrightarrow g \cdot c(y)$ from $G \times (G \setminus X)$ to X is Borel and bijective as (G;X) is effective. By the smoothness of (G;X), $G \setminus X$ is a standard space and so is $G \times (G \setminus X)$. Hence the inverse of Ψ is Borel. Thus we have a Borel map a from X to G and a Borel map b from X to a cross section of X under G such that $\Psi^{-1}(x) = (a(x), b(x))$ i.e. $x = a(x) \cdot b(x)$ for each $x \in X$. For an arbitrary cocycle C(g,x), we put

A(x) = C(a(x),b(x)) for $x \in X$.

Then A is an σ^{u} -valued Borel function on X satisfying

$$C(g,x) = A(x)^{*}A(g \cdot x)$$

for all geG and all xeX. (2.1.1)

Indeed, observing g.x in two ways, i.e.

$$g.x = a(g.x) \cdot b(g.x) = a(g.x) \cdot b(x)$$

 $g.x = g(a(x) \cdot b(x)) = (a(x)g) \cdot b(x)$

we have, by the fact that (G;X) is effective,

$$a(g \cdot x) = a(x)g$$
 and $b(g \cdot x) = b(x)$.

Therefore, we get

This implies (2.1.1), so that C is a coboundary of (G;X).

[Q.E.D.]

When (G;X) is not effective, it is hard to get general results. Here we only mention the following proposition.

Proposition 2.1.2. If (G;X) is transitive and \mathcal{A} is onedimensional, then the cohomology group $\mathrm{H}^{\mathfrak{A}}(G;X)$ is isomorphic with X(G₀) which is the group of all continuous unitary characters of some stability group G₀.

Proof. Fix an $x_0 \in X$ and let G_0 be the stability group of G at x_0 . The map $G \ni g \longrightarrow g \cdot x_0$ gives a Borel isomorphism from $G_0 \setminus G$ onto X. For each cocycle C(g, x),

 $C(g_1g_2,x_0) = C(g_1,x_0)C(g_2,x_0)$ for all $g_1,g_2 \in G_0$

holds. Then the map $g \longrightarrow C(g, x_0)$ from G_0 to **T** is Borel homomorphic, therefore continuous, where **T** is the onedimensional torus and equals π^u . So there exists a continuous unitary character χ_C of G_0 such that

$$C(g,x_0) = \chi_C(g)$$
 for all $g \in G_0$.

In this way we get a map Ψ : $C \longrightarrow X_C$ from $Z^{\mathbb{C}}(G;X)$ to $X(G_0)$. It is verified with no trouble that this Ψ is homomorphic and surjective. Rest to show is that $Ker\Psi = B^{\mathbb{C}}(G;X)$. Suppose that

$$C(g_0, x_0) = 1$$
 for all $g_0 \in G_0$.

Then we get

$$C(g_0g,x_0) = C(g,x_0)$$
 for all $g_0\in G_0$ and all $g\in G$,

and it follows from this that there exists a Borel function

- 31 -

A on X such that

$$C(g,x_0) = A(g \cdot x_0)$$
 for all geG.

Since for each $x \in X$ there exists $g_1 \in G$ such that $x = g_1 \cdot x_0$, we have

$$C(g,x) = C(g,g_{1} \cdot x_{0}) = C(g_{1},x_{0})^{*}C(g_{1}g,x_{0}) = A(g_{1} \cdot x_{0})^{*}A(g \cdot (g_{1} \cdot x_{0})) = A(x)^{*}A(g \cdot x_{0}).$$

This implies that C is a coboundary of (G;X) and $\text{Ker}\Psi \subset B^{\mathbb{C}}(G;X)$. As the inverse inclusion relation is clear, we obtain that $\text{Ker}\Psi = B^{\mathbb{C}}(G;X)$. [Q.E.D.]

2.2. Double transformation groups

When (G;X) is not smooth, instead of considering (G;X), we take a double transformation group (G;Y;H) such that (G;Y) is smooth and (G;X) can be looked at as the same with (G;Y/H) as topological transformation groups.

Definition 2.2.1. We call (G;X;H) a double transformation group if groups G and H act on the same space X as topological transformation groups, where the actions of G and H on X are denoted by $(g,x) \longrightarrow g \cdot x$ and $(h,x) \longrightarrow x \cdot h$, and the following conditions are satisfied.

(1) $(g \cdot x) \cdot h = g \cdot (x \cdot h)$ for all $g \in G$, all $h \in H$, and all $x \in X$,

- 32 -

(2) the map: $(g,x,h) \longrightarrow g \cdot x \cdot h$ from $G \times X \times H$ to X is continuous.

Given a double transformation group (G;X;H) and a von Neumann algebra Ω , we will define an $\mathcal{O}_{-}^{\text{ll}}$ -valued cocycle, coboundary, and cohomology of (G;X;H) as follows.

Definition 2.2.2. We call an \mathfrak{A}^u -valued Borel function A on X an \mathfrak{A}^u -valued cocycle of (G;X;H) if

 $A(g x h) = A(g \cdot x)A(x)^{*}A(x \cdot h)$ for all g(G, all h(H, and all x(X)) (2.2.1)

is satisfied. Such cocycles A_1 and A_2 are said to be cohomologous if there exist an H-invariant cocycle B_1 and a G-invariant cocycle B_2 such that

$$A_2(x) = B_1(x)A_1(x)B_2(x)$$
 for all x $\in X$. (2.2.2)

If a cocycle A is cohomologous to the one which equals identically the identity operator of \Im , we say that A is an \Im^{u} -valued coboundary of (G;X;H).

We denote the set of cocycles by $Z^{\mathcal{R}}(G;X;H)$, the set of cohomology classes by $H^{\mathcal{R}}(G;X;H)$, and the set of coboundaries by $B^{\mathcal{R}}(G;X;H)$. If \mathcal{R} is abelian, they all have abelian group structures and $H^{\mathcal{R}}(G;X;H) \cong Z^{\mathcal{R}}(G;X;H)/B^{\mathcal{R}}(G;X;H)$.

Proposition 2.2.3. Let (G;X;H) be a double transformation group and Π be an abelian von Neumann algebra. (G;X) and (H;X) are supposed to be smooth and effective. Then the

- 33 -

following three abelian groups are isomorphic to each other.

- (1) H⁰¹(G;X;H)
 - (2) H⁷¹(G;X/H)
 - (3) H⁰⁷(H;GXX)

Proof. In general, the orbit spaces X/H and G\X may not be Hausdorff. However, $H^{\Omega}(G;X/H)$ and $H^{\Omega}(H;G\backslashX)$ are well-defined because the definition of a cohomology group depends only on the Borel structure, and we remark that the Borel structures of X/H and G\X induced by their topologies coincide with the quotient Borel structures by the smoothness of (H;X) and (G;X) (see [12]).

Let A be an \mathfrak{A}^{u} -valued cocycle of (G;X;H). Using this A, we define C and D by

 $C(g,x) = A(x)A(g\cdot x)^{*} \text{ for g} \in G \text{ and } x \in X, \quad (2.2.3)$ $D(h,x) = A(x)^{*}A(x \cdot h) \text{ for h} \in H \text{ and } x \in X. \quad (2.2.4)$

Then, the equality (2.2.1) implies that C(g,x) is H-invariant with respect to the variable $x \in X$ and D(h,x) is Ginvariant with respect to $x \in X$. Hence we may regard C as a cocycle of (G;X/H) and D as a cocycle of $(H;G\setminus X)$ because the cocycle conditions about C and D follow immediately from their definitions. The corespondences $A \longrightarrow C$ and $A \longrightarrow D$ induce the isomorphism from $H^{n}(G;X;H)$ onto $H^{n}(G;X/H)$ and from $H^{n}(G;X;H)$ onto $H^{n}(H;G\setminus X)$. In fact, let ψ be the map $A \longrightarrow C$ from $Z^{n}(G;X;H)$ to $Z^{n}(G;X/H)$. Then it is not hard to see that ψ is homomorphic and $\psi^{-1}(B^{n}(G;X/H)) = B^{n}(G;X;H)$. Moreover, Proposition 2.1.1 together with the assumptions imply that ψ is surjective.

Remark 2.3.4. Let \widetilde{G} be a group, and G and H be closed subgroups of \widetilde{G} . In this case, we define the actions of G and H on \widetilde{G} by

$$G \times \widetilde{G} \times H \ni (g, x, h) \longrightarrow h^{-1} \cdot x \cdot g \in \widetilde{G}.$$

Then, $(G; \tilde{G}; H)$ is a double trasformation group satisfying the assumptions of Proposition 2.2.3.

2.3. Cohomology subordinate to measures

When a measure is put on the space X, we shall consider the cohomology groups of (G;X) and (G;X;H) subordinate to this measure. Let μ be a positive Radon measure on a topological space X, and α be a von Neumann algebra. In the case of a topological transformation group (G;X), we shall change the former definitions as follows.

Let C_1 and C_2 be in $Z^{\Pi}(G;X)$. Then we say that C_1 is µ-cohomologous to C_2 if there exists an \mathfrak{N}^{u} -valued Borel function A on X such that for each geG

 $C_2(g,x) = A(x)^* C_1(g,x)A(g \cdot x)$ for μ -a.a. $x \in X$.

We denote the set of all μ -cohomology classes of $Z^{\mathcal{R}}(G;X)$ by $H^{\mathcal{R}}_{\mu}(G;X)$. A cocycle C is said to be a μ -coboundary if C is μ cohomologous to the one which equals identically the identity operator of \mathcal{R} , and we denote the set of all μ -coboundaries of (G;X) by $B^{\mathcal{R}}_{\mu}(G;X)$.

Next, in the case of a double transformation group

- 35 -

(G;X;H), for A_1 and A_2 in $Z^{(l)}(G;X;H)$, we say that A_1 is μ cohomologous to A_2 if there exist an H-invariant cocycle B_1 and a G-invariant cocycle B_2 such that

$$A_{2}(x) = B_{1}(x)A_{1}(x)B_{2}(x)$$
 for μ -a.a. $x \in X$.

The set of all μ -cohomology classes of $Z^{\Omega}(G;X;H)$ is denoted by $H^{\Omega}_{\mu}(G;X;H)$. A μ -coboundary is defined in the same way as above, and we denote the set of all μ -coboundaries of (G;X;H) by $B^{\Omega}_{\mu}(G;X;H)$.

Assuming \Re abelian, Z^{\Re} , B^{\Re}_{μ} , and H^{\Re}_{μ} have abelian group structures and we have $H^{\Re}_{\mu} \stackrel{\sim}{=} Z^{\Re}/B^{\Re}_{\mu}$ as groups in either case.

Note that the above definitions depend only on the measure class $C(\mu)$ of μ and not on μ itself. Therefore we write sometimes $B_{C(\mu)}^{\Omega}$ and $H_{C(\mu)}^{\Omega}$. A measure class $C(\mu)$ is said to be a quasi-orbit if μ is quasi-invariant and ergodic under the action of G on X.

Proposition 2.3.1. If a topological transformation group (G;X) is smooth and C(μ) is a quasi-orbit on X, then $H^{\mathbb{C}}_{C(\mu)}(G;X)$ is isomorphic to X(G₀) as abelian groups where G₀ is some closed subgroup of G.

Proof. By the smoothness of (G;X), $C(\mu)$ must be a transitive quasi-orbit. Therefore there exists an $x_0 \in X$ such that $C(\mu)$ is concentrated on the orbit $G \cdot x_0$ which is isomorphic to $G_0 G$ as topological transformation groups where G_0 is the stability group of G at x_0 , and such $C(\mu)$ is the unique measure class which corresponds to the canonical class on $G_0 \cap G$.

Let C be in $B^{\mathbb{C}}_{C(\mu)}(G;X)$. Then there exists a Borel function A on X such that for each $g\in G$

$$C(g,x) = A(x)^{*}A(g.x)$$
 for μ -a.a. $x \in X$. (2.3.1)

Now we define cocycles C_1 and C_2 by

$$C_{1}(g,x) = A(x)^{*}A(g \cdot x)$$
 (2.3.2)

$$C_2(g,x) = C_1(g,x)^* C(g,x)$$
 for geG and xeX. (2.3.3)

Then, we have, for each $g\in G$

$$C_2(g,x) = 1$$
 for µ-a.a. $x \in X$. (2.3.4)

Suppose that there exists a $g_0 \in G_0$ such that

$$C_2(g_0, x_0) \neq 1.$$
 (2.3.5)

The cocycle condition implies that

$$C_2(g_2,g_1,x_0) = C_2(g_1,x_0)^* C_2(g_1g_2,x_0)$$
 for all $g_1,g_2 \in G$.

If we define a Borel function B on G by $B(g) = C_2(g,x_0)$, we have

$$C_2(g_2,g_1 \cdot x_0) = B(g_1)^{B}(g_1g_2)$$

for all $g_1,g_2 \in G.$ (2.3.6)

According to (2.3.4), for each $g_2 \in G$

$$B(g_1) = B(g_1g_2)$$
 for a.a. $g_1 \in G$

holds because $C(\mu)$ may be considered as the canonical class on $G_0 \$ G. By Fubini's Theorem, we get, for almost all $g_1 \in G$

$$B(g_1) = B(g_1g_2)$$
 for a.a. $g_2 \in G$.

Therefore we have

 $B(g) = K \text{ (constant) for a.a. } g \in G. \qquad (2.3.7)$ However, by (2.3.6)

 $B(g_0g) = B(g_0)B(g)$ for all $g\in G$

holds and $B(g_0) \neq 1$ by (2.3.5), so that we have

$$B(g) \neq K$$
 for a.a. $g \in G$.

This fact contradicts with (2.3.7). So we get

$$C_2(g, x_0) = 1$$
 for all $g \in G_0$.

We have already shown in the proof of Proposition 2.1.2 that this fact implies $C_2 \in B^{\mathbb{C}}(G;X)$. Since C_1 is in $B^{\mathbb{C}}(G;X)$ and $B^{\mathbb{C}}(G;X)$ is an abelian group, we get $C \in B^{\mathbb{C}}(G;X)$ by (2.3.3) and so $B_{C(\mu)}^{\mathbb{C}}(G;X) \subset B^{\mathbb{C}}(G;X)$ has been shown. It is clear that the inverse inclusion relation holds, so that we have $B_{C(\mu)}^{\mathbb{C}}(G;X) = B^{\mathbb{C}}(G;X)$. Hence we get $H_{C(\mu)}^{\mathbb{C}}(G;X) = H^{\mathbb{C}}(G;X)$ = $X(G_0)$ by Proposition 2.1.2. [Q.E.D.]

Next we shall consider the case where (G;X) is not necessarily smooth. For a quasi-orbit $C(\mu)$ on X, we often find a large group \tilde{G} and its closed subgroup H such that $C(\mu)$ can be identified with the \tilde{G} -quasi-invariant measure class on $H\setminus\tilde{G}$. In this case we have the following theorem.

Theorem 2.3.2. Let \mathfrak{A} be an abelian von Neumann algebra and let (G; \tilde{G} ;H) be a double transformation group where G and H are closed subgroup of a group \tilde{G} and their actions are defined as in Remark 2.2.4. If σ is a Haar measure of \tilde{G} and μ (resp. ν) is a canonical quasi-invariant measure on $H \setminus \tilde{G}$ (resp. \tilde{G}/G), then the following three abelian groups are isomorphic to each other.

(1) H^π_σ(G;G;G;H)
 (2) H^π_μ(G;H\G)
 (3) H^π_ν(H;G/G)

Proof. This follows from Proposition 2.2.3 combined with some measure theoretic arguments. We omit the details.

[Q.E.D.]

2.4. Weak cohomology

Let (G;X) be a topological transformation group and \mathcal{O} be an abelian von Neumann algebra. Then, we define \mathfrak{O}^{u} -valued weak cohomology of (G;X) as follows.

Definition 2.4.1 For two \mathfrak{A}^u -valued cocycles C_1 and C_2 of (G;X), we call that C_1 is weakly cohomologous to C_2 if $C_1C_2^*$ is cohomologous to some continuous homomorphism from G to \mathfrak{A}^u . We denote all \mathfrak{A}^u -valued weakly cohomologous classes of (G;X) by $\widetilde{H}^{\mathfrak{A}}(G;X)$, which has also an abelian group structure. We call $\widetilde{H}^{\mathfrak{A}}(G;X)$ \mathfrak{A}^u -valued weak cohomology group.

Let $Z_0^{\mathfrak{A}}(G;X)$ denote all continuous homomorphisms from G to \mathfrak{A}^u and $H_0^{\mathfrak{A}}(G;X)$ be the factor group of $Z_0^{\mathfrak{A}}(G;X)$ by $Z_0^{\mathfrak{A}}(G;X) \wedge B^{\mathfrak{A}}(G;X)$. Then, we see that

 $\widetilde{H}^{\mathfrak{A}}(G; X) = H^{\mathfrak{A}}(G; X) / H_0^{\mathfrak{A}}(G; X).$

When a positive Radon measure μ is put on X, we can also define \mathfrak{A}^{u}_{μ} -valued weak μ -cohomology group $\widetilde{H}^{\mathfrak{A}}_{\mu}(G;X)$ of (G;X) by routine arguments as in section 2.3.

Then, we get immediately the following propositions, according to Proposition 2.1.2 and Proposition 2.3.1.

Proposition 2.4.2. Let (G;X) be a transitive topological transformation group where G is supposed to be a loclly compact abelian group. Then, $\tilde{H}^{\mathbb{C}}(G;X)$ is trivial.

Proposition 2.4.3. Let (G;X) be a smooth topological transformation group where G is supposed to be abelian. Then, for any quasi-orbit $C(\mu)$ on X, $\tilde{H}^{\mathbb{C}}_{C(\mu)}(G;X)$ is trivial.

Now, let G be a locally compact group. Taking two closed subgroups H and K of G, we consider a double transformation group (K;G;H). Let Ω be an abelian von Neumann algebra. Then we can also define α^u -valued weak cohomology group $\widetilde{H}^{\Omega}(K;G;H)$ of (K;G;H) as follows.

Dfinition 2.4.4. For two \Re^{u} -valued cocycles A_{1} and A_{2} of (K;G;H), A_{1} is called to be weakly cohomologous to A_{2} if $A_{1}A_{2}^{*}$ is σ -cohomologous to some continuous homomorphism from G to \Re^{u} where σ is a Haar measure of G. We denote all \Re^{u} -valued weakly cohomologous classes by $\widetilde{H}^{\alpha}(K;G;H)$ and we call it \Re^{u} -valued weak cohomology group of (K;G;H).

Proposition 2.4.5. Let G be a locally compact abelian group and μ (resp. ν) denote a Haar measure of G/H (resp. G/K). then, the following three abelian groups are isomorphic with each other.

- (1) H^o(K;G;H)
- (2) H (K;G/H)
- (3) $\tilde{H}_{v}(h;G/K)$

Proof. This follows from the definition and Theorem [Q.E.D.]

Our considerations go on in the situation that G is abelian and \mathfrak{A} is one-dimensional. Let $\widehat{\mathsf{G}}$ be the dual group of G and H[⊥] (resp. K[⊥]) denote the annihilator of the subgroup H (resp. K) of G in $\widehat{\mathsf{G}}$. We denote by H^C₀(K;G;H) the factor group of $\widehat{\mathsf{G}}$ by $\widehat{\mathsf{GAB}}^{\mathbb{C}}_{\mathfrak{G}}(K;G;H)$. Then, we see that

$$\widetilde{H}^{\mathbb{C}}(K;G;H) \cong H^{\mathbb{C}}_{\sigma}(K;G;H)/H^{\mathbb{C}}_{\Omega}(K;G;H).$$

Furthermore, we get the following.

Proposition 2.4.6. If K + H is dense in G,

 $H_0^{\mathbb{C}}(K;G;H) \stackrel{\sim}{=} \frac{1}{G}/(K^{\perp}+H^{\perp}).$

Proof. If $\chi \in G$ is written as $\chi(t) = \chi_1(t) \chi_2(t)$ for some $\chi_1 \in K^{\perp}$ and $\chi_2 \in H^{\perp}$, it is clear that χ is a coboundary by definition. So we shall show the converse.

Suppose that for $\chi \in \hat{G}$, $\chi(t) = E(t)F(t)$ for almost all teG, where E is an H-invariant cocycle and F is an K-invariant

- 41 -

cocycle. Since X satisfies $\chi(t_1+t_2)\overline{\chi(t_1)}\overline{\chi(t_2)} = 1$ for all $(t_1, t_2)\in G^2$, we get

$$E(t_1+t_2)\overline{E(t_1)}\overline{E(t_2)} = \overline{F(t_1+t_2)}F(t_1)F(t_2)$$

for almost all $(t_1, t_2) \in G^2$. Put

$$\phi(t_1, t_2) = E(t_1 + t_2) \overline{E(t_1)} \overline{E(t_2)}$$

for $(t_1, t_2) \in G^2$. Then, by the property of E and F, ϕ is $(K+H)^2$ -invariant. Since K+H is dense in G, $(K+H)^2$ is also dense in G^2 . Hence, $(K+H)^2$ acts on G^2 ergodically. Therefore, we get

$$\phi(t_1, t_2) = c$$
 (constant) for a.a. $(t_1, t_2) \in G^2$.

When we put E' = cE and F' = $\overline{c}F$, we see that

$$E'(t_{1}+t_{2}) = E'(t_{1})E'(t_{2}),$$

$$F'(t_{1}+t_{2}) = F'(t_{1})F'(t_{2}) \text{ for a.a. } (t_{1},t_{2}) \in$$

 G^2

and

$$X(t) = E'(t) F'(t)$$
 for a.a. teg.

By Theorem 5.1 in [41], there exists χ_1 , $\chi_2 \in G$ such that $\chi_1(t)$ = E'(t) and $\chi_2(t)$ = F'(t) for a.a. teg. Moreover, by the continuity of χ_1 and χ_2 , χ_1 must be H-invariant (i.e. $\chi_1 \in H$), χ_2 must be K-invariant (i.e. $\chi_2 \in K^{\perp}$), and $\chi(t) = \chi_1(t)\chi_2(t)$ for all teg. Therefore, χ must be in $K^{\perp} + H^{\perp}$. [Q.E.D.]

The proof of this lemma was suggested by Professor M. Takesaki.

Remark 2.4.7. Assume that $K^{\perp} + H^{\perp}$ is not closed in \hat{G} . Then, the induced Borel structure of $\hat{G}/(K^{\perp} + H^{\perp})$ is not standard so that the cardinal number of $H_0^{\mathbb{C}}(K;G;H)$ must be uncountable infinity. Therefore, in such a case, we can conclude that the cardinal number of $H_{\sigma}^{\mathbb{C}}(K;G;H)$ is also uncountable infinity. In some indivisual cases, we know that the cardinal number of $\widetilde{H}^{\mathbb{C}}(K;G;H)$ is uncountable infinity (see [27]) but general considerations about the weak cohomology group have not yet been obtained.

2.5. Examples and some calculations

Here we shall treat the following two transformation groups.

(a) $(\mathbf{Z};\mathbf{T})$ where \mathbf{Z} is the additve group of integers and \mathbf{T} is the one-dimensional torus. The action of \mathbf{Z} on \mathbf{T} is defined by

 $n \cdot \xi = e^{in} \xi$ for $n \in \mathbb{Z}$ and $\xi \in \mathbb{T}$.

(b) $(R;T^2)$ where R is the additive group of real numbers and T^2 is the two-dimensional torus. The action of R on T^2 is defined by

 $t \cdot (\xi, \eta) = (e^{it}\xi, e^{2\pi it}\eta)$ for $t \in \mathbb{R}$ and $(\xi, \eta) \in \mathbb{T}^2$.

Now we find the following double transformation groups corresponding to the cases (a) and (b).

(a-1) (Z;R;2 π Z). The actions of Z and 2 π Z on R are defined by

- 43 -

 $n \cdot z = z + n$ for $n \in \mathbb{Z}$ and $z \in \mathbb{R}$,

 $z \cdot (2\pi m) = z + 2\pi m$ for $2\pi m \epsilon 2\pi Z$ and $z \epsilon R$. (Z;T) $\stackrel{\checkmark}{=} (Z;R/2\pi Z)$ as topological transformation groups. (b-1) $(R;R^2;(2\pi Z)^2)$. The actions of R and $(2\pi Z)^2$ on R^2 are defined by

> $t \cdot (x,y) = (x+t,y+2\pi t)$ for $t \in \mathbb{R}$ and $(x,y) \in \mathbb{R}^2$, $(x,y) \cdot (2\pi m, 2\pi n) = (x+2\pi m, y+2\pi n)$ for $(2\pi m, 2\pi n) \in (2\pi Z)^2$ and $(x,y) \in \mathbb{R}^2$.

 $(\mathbf{R};\mathbf{T}^2) \cong (\mathbf{R};\mathbf{R}^2;(2\pi \mathbf{Z})^2)$ as topological transformation groups. Let μ,ν,α , and β be Haar measures of $\mathbf{T}, \mathbf{T}^2, \mathbf{R}$, and \mathbf{R}^2 respectively. According to Theorem 2.3.2, we get

$$(a-2) H_{\mu}^{\mathbb{C}}(\mathbb{Z};\mathbb{T}) \stackrel{\simeq}{=} H_{\alpha}^{\mathbb{C}}(\mathbb{Z};\mathbb{R};2\pi\mathbb{Z}),$$
$$(b-2) H_{\nu}^{\mathbb{C}}(\mathbb{R};\mathbb{T}^{2}) \stackrel{\simeq}{=} H_{\beta}^{\mathbb{C}}(\mathbb{R};\mathbb{R}^{2};(2\pi\mathbb{Z})^{2}).$$

We shall determine a part of these cohomology groups. Let us define the abelian group Z^0 and its subgroup B^0_{μ} by

 $\begin{aligned} z^{0} &= \left\{ \text{all } \mathbf{T} \text{-vlued Borel functions on } \mathbf{T} \right\} \\ B^{0}_{\mu} &= \left\{ b(\xi) \in Z^{0}; \text{ there exists an } a(\xi) \in Z^{0} \text{ such that} \right. \\ &\quad b(\xi) &= a(\xi)^{*} a(e^{i\xi}) \quad \text{ for } \mu \text{-almost all } \xi \in \mathbf{T} \right\} \end{aligned}$

We denote the quotient group $\text{Z}^0/\text{B}^0_\mu$ by $\text{H}^0_\mu.$

Lemma 2.5.1. ${\tt H}^{C}_{\mu}({\bf Z};T)$ is isomorphic with ${\tt H}^{0}_{\mu}$ as abelian groups.

Proof. For $C(n,\xi) \in Z^{\mathbb{C}}(\mathbb{Z};\mathbb{T})$, we put $a(\xi) = C(1,\xi) \in Z^{\mathbb{C}}$. Then we have a map $\psi : C \longrightarrow$ a from $Z^{\mathbb{C}}(\mathbb{Z};\mathbb{T})$ to $Z^{\mathbb{O}}$. It is easily checked that this ψ is an injective homomorphism and $\psi(B^{\mathbb{C}}_{\mu}(\mathbf{Z};\mathbf{T}))\mathcal{C}B^{0}_{\mu}$. We show that ψ is surjective and $\psi(B^{\mathbb{C}}_{\mu}(\mathbf{Z};\mathbf{T})) = B^{0}_{\mu}$. Let $a \in \mathbb{Z}^{0}$, we construct $C(n,\xi)$ as follows.

$$C(n,\xi) = \begin{cases} \frac{n-1}{k=0} a(e^{ik}\xi) & \text{if } n \ge 1 \\ 1 & \text{if } n = 0 \\ \frac{-(n+1)}{\prod_{k=0}^{k} a(e^{i(n+k)}\xi)} & \text{if } n \le -1. \end{cases}$$

Then $C(n,\xi)$ is in $Z^{\mathbb{C}}(\mathbf{Z};\mathbf{T})$ and $\psi(C) = a$, so ψ is surjective. If b is in B⁰, then this construction gives a B(n, ξ) in $B^{\mathbb{C}}_{\mu}(\mathbf{Z};\mathbf{T})$. Indeed, if $b(\xi) = a(\xi)^* a(e^{i\xi})$ for μ -almost all $\xi \in \mathbf{T}$, then, for example, when $n \ge 1$ we see that

$$B(n,\xi) = \frac{n-1}{\prod} b(e^{ik}\xi)$$

= $\frac{n-1}{\prod} a(e^{ik}\xi)^* a(e^{i(k+1)}\xi)$
= $a(\xi)^* a(e^{in}\xi)$ for μ -a.a. $\xi \in \mathbf{T}$.

Therefore ψ induces an isomorphism ψ_{*} from $H^{\mathbb{C}}_{\mu}(\mathbb{Z};\mathbb{T})$ onto $H^{\mathbb{O}}_{\mu}$.

We also take the following abelian groups $Z^1, \ B^1_{\alpha},$ and H^1_{α} defined by

$$\begin{split} \textbf{Z}^1 &= \big\{ \text{all \mathbf{T}-valued Borel functions on \mathbf{R} with period 2π} \big\} , \\ \textbf{B}^1_\alpha &= \big\{ \textbf{a}(z) \textbf{\textit{e}} \textbf{\textit{Z}}^1 \text{ ; there exists } \textbf{b}(z) \textbf{\textit{e}} \textbf{\textit{Z}}^1 \text{ such that} \\ &\quad \textbf{a}(z) \,= \, \textbf{b}(z)^* \textbf{b}(z+1) \quad \text{for α-almost all $z \textbf{\textit{e}} \textbf{R}$} \big\} , \end{split}$$

 $H^{l}_{\alpha} = Z^{l}/B^{l}_{\alpha}.$

Then it is clear that $Z^0 \cong Z^1$, $B^0_{\mu} \cong B^1_{\alpha}$, and $H^0_{\mu} \cong H^1_{\alpha}$.

- 45 -

Lemma 2.5.2. $(\mathbb{R}/\mathbb{Z}+2\pi\mathbb{Z})+\mathbb{Z}$ is a subgroup of $H^{\mathbb{C}}_{\mu}(\mathbb{Z};\mathbb{T})$. Proof. For $\lambda \in \mathbb{R}$ and $d \in \mathbb{Z}$, we define $a^{(\lambda,d)} \in \mathbb{Z}^1$ by

$$a^{(\lambda,d)}(z) = e^{i(dz+\lambda)}$$
 for zer.

The set of all $a^{(\lambda,d)}$ with $\lambda \in \mathbb{R}$ and $d \in \mathbb{Z}$ forms a subgroup of Z^{l} and we will determine which ones fall in B^{l}_{α} among these $a^{(\lambda,d)}$, s.

Suppose that $a^{(\lambda,d)} \in B^{l}_{\alpha}$. Then it implies that there exists a $b(z) \in Z^{l}$ such that

$$b(z)a^{(\lambda,d)}(z) = b(z+1)$$
 for $\alpha-a.a. z \in \mathbb{R}$. (2.5.1)

For Borel functions f(z) on R with period 2π , we adopt L^2 -norm $\|\cdot\|_2$ defined by

$$\|f\|_{2}^{2} = \int_{0}^{2\pi} |f(z)|^{2} dz.$$

Then, (2.5.1) is equivalent to

$$b(z)a^{(\lambda,d)}(z) = b(z+1)$$
 in L²-norm. (2.5.2)

We have the Fourier expansion of b(z).

$$b(z) = \sum_{n \in \mathbb{Z}} b_n e^{inz}$$
 in L²-norm. (2.5.3)

Then, by (2.5.2) we get

$$\sum_{n \in \mathbb{Z}} b_n e^{i\lambda} e^{i(n+d)z} = \sum_{n \in \mathbb{Z}} b_n e^{in} e^{inz} \text{ in } L^2 \text{-norm. (2.5.4)}$$

By the uniqueness of the coefficients of a Fourier expansion,

$$b_n e^{i\lambda} = b_{n+d} e^{i(n+d)}$$
 for all neZ. (2.5.5)

Hence we have

- 46 -

 $|b_n| = |b_{n+d}|$ for all $n \in \mathbb{Z}$. (2.5.6)

On the other hand,

$$\sum_{n \in Z} |b_n|^2 = ||b||_2^2 < \infty.$$

If $d \neq 0$, this fact contradicts (2.5.6). Therefore d must be 0, and in this case, by (2.5.5), we get

$$b_n e^{i\lambda} = b_n e^{in}$$
.

By the fact $\sum_{n \in \mathbb{Z}} |b_n|^2 \neq 0$, there exists an $n_0 \in \mathbb{Z}$ such that $b_{n_0} \neq 0$. Hence we have $e^{i\lambda} = e^{in_0}$, which implies that $\lambda \in \mathbb{Z} + 2\pi \mathbb{Z}$. Conversely, we can check easily that $a^{(\lambda,d)} \in B^1_{\alpha}$ if d = 0 and $\lambda \in \mathbb{Z} + 2\pi \mathbb{Z}$. [Q.E.D.]

Here we note that using this lemma we have the family of representations which L.Baggett has got in [4], and the argument in the above is parallel with his in some sense. But the next proposition will give rise to an essentially new parametrized family of irreducible representations of the Mautner group. Let Q be the additive group of rational numbers.

Proposition 2.5.3. $(\mathbb{R}/\mathbb{Z}+2\pi\mathbb{Z})+\mathbb{Q}$ is a subgroup of $H_{\mu}^{\mathbb{C}}(\mathbb{Z};\mathbb{T})$.

Proof. For $p \in \mathbb{N}$ (positive integers), $q \in \mathbb{Z}$, and $\lambda \in \mathbb{R}$, we define $\tilde{a}^{(\lambda,q/p)}$ by

$$\hat{a}^{(\lambda,q/p)}(z) = e^{i((q/p)z+\lambda)} \quad \text{for } 0 \leq z < 2\pi.$$

Then we are able to extend it to $a^{(\lambda,q/p)} \epsilon z^1$. By definition,

- 47 -

$(a^{(\lambda,q/p)})^{p} = a^{(p\lambda,q)}.$

If $a^{(\lambda,q/p)} \in B^{1}_{\alpha}$, then $(a^{(\lambda,q/p)})^{p}$ is also in B^{1}_{α} , because B^{1}_{α} is a group, so that $a^{(p\lambda,q)} \in B^{1}_{\alpha}$. This fact implies q = 0. (see the proof of Lemma 2.5.2). In the same way, we have $\lambda \in \mathbb{Z} + 2\pi \mathbb{Z}$. The converse is trivial. [Q.E.D.]

There exists a relation between ${\tt H}^{C}_{\mu}({\bf Z};{\bf T})$ and ${\tt H}^{C}_{\nu}({\bf R};{\bf T}^2)$, shown in the next lemma, so that we can get an information about ${\tt H}^{C}_{\nu}({\bf R};{\bf T}^2)$ from that of ${\tt H}^{C}_{\mu}({\bf Z};{\bf T})$.

Lemma 2.5.4. $H^{\mathbb{C}}_{\mu}(\mathbf{Z};\mathbf{T})$ can be imbedded isomorphically into $H^{\mathbb{C}}_{\nu}(\mathbf{R};\mathbf{T}^2)$ as an abelian group.

Proof. Instead of $H^{\mathbb{C}}_{\mu}(\mathbf{Z};\mathbf{T})$ and $H^{\mathbb{C}}_{\nu}(\mathbb{R};\mathbf{T}^2)$, we take $H^{\mathbb{C}}_{\alpha}(\mathbf{Z};\mathbb{R};2^{\pi}\mathbf{Z})$ and $H^{\mathbb{C}}_{\beta}(\mathbb{R};\mathbb{R}^2;(2\pi\mathbf{Z})^2)$ (see (a-2) and (b-2)). For $A(z)\in \mathbb{Z}^{\mathbb{C}}(\mathbf{Z};\mathbb{R};2^{\pi}\mathbf{Z})$, we define a **T**-valued Borel function $\widetilde{A}(x,y)$ on \mathbb{R}^2 by

 $\hat{A}(x,y) = A(x - \frac{1}{2\pi}\overline{y})$

where for $y \in \mathbb{R}$, $y = \overline{y} + [y]$, $0 \leq \overline{y} < 2\pi$, and $[y] \in 2\pi \mathbb{Z}$. Then we have $A(x,y) \in \mathbb{Z}^{\mathbb{C}}(\mathbb{R};\mathbb{R}^{2};(2\pi \mathbb{Z})^{2})$. In fact, for $t \in \mathbb{R}$, $(x,y) \in \mathbb{R}^{2}$, and $(2\pi m, 2\pi n) \in (2\pi \mathbb{Z})^{2}$,

 $\widetilde{A}(t \cdot (x, y) \cdot (2\pi m, 2\pi n))$ $= \widetilde{A}((x+2\pi m+t, y+2\pi n+2\pi t))$ $= A(x+2\pi m+t-\frac{1}{2\pi}\overline{y+2\pi n+2\pi t})$ $= A(x+2\pi m+t-\frac{1}{2\pi}(y+2\pi t)+\frac{1}{2\pi}[y+2\pi y])$ $= A(\frac{1}{2\pi}(-[y]+[y+2\pi t])+(x-\frac{1}{2\pi}\overline{y})+2\pi m))$

- 48 -

$$= A((x - \frac{1}{2\pi}y) + \frac{1}{2\pi}(-[y] + [y + 2\pi t]))A(x - \frac{1}{2\pi}y)^*A(x - \frac{1}{2\pi}y + 2\pi m)$$

$$= A(x + t - \frac{1}{2\pi}y + 2\pi t)A(x - \frac{1}{2\pi}y)^*A(x + 2\pi m - \frac{1}{2\pi}y + 2\pi m)$$

$$= \widetilde{A}(x + t, y + 2\pi t)\widetilde{A}(x, y)^*\widetilde{A}(x + 2\pi m, y + 2\pi n)$$

$$= \widetilde{A}(t \cdot (x, y))\widetilde{A}(x, y)^*\widetilde{A}((x, y) \cdot (2\pi m, 2\pi n)).$$

Therefore a map ψ : A $\longrightarrow \widehat{A}$ from $Z^{\mathbb{C}}(\mathbf{Z};\mathbf{R};2\pi\mathbf{Z})$ into $Z^{\mathbb{C}}(\mathbf{Z};\mathbf{R};(2\pi\mathbf{Z})^2)$ is obtained and it is easily verified that this map is an injective homomorphism.

Suppose that A_1 is cohomologous to A_2 , in other words, there exists a Z-invariant cocycle B and a $2\pi Z$ -invariant cocycle C such that

$$A_2(z)A_1(z)^* = B(z)C(z)$$
 for all $z \in \mathbb{R}$.

We will see that $\tilde{B}(x,y)$ is R-invariant and $\tilde{C}(x,y)$ is $(2\pi Z)^2$ -invariant and moreover

$$\widetilde{A}_{2}(x,y)\widetilde{A}_{1}(x,y)^{*} = \widetilde{B}(x,y)\widetilde{C}(x,y)$$
 for all $(x,y)\in \mathbb{R}^{2}$,

which imply that A_1 is cohomologous to A_2 . Indeed, for $t \in \mathbb{R}$ and $(x,y) \in \mathbb{R}^2$, we get

> > - 49 -

and for $(2\pi m, 2\pi n) \epsilon (2\pi Z)^2$ and $(x, y) \epsilon \mathbb{R}^2$,

$$\widetilde{C}((x,y) \cdot (2\pi m, 2\pi n)) = \widetilde{C}(x+2\pi m, y+2\pi n) = C(x+2\pi m - \frac{1}{2\pi}\overline{y+2\pi n}) = C(x-\frac{1}{2\pi}\overline{y}+2\pi m) = C(x-\frac{1}{2\pi}\overline{y}) = \widetilde{C}(x,y).$$

Conversely, suppose that \widetilde{A}_1 is cohomologous to \widetilde{A}_2 . Let B'(x,y) be an R-invariant cocycle of $(R; R^2; (2\pi Z)^2)$ and C'(x,y) be a $(2\pi Z)^2$ -invariant one, satisfying

$$\widetilde{A}_{2}(x,y)\widetilde{A}_{1}(x,y)^{*} = B'(x,y)C'(x,y)$$
 for all $(x,y) \in \mathbb{R}^{2}$.

We define cocycle B(z) and C(z) of $(\mathbf{Z};\mathbf{R};2\pi\mathbf{Z})$ by

$$B(z) = B'(z,0)$$
 and $C(z) = C'(z,0)$ for all $z \in \mathbb{R}$.

Then we have

$$A_{2}(z)A_{1}(z)^{*} = \widetilde{A}_{2}(z,0)\widetilde{A}_{1}(z,0)^{*}$$
$$= B(z)C(z) \quad \text{for all } z \in \mathbb{R},$$

where B(z) is **Z**-invariant and C(z) is $2\pi Z$ -invariant. In fact, observing that B'(x,y) is $2\pi Z$ -invariant with respect to the second variable, as A_1 , A_2 , and C' are so, we have

$$B(z+n) = B'(z+n,0)$$

= B'(z,-2\pi n)
= B'(z,0)

- 50 -

= B(z)

and likely

 $C(z+2\pi m) = C(z)$ for all $z \in \mathbb{R}$.

Therefore A_1 is cohomologous to A_2 .

At last, we shall show that A_1 is α -cohomologous to A_2 if and only if \widetilde{A}_1 is β -cohomologous to \widetilde{A}_2 . To this end, by the fact shown above, it is sufficient to show that A(z) = 1for α -almost all $z \in \mathbb{R}$ if and only if $\widetilde{A}(x,y) = 1$ for β -almost all $(x,y) \in \mathbb{R}^2$. Let us define N, \widetilde{N} , and N_y by

$$N = \{ z \in \mathbb{R} ; A(z) \neq 1 \},$$

$$\widetilde{N} = \{ (x,y) \in \mathbb{R}^2 ; A(x,y) \neq 1 \},$$

$$N_y = \{ x \in \mathbb{R} ; (x,y) \in \widetilde{N} \}, \text{ for } y \in \mathbb{R}.$$

Then it is verified with no trouble that

$$\begin{aligned} & \alpha(N) = \alpha(N_y) & \text{for all } y \ R, \\ & \beta(N) = \int_R \alpha(N_y) \ d\alpha(y) \ , \end{aligned}$$

so that $\alpha(N) = 0$ if and only if $\beta(\tilde{N}) = 0$.

Therefore ψ induces an isomorphism ψ_* from $H^{\mathbb{C}}_{\alpha}(\mathbf{Z};\mathbf{R};2\pi\mathbf{Z})$ into $H^{\mathbb{C}}_{\beta}(\mathbf{R};\mathbf{R}^2;(2\pi\mathbf{Z})^2)$. [Q.E.D.]

By the above Lemma 2.5.4 and Proposition 2.5.3, we have the following proposition.

Proposition 2.5.5. $(R/Z+2\pi Z)+Q$ is a subgroup of $H_{v}^{C}(R;T^{2})$.

2.5.6. Consequences and examples

(a-3) (R/Z+2 π Z)+Q is a subgroup of H^C_{\alpha}(Z;R;2 π Z).

For example, if we take $A^{q}(z) = e^{-(i/2)qz^{2}}$ (q*E*Z), we get

$$C^{q}(n,z) = e^{(iq/2)n^{2}}e^{iqnz} \qquad (q \in \mathbf{Z}).$$

(b-3) $(\mathbb{R}/\mathbb{Z}+2\pi\mathbb{Z})+\mathbb{Q}$ is a subgroup of $H^{\mathbb{C}}_{\beta}(\mathbb{R};\mathbb{R}^2;(2\pi\mathbb{Z})^2)$. If we take $\widetilde{A}^q(x,y) = \widetilde{A}^q(x-(1/2\pi)\overline{y}) = e^{-(1/2)q(x-(1/2\pi)\overline{y})^2}$ induced from the above A^q (q Z), we get

$$\widetilde{C}^{q}(t,(x,y)) = \widetilde{A}^{q}(x,y)\widetilde{A}^{q}(t (x,y))^{*}$$
$$= e^{(q/2\pi)ix[\overline{y}+2\pi t]}e^{(q/8\pi^{2})i([\overline{y}+2\pi t]-2\overline{y}[\overline{y}+2\pi t]^{2})}$$

for teR, and $(x,y) \in \mathbb{R}^2$. $(q \in \mathbb{Z})$

(a-4) Q is a subgroup of $H^{\mathbb{C}}(\mathbf{Z};\mathbb{R};2^{\pi}\mathbf{Z})$.

(b-4) Q is a subgroup of
$$\tilde{H}^{\mathbb{C}}(R;R^2;(2\pi \mathbf{Z})^2)$$
.

Remark 2.5.7. Furthermore, we note that the weak cohomology groups $H^{\mathbb{C}}(\mathbf{Z};\mathbf{R};2\pi\mathbf{Z})$ and $H^{\mathbb{C}}(\mathbf{R};\mathbf{R}^{2};(2\pi\mathbf{Z})^{2})$ have the cardinal number of uncountable infinity (see [27].).

2.6. An application to decompositions of regular representations

In this section, we consider again decompositions of regular representations of semi-direct product groups related with the cohomology groups. We assume that all notaions and situations are similar as described in section 1.3.

Let λ be the right regular representation of a semi-direct

product group G = $N_{X_S}K$ where N and K are locally compact abelian groups. In this section, for simplicity, we assume that the Haar measure μ of \hat{N} is invariant under the action of K. Then, the regular representation λ of G may be realized on the Hilbert space $L^2(\hat{N} \times K)$ as follows.

Lemma 2.6.1. For $\xi(\chi,t) \in L^2(\hat{N} \times K)$,

$$(\lambda_{(z,k)}\xi)(\chi,t) = \langle z,\chi\rangle\xi(k\cdot\chi,t-k)$$

for $(z,k) \in N \times_{S} K = G$.

Proof. For $\xi(g) \in L^2(G)$, put $(W_1\xi)(g) = \xi(g^{-1})$. Then, W_1 is a unitary operator on $L^2(G)$ because G is unimodular. Let F denote the Fourier transformation of $L^2(N)$ onto $L^2(\widehat{N})$. Take a unitary operator W_2 from $L^2(G)$ onto $L^2(\widehat{N} \times K)$ defined by $W_2 = F \otimes I$ where I is the identity operator on $L^2(K)$ and we identify $L^2(G)$ with $L^2(N) \otimes L^2(K)$ and $L^2(\widehat{N} \times K)$ with $L^2(\widehat{N}) \otimes L^2(K)$. Then, we see that $W_2 W_1 \lambda_{(z,k)} W_1^* W_2^*$ is the desired one by simple calculations. [Q.E.D.]

For a cocycle $C(k,\chi) \in Z^{C}(K; \widehat{N})$, we define a unitary representation λ^{C} of G by, for $\xi(\chi,t) \in L^{2}(\widehat{N} \times K)$,

 $(\lambda_{(z,k)}^{C}\xi)(\chi,t) = C(k,\chi)\langle z,\chi \rangle \xi(k\cdot\chi,t-k)$

for $(z,k) \in \mathbb{N} \times_{S} K = G$. Then, we get the following lemma.

Lemma 2.6.2. λ is unitarily equivalent to $\lambda^{C}.$

Proof. The cocycle $C(k,\chi)$ may be regarded as an element of $Z^{\mathbb{C}}(K; \hat{N} \times K)$. Since the action of K on the space $\hat{N} \times K$ is

smooth and effective, we see that $C(k,\chi)$ becomes a coboundary in $Z^{\mathbb{C}}(K;N\times K)$ by Theorem 2.1.1. Then, there exists an T-valued Borel function $B(\chi,t)$ on $\widehat{N}\times K$ such that, for each $k\in K$,

$$C(k,\chi) = \overline{B(\chi,t)}B(k,\chi,t-k)$$
(2.6.1)

for all $(\chi,t) \in \widehat{\mathbb{N}} \times \mathbb{K}$. Take a unitary operator W given by, for $\xi(\chi,t) \in L^2(\widehat{\mathbb{N}} \times \mathbb{K})$,

$$(W\xi)(\chi,t) = B(\chi,t)\xi(\chi,t).$$

Then, it is easy to check that $W^{*\lambda}_{(z,k)}W = \lambda^{C}_{(z,k)}$ by the equation (2.6.1) and Lemma 2.6.1. [Q.E.D.]

In section 1.3, we gave two kinds of entirely different decompositions of λ related with ergodic measures. Here we can give other many decompositions of λ related with the cohomology group.

The Haar measure μ on $\stackrel{\frown}{N}$ was decomposed into ergodic measures as

$$\mu = \int_{Z}^{\bigoplus} \mu_{\zeta} d\sigma(\zeta).$$

For simplicity of our arguments, we also assume that all components μ_{ζ} ($\zeta \in \mathbb{Z}$) are invariant under the action of K. Then, for a cocycle $C(k,\chi)\in\mathbb{Z}^{\mathbb{C}}(K;\widehat{N})$ and $\eta\in K$, we can define a unitary representation $V^{(C,\eta,\zeta)}$ of G as follows.

For $\xi(\chi) \in L^2(\hat{N}, \mu_{\zeta})$,

$$(\mathbb{V}_{(z,k)}^{(C,n,\zeta)}\xi)(\chi) = C(k,\chi)\langle k,\eta\rangle\langle z,\chi\rangle\xi(k,\chi).$$

Theorem 2.6.3. The right regular representation λ of

- 54 -

 $G = N \times_{S} K$ is decomposed as

$$\lambda \cong \int_{\widehat{K}} \bigoplus_{Z} \Psi^{(C,n,\zeta)} d\sigma(\zeta) d\nu(n).$$

(A) $V^{(C,n,\zeta)}$ (C $\in \mathbb{Z}^{\mathbb{C}}(K; \widehat{\mathbb{N}}), \eta \in \widehat{\mathbb{K}}, \zeta \in \mathbb{Z}$) are irreducible.

(B) $V^{(C,\eta,\zeta)}$ is unitarily equivalent to $V^{(C',\eta',\zeta')}$ if and only if $\zeta' = \zeta$ and $C + \eta$ is μ_{ζ} -cohomologous to $C' + \eta'$.

Proof. We realize λ^{C} on the Hilbert space $L^{2}(\widehat{N} \times \widehat{K})$ as, for $\xi(\chi, \eta) \in L^{2}(\widehat{N} \times \widehat{K})$,

$$(\lambda_{(z,k)}^{\mathbb{C}}\xi)(\chi,\eta) = C(k,\chi)\langle k,\eta \rangle \langle z,\chi \rangle \xi(k \cdot \chi,\eta).$$

Then, by similar arguments as in section 1.3, we get

$$\lambda^{C} \cong \int_{\widehat{K}}^{\Phi} \int_{Z}^{\Phi} \nabla^{(C, \eta, \zeta)} d\sigma(\zeta) d\nu(\eta).$$

Since $\lambda \stackrel{\sim}{=} \lambda^{C}$ by Lemma 2.6.2,

$$\lambda \cong \int_{\widehat{K}}^{\bigoplus} V^{(C,n,\zeta)} d\sigma(\zeta) d\nu(n).$$

The properties will be obtained by the modification of the proof in Theorem 1.3.3. We omit the detail. [Q.E.D.]

By Theorem 2.6.3, we see that regular representations of some concrete non-type I groups, for example, the discrete Mautner group, the discrete Heisenberg group, the Mautner group, and the Dixmier group, have infinitely many completely different irreducible decompositions. For the detail, see [24] and [27]. Chapter III. Generalized induced representations

In this chapter, we investigate generalized induced representations for double transformation groups, related with cohomology and we construct families of non-Mackey representations of certain non-regular semi-direct product groups as a generalization of Mackey's method. Applying this construction to the Mautner group, we obtain a new parametrized family of non-Mackey representations. The representations found by L.Baggett [4] form a part of this family.

In 1978, L.Baggett found a family of non-Mackey irreducible representations of the Mautner group via the decompositions of a generalizd tensor product of some concrete representations [4]. In order to elucidate the mechanism of his family, we develop a theory of generalized induced representations in this chapter. In 1976, A.Ramsay turned the Mackey's theory into a representation theory of measure groupoids [44] and obtained a generalization of induced representations. Our notion is close to his but there are some differences. These differences will be seen to be crucial in the decomposition theory in later chapters. It is known that, for a connected and simply connected solvable Lie group G. there exists an algebraic solvable Lie group \widetilde{G} which contains G such that $[\tilde{G},\tilde{G}] = [G,G] = N$ and \tilde{G} acts on \hat{N} (the dual of N) smoothly. L. Pukanszky made an extensive use of this fact in [41], [42]. We impose similar assumption (*) for nonregular semi-direct product groups, which will be used

effectively as a substitute of this fact.

In section 3.1, for a double transformation group, we define unitary representations in relation to cohomology, which appears as a generalization of the Mackey's induced representations [30], [33].

In section 3.2, following the construction in section 3.1, we have families of non-Mackey representations of non-regular semi-direct product groups satisfying a certain condition (*). In Theorem 3.2.6, we show when such representations are mutually equivalent, and in Theorem 3.2.7, we give a criterion of the irreducibility. In Proposition 3.2.9, we mention a property which characterizes such representations. The results obtained are akin to the results in [33] or [44] but ours are more precise according to the strong conditions imposed. Moreover, the techniques employed by L.Baggett [4] will be better understood from our points of view.

In section 3.3, we apply our general results to the discrete Mautner group and the Mautner group.

3.1. A generalization of induced representations

Let (G;X;H) be a double transformation group. When a continuous unitary representation L of H on a separable Hilbert space $\frac{1}{2}(L)$ is given, we construct a unitary representation of G in the following way. Let Ω denote the commuting algebra of L, in other words, the set of all bounded oparators on $\frac{1}{2}(L)$ which commute with all L_h for $h \in H$. We take an α^u -valued cocycle A(x) of (G;X;H) and denote D(h,x) = A(x)^{*}A(x \cdot h). Then D(h,x) is an α^u -valued cocycle of (H;X) which is G-invariant

- 57 -

with respect to the variable x EX.

We assume that (H;X) is smooth. Let μ be a quasiinvariant Radon measure on X/H under the action of G. We put $\mu_{g}(E) = \mu(g \cdot E)$ for Borel sets E of X/H and let $\sigma(g, \dot{x})$ be a Radon-Nikodym derivative of μ_{g} by μ for each géG. We assume that there exists a Borel function $\tilde{\sigma}(g, \dot{x})$ on G×X/H such that for each géG, $\tilde{\sigma}(g, \dot{x}) = \sigma(g, \dot{x})$ for μ -almost all $\dot{x} \in X/H$. Put $\rho(g, x) = \tilde{\sigma}(g, \dot{x})$, then $\rho(g, x)$ is a Borel function on G×X and H-invariant with respect to the variable $x \in X$.

Let ξ^A denote the set of all f's satisfying the following conditions.

- (1) f is a weakly Borel function on X with values in f(L),
- (2) $f(x \cdot h) = D(h,x)^* L_h^* f(x)$ for all $x \in X$ and all $h \in H$,
- (3) $\int_{\mathbf{Y}/\mathbf{H}} \|f(\mathbf{x})\|^2 d\mu(\mathbf{x})$ is finite.

We define the inner product of $\frac{A}{A}$ by

$$(f;f') = \int_{X/H} (f(x),f'(x)) d_{\mu}(x) \quad \text{for } f \text{ and } f' \in \mathcal{J}^{A}$$

Then it is verified by usual arguments that f_a^A is a Hilbert space with (;). For each geG, if $f(x) \in f_a^A$, then $\rho(g,x)^{1/2} f(g \cdot x) \in f_a^A$, and they have the same norm. So we have a unitary oparator U_g : $f(x) \longrightarrow \rho(g,x)^{1/2} f(g \cdot x)$ on f_a^A .

Proposition 3.1.1. U (g \longrightarrow Ug) is a continuous unitary representation of G.

Proof. This follows by routine arguments. [Q.E.D.]

We note that if (H;X) is effective, g_{μ}^{A} is not empty and

- 58 -

moreover isomorphic to $L^2(X/H; \frac{1}{2}(L); \mu)$. In fact, by the assumption that (H;X) is smooth and effective, we can decompose $x \in X$ such that $x = b(x) \cdot a(x)$ where a is a Borel function from X to H and b is a Borel function from X to a cross section of X under the action of H. Let f(x) be an $\frac{1}{2}(L)$ -valued weakly Borel function on X/H such that $\int_{X/H} ||f(x)||^2 d_{\mu}(x)$ is finite. Put

$$\tilde{f}(x) = A(x)^* L_{a(x)}^* f(x)$$
 for $x \in X$,

then we get $f \in h^A$. The correspondence $f \longrightarrow \tilde{f}$ induces an isomorphism from $L^2(X/H; f_2(L); \mu)$ onto f_2^A .

Let us consider the case where X is a group \tilde{G} , H is a closed subgroup of \tilde{G} , and G is taken to be equal to \tilde{G} . Then the cocycles A are all trivial. Let \nvdash be the canonical quasi-invariant measure on $H\backslash\tilde{G}$. Under these situations, the representation U defined in the above reduces to the ordinary induced representation of G from L (see [30]). If we take G as a closed subgroup of \tilde{G} , and the cocycle A to be trivial, then the above U reduces to the restriction of Ind $\frac{\tilde{G}}{H}L$ to G. Therefore we can regard the above representation U as a "generalized induced representation of G from L through (G;X;H) twisted by the cocycle A", and we denote U by Ind $_{H}^{G}(\mu,A,L)$.

For A_i in $Z^{(G;X;H)}$ we denote $\operatorname{Ind}_H^G(\mu, A_i, L)$ by $U^{A_i}(i=1,2)$. Then we have the following proposition.

Proposition 3.1.2. If A_1 is cohomologous to A_2 , U^{A_1} is unitarily equivalent to U^{A_2} .

- 59 -

Proof. From A_i , we get σ^u -valued cocycle D_i of (H;X) by

$$D_{i}(h,x) = A_{i}(x)^{*}A_{i}(x \cdot h)$$
 for $h \in H$ and $x \in X$ (i=1,2).

Since A_1 is cohomologous to A_2 , there exists an \mathcal{O}_1^u -valued G-invariant Borel function B on X such that

$$D_2(h,x) = B(x)^* D_1(h,x)B(x \cdot h)$$
 for all $h \in H$ and all $x \in X$.

Let $f_{\lambda}^{A_{i}}$ be the Hilbert space corresponding to A_{i} . If $f(x) \in f_{\lambda}^{A_{i}}$, then $\tilde{f}(x) = B(x)^{*}f(x)$ is in $f_{\lambda}^{A_{2}}$ and $||f|| = ||\tilde{f}||$. Indeed, for example,

$$\begin{aligned} &\widehat{f}(x \cdot h) \\ &= B(x \cdot h)^* f(x \cdot h) \\ &= B(x \cdot h)^* D_1(h, x)^* L_h^* f(x) \\ &= B(x \cdot h)^* D_1(h, x)^* B(x) L_h^* B(x)^* f(x) \\ &= D_2(h, x)^* L_h^* f(x) \quad \text{for all } h \in H \text{ and all } x \in X \end{aligned}$$

Hence we get a unitary operator $W : f(x) \longrightarrow B(x)^* f(x)$ from g^A to g^A , and it is easy to see that

$$U_{\alpha}^{A'} = W^* U_{\alpha}^{A_2} W$$
 for all geG.

This implies that $U^{A_{f}}$ is unitarily equivalent to $U^{A_{2}}$. [Q.E.D.]

As a result of the above proposition, we see that U^A is defined essentialy by the cohomology class [A] in $Z^{n}(G;X;H)$. Suppose that L is a multiplicity free representation of H and a double transformation group (G;X;H) satisfies the assumption of Proposition 2.2.3. Then the elements of the

- 60 -

three groups $H^{\Omega}(G;X;H)$, $H^{\Omega}(G;X/H)$, and $H^{\Omega}(H;GX)$, which correspond one another by Proposition 2.2.3, determine a generalized induced representation of G up to equivalence.

At this moment, we shall look at the representations corresponding to the elements of $H^{I\!\!Q}(G;X/H)$ somewhat in detail. Take an \mathfrak{R}^{U} -valued H-invariant cocycle C of (G;X) derived from a cocycle A by $C(g,x) = A(x)A(g\cdot x)^{*}$. We define a representation V^{C} of G as follows. In place of the condition (2) in the definition of \mathcal{L}^{A}_{r} , we put

(2)' $f(x \cdot h) = L_h^* f(x)$ for $h \in H$ and $x \in X$, and

leave the other conditions behind. We denote the set of all such f's by ξ . Define V_g^C by, for geG,

$$V_{g}^{C}: \ \xi \ni f(x) \longrightarrow \rho(g,x)^{1/2} C(g,x) f(g \cdot x) \in \xi.$$

It is easily seen that $V^{C}(g \longrightarrow V_{g}^{C})$ is also a unitary representation of G and it is unitarily equivalent to U^{A} . Moreover, it is verified with no trouble that $V^{C_{I}}$ is unitarily equivalent to $V^{C_{2}}$ even if C_{1} is " μ "-cohomologous to C_{2} as elements in $Z^{O}(G;X/H)$. In the case of Remark 2.2.4, generalized induced representations are determined up to equivalence by the mutually corresponding elements of the cohomology groups $H^{O}_{\alpha}(G;G;H)$, $H^{O}_{\mu}(G;H|G)$, and $H^{O}_{\nu}(H;G/G)$ by Theorem 2.3.2.

3.2. Irreducible representations of semi-direct product groups

Let G be a semi-direct product group $N \times_{s}^{K} K$, where K acts

- 61 -

on N as an automorphism group. We assume that N and K are abelian groups. The action is denoted by $N \rightarrow z \longrightarrow k \cdot z \in N$. The element of G is written as (z,k) $(z \in N, k \in K)$ and the multiplication is given by

$$(z,k)(z',k') = (z+k z', k+k').$$

Then an action of K on the topological space \hat{N} , the dual of N, is defined, for k K and $\chi \in \hat{N}$, by

$$\langle z, k \cdot \chi \rangle = \langle k \cdot z, \chi \rangle$$
 for all $z \in \mathbb{N}$.

So we get a topological transformation group $(K; \hat{N})$. G.W.Mackey called G = N×_SK a "regular" semi-direct product group when $(K; \hat{N})$ is smooth, and determined all irreducible representations of such a group. We shall treat mainly the case where G is not a regular semi-direct product group, and try to construct a family of non-Mackey irreducible representations of G. To do this, our main assumption is this.

(*) There is an abelian group \tilde{K} containing K as closed subgroup. The group \tilde{K} acts on \hat{N} as an automorphism group and, as such, it is an extention of K. $\hat{G} = N \times_{S} \tilde{K}$ is a regular semi-direct product group.

3.2.1. A construction of representations of G

First, we take $\chi \in \widehat{\mathbb{N}}$ and denote by \mathbb{H}_{χ} , the stability group of $\widetilde{\mathbb{K}}$ at χ , i.e. the set of $t \in \widetilde{\mathbb{K}}$ such that $t \cdot \chi = \chi$. \mathbb{H}_{χ} is a closed subgroup of $\widetilde{\mathbb{K}}$. Put $\mathbb{G}_{\chi} = \mathbb{N}^{\times}{}_{\mathrm{s}}\mathbb{H}_{\chi}$. We define \mathbb{L}^{χ} by

$$L_{(z,h)}^{\chi} = \langle z, \chi \rangle$$
 for all $(z,h) \in G_{\chi}$

- 62 -

Then L^{χ} is a unitary character of G_{χ} . K and H_{χ} are closed subgroups of \tilde{K} , G and G_{χ} are closed subgroups of \tilde{G} , and so we get double transformation groups $(K;\tilde{K};H_{\chi})$ and $(G;\tilde{G};G_{\chi})$ as in Remark 2.2.4. Next, we take a **T**-valued cocycle A of $(K;\tilde{K};H_{\chi})$ where **T** is the one-dimensional torus which equals $\mathbb{C}^{u} = \{z \in \mathbb{C}; |z| = 1\}$. If we put

$$\widetilde{A}(z,t) = A(t)$$
 for all $(z,t) \in \widetilde{G}$,

then \tilde{A} is a **T**-valued cocycle of $(G; \tilde{G}; G_{\chi})$. Let μ be a Haar measure of the abelian group $H_{\chi} \setminus \tilde{K} \cong G_{\chi} \setminus \tilde{G}$. Under these preparations, we define a unitary representation $U^{(\chi, A)} = \operatorname{Ind}_{G_{\chi}}^{G}(\mu, \tilde{A}, L^{\lambda})$ of G, as described in section 3.1.1.

Remark 3.2.2. Since a unitary character ϕ of \widetilde{K} is a cocycle of $(K;\widetilde{K};H_{\chi})$, we get $U^{(\chi,\phi)}$. This is a typical example, and, in the case that $G = \widetilde{G}$ and therefore $K = \widetilde{K}$, $U^{(\chi,\phi)}$ coincides with the representation obtained by the Mackey's method.

3.2.3. Realization of $U^{(\chi,A)}$ on $L^{2}(H_{\chi} \setminus \widetilde{K}, \mu)$

Let C(k,t) be a T-valued cocycle of $(K;\tilde{K})$ defined by $C(k,t) = A(t)A(t+k)^*$. As C(k,t) is H_{χ} -invariant with respect to $t\in \tilde{K}$, we can regard it as a cocycle of $(K;H_{\chi}\setminus \tilde{K})$ and when it is considered in such a way, it is written as C(k,x). Since H_{χ} is a closed subgroup of \tilde{K} , there exists a Borel cross section $c : x \longrightarrow c(x)$ from $H_{\chi}\setminus \tilde{K}$ to \tilde{K} . Then, by routine arguments, we have the following result.

For
$$f(x) \in L^{2}(H_{\chi} \widetilde{K}, \mu)$$

 $(U_{(z,k)}^{(\chi,A)}f)(x) = C(k,x) \langle z, c(x) \cdot k \rangle f(x \cdot k)$

for all $(z,k)\in G$.

3.2.4. Realization of $U^{(\chi,A)}$ on $L^{2}(\widehat{N},\widetilde{\mu})$

The Borel map ψ : $H_{\chi}+t \longrightarrow t\cdot\chi$ from $H_{\chi} \stackrel{\sim}{K}$ to $\stackrel{\sim}{N}$ induces a Radon measure $\stackrel{\sim}{\mu} = \psi_{\ast}(\mu)$ on $\stackrel{\sim}{N}$ concentrated on $Orb_{\widetilde{K}}(\chi)$, the set{ $t \cdot \chi \in \stackrel{\sim}{N}$; $t \in \stackrel{\sim}{K}$ }. We define a cocycle $C(k, \omega)$ of $(K; \stackrel{\sim}{N})$ by

$$C(k,\omega) = \begin{cases} C(k,t) & \text{if } \omega \in \operatorname{Orb}_{\widetilde{K}}(\chi) \text{ and } \omega = t \cdot \chi \\ \\ 1 & \text{if } \omega \notin \operatorname{Orb}_{\widetilde{K}}(\chi). \end{cases}$$

Then we can also realize $U^{(\chi,A)}$ on $L^2(\hat{N},\hat{\mu})$ in the following form.

For
$$f(\omega) \in L^2(\tilde{N}, \tilde{\mu})$$

 $(U_{(z,k)}^{(\chi,A)}f)(\omega) = C(k,\omega) \langle z, \omega \rangle f(k \cdot \omega) \text{ for all } (z,k) \in G.$

Define an action of \widetilde{K} on $Z^{\mathbb{C}}(K;\widetilde{K};H_{\chi})$ by

$$(t \cdot A)(t') = A(t+t')$$
 for $t \in \widetilde{K}$ and all $t' \in \widetilde{K}$,

and transfer this action to C in $Z^{\mathbb{C}}(K;\mathbb{N})$. Then we get

$$(t \cdot C)(k, \omega) = C(k, t \cdot \omega)$$

for $t \in \tilde{K}$ and all $(k, \omega) \in K \times N$.

Let V be a unitary representation of \tilde{K} on $L^2(\hat{N}, \tilde{\mu})$ obtained by putting, for $f(\omega) \in L^2(\hat{N}, \tilde{\mu})$,

$$(V_t f)(\omega) = f(t \cdot \omega)$$
 for $t \in \widetilde{K}$.

- 64 -
Then we have, for $f(\omega) \in L^2(\widehat{N}, \widetilde{\mu})$,

$$(\mathbb{V}_{t}\mathbb{U}_{(z,k)}^{(\chi,A)}\mathbb{V}_{t}^{*}f)(\omega) = \mathbb{C}(k,t\cdot\omega) \langle z,t\cdot\omega \rangle f(k\cdot\omega)$$
$$= (\mathbb{U}_{(z,k)}^{(t,\chi,tA)}f)(\omega) \quad \text{for all } (z,k) \in \mathbb{G}.$$

This fact shows the following lemma.

Lemma 3.2.5. $U^{(\chi,A)}$ is unitarily equivalent to $U^{(t\cdot\chi,t\cdot A)}$ for for all $t\in \widetilde{K}$.

Let α be a Haar measure of $\widetilde{K}.$ Then we have the following theorem.

Theorem 3.2.6. $U^{(X_1,A_1)}$ is unitarily equivalent to $U^{(X_2,A_2)}$ if and only if $X_2 \in \operatorname{Orb}_{\widetilde{K}}(X_1)$ and A_2 is α -cohomologous to $t_0 \cdot A_1$ where $t_0 \in \widetilde{K}$ satisfies $X_2 = t_0 \cdot X_1$.

Proof. The "if" part is easily verified by Lemma 3.2.5 and the last assertion in section 3.1. We show the "only if" part.

Suppose that $U^{(\chi_1,A_1)} = U^{(\chi_2,A_2)}$. Then, We have $\begin{bmatrix} U^{(\chi_1,A_1)} = & U^{(\chi_2,A_2)} \end{bmatrix}$ which are restrictions to N of $U^{(\chi_1,A_1)}$ and $U^{(\chi_2,A_2)}$. Let

$$N \left| U^{(\chi_{i},A_{i})} \right| = \int_{N}^{\Phi} \omega d\tilde{\mu}_{i}(\omega) \qquad (i = 1,2)$$

be the irreducible decompositions of $_{N}|_{U}^{(\chi_{i}^{\cdot},A_{i}^{\cdot})}$, the measures μ_{i} being concentrated on $Orb_{\widetilde{K}}(\chi_{i})$ by the definition of $U^{(\chi_{i}^{\cdot},A_{i}^{\cdot})}$. As such a decomposition of an abelian group is unique, we have first $Orb_{\widetilde{K}}(\chi_{1}) = Orb_{\widetilde{K}}(\chi_{2})$. Hence there exists a $t_{0} \in \widetilde{K}$ such that $\chi_{2} = t_{0} \cdot \chi_{1}$. By Lemma 3.2.5,

- 65 -

 $U^{(X_1,A_1)} \cong U^{(t_0X_1,t_0,A_1)} \cong U^{(X_2,t_0,A_1)}$ holds, so that we have $U^{(X_2,t_0,A_2)} \cong U^{(X_2,A_2)}$. Therefore if we show that $U^{(X,A_1)} \cong U^{(X,A_2)}$ implies that A_2 is α -cohomologous to A_1 , the proof will be complete.

Let $U^{(\chi,A_1)}$ and $U^{(\chi,A_2)}$ be realized on $L^2(\hat{N}, \tilde{\mu})$ and let W be a unitary operator on $L^2(\hat{N}, \tilde{\mu})$ such that

$$U_{(z,k)}^{(X,A_2)} = W^* U_{(z,k)}^{(X,A_1)} W \quad \text{for all } (z,k) \in G.$$

From the expressions of the operators $U_{(z,k)}^{(\chi,A)}$ in 3.2.4, we have

$$(U_{(z,0)}^{(\chi,A)}f)(\omega) = \langle z, \omega \rangle f(\omega) \qquad f(\omega) \in L^{2}(\hat{N}, \hat{\mu}),$$

therefore $U_{(z,0)}^{(\chi,A_1)} = U_{(z,0)}^{(\chi,A_2)}$ for all z6N, and these operators generate a maximal abelian von Neumann algebra $L^{\infty}(\hat{N}, \tilde{\mu})$ on $L^{2}(\hat{N}, \tilde{\mu})$. Then by the condition $U_{(z,0)}^{(\chi,A_1)}W = WU_{(z,0)}^{(\chi,A_2)}$ for all z6N, W must be in $L^{\infty}(\hat{N}, \tilde{\mu})$, i.e. W is equal to a multiplicative operator $\hat{B}(\omega)$ such that $|\tilde{B}(\omega)| = 1$. Then we see that, for $f(\omega) \in L^{2}(\hat{N}, \tilde{\mu})$, on the one hand we get

$$(W^{*}U_{(0,k)}^{(X,A_{i})}Wf)(\omega) = B(\omega)^{*}\widetilde{C}_{1}(k,\omega)\widetilde{B}(k\cdot\omega)f(k\cdot\omega),$$

and on the other hand we get

$$(U_{(0,k)}^{(\chi,A_2)}f)(\omega) = \widehat{C}_2(k,\omega)f(k\cdot\omega),$$

where \hat{C}_1 and \hat{C}_2 are the canonical cocycle of $(K; \hat{N})$ corresponding to A_1 and A_2 . So we have, for each k $\in K$,

$$\widetilde{C}_{2}(k,\omega) = \widetilde{B}(\omega)^{*}\widetilde{C}_{1}(k,\omega)\widetilde{B}(k\cdot\omega)$$
 for μ -a.a. $\omega \in \mathbb{N}$.

Take a T-valued Borel function B_0 on $H_\chi \widetilde{K}$ such that

- 66 -

$$\begin{split} \widetilde{B}_{0}(\omega) &= \widetilde{B}(\omega) \text{ for } \widetilde{\mu} - \text{almost all } \omega \in \widehat{N}, \text{ where } \widetilde{B}_{0} \text{ is defined by} \\ \\ \widetilde{B}_{0}(\omega) &= \begin{cases} B_{0}(t) & \text{if } \omega \in \operatorname{Orb}_{\widetilde{K}}(X) \text{ and } \omega = t \cdot X, \\ 1 & \text{if } \omega \notin \operatorname{Orb}_{\widetilde{K}}(X). \end{cases} \end{split}$$

Then we see that for each $k \in K$

$$C_2(k,x) = B_0(x)^* C_1(k,x) B_0(x k)$$
 for μ -a.a. $x \in H_{\chi} \setminus K$,

in other words, C_2 is μ -cohomologous to C_1 . Therefore it follows that A_2 is α -cohomologous to A_1 by Theorem 2.3.2. [Q.E.D.]

Next we give a necessary and sufficient condition that $U^{(\chi,A)}$ is irreducible. Let $\operatorname{Orb}_{K}(\chi)$ be the set $\{k \cdot \chi \in \widehat{N} : k \in K\}$ and $\operatorname{Orb}_{\widetilde{K}}(\chi)$ be the set $\{t \cdot \chi \in \widehat{N} : t \in \widetilde{K}\}$, both of them are considered as a topological space with the topology induced from \widehat{N} .

Theorem 3.2.7. $U^{(X,A)}$ is an irreducible representation of G if and only if $Orb_{K}(X)$ is dense in $Orb_{\widetilde{K}}(X)$.

Proof. First, we show the "if" part. Since $(\tilde{K}; \hat{N})$ is smooth, we see that the map Ψ : $H_{\chi}+t \longrightarrow t \cdot X$ from $H_{\chi} \cdot \tilde{K}$ to $\operatorname{Orb}_{\widetilde{K}}(\chi)$ is homeomorphic (see[12]). The set $\Psi^{-1}(\operatorname{Orb}_{K}(\chi))$ is equal to the orbit of K on $H_{\chi} \cdot \tilde{K}$ passing the unit element of $H_{\chi} \cdot \tilde{K}$, and it is dense in $H_{\chi} \cdot \tilde{K}$ by the assumption. Moreover the action of K on $H_{\chi} \cdot \tilde{K}$ is equal to the action of $H_{\chi} \cdot H_{\chi} + K$ on $H_{\chi} \cdot \tilde{K}$ as under a subgroup. Therefore the Haar measure μ of $H_{\chi} \cdot \tilde{K}$ is invariant and ergodic under the action of K, and so is $\tilde{\mu}$ on \hat{N} . Suppose that an operator W on $L^2(\hat{N}, \hat{\mu})$ satisfies

$$U_{(z,0)}^{(X,A)}W = WU_{(z,k)}^{(X,A)}$$
 for all $(z,k)\in G$.

Then we get

$$U_{(z,0)}^{(X,A)}W = WU_{(z,0)}^{(X,A)} \quad \text{for all } z \in \mathbb{N},$$

so that W must be a multiplicative operator $\widetilde{B}(\omega)$ in the same way as in the above proof. We observe that, for $f(\omega)\in L^2(\hat{N},\hat{\mu})$,

$$(U_{(0,k)}^{(\chi,A)}Wf)(\omega) = \widetilde{C}(k,\omega)\widetilde{B}(k \ \omega)f(k \cdot \omega)$$
$$(WU_{(0,k)}^{(\chi,A)}f)(\omega) = \widetilde{B}(\omega)\widetilde{C}(k,\omega)f(k \cdot \omega) \quad \text{for } k \in K.$$

Hence we get, for each $k \in K$,

$$\widetilde{B}(k \cdot \omega) = \widetilde{B}(\omega)$$
 for $\widetilde{\mu}$ -a.a. $\omega \in \mathbb{N}$.

Then

$$\widehat{B}(\omega) = \text{constant}$$
 for $\widehat{\mu}-a.a. \ \omega \in N$,

by the ergodicity of $\tilde{\mu}$, so that W must be a scalar operator on $L^2(\tilde{N}, \tilde{\mu})$. This fact implies that $U^{(\chi, A)}$ is irreducible. The "only if" part is shown in the following way. If $U^{(\chi, A)}$ is irreducible, the measure $\tilde{\mu}$ on \tilde{N} corresponding to $U^{(\chi, A)}$ must be ergodic and so is μ . Moreover, the ergodicity of μ implies that $H_{\chi} H_{\chi} + K$ is a dense subgroup of $H_{\chi} K$ and so $Orb_{K}(\chi)$ is dense in $Orb_{\widetilde{K}}(\chi)$ through the homeomorphism ψ .

Remark 3.2.8. The action of \widetilde{K} on $Z^{\mathbb{C}}(K;\widetilde{K};H_{\chi})$ induces naturally an action of \widetilde{K} on $H^{\mathbb{C}}_{\alpha}(K;\widetilde{K};H_{\chi})$. Let Ω be a \widetilde{K} -invariant Borel subset of \widetilde{N} such that $(K;\Omega)$ is essentially

- 68 -

free i.e. all stability groups H_{χ} of \tilde{K} at $\chi \in \Omega$ are equal to H. Moreover we assume that K+H is dense in \tilde{K} . Then the above two theorems assert that the irreducible representations $U^{(\chi,A)}$ of G are parametrized by the orbits of $\Omega \times H^{\mathbb{C}}_{\alpha}(K;\tilde{K};H)$ under the action of \tilde{K} .

Let $C(\tilde{\mu})$ be a quasi-orbit for the action of G on \hat{N} concentrated on $Orb_{\tilde{K}}(\chi)$ for some $\chi \in \hat{N}$, where $\tilde{\mu} = \psi_*(\mu)$ for a Haar measure μ of $H_{\chi} \setminus \tilde{K}$ through the above map ψ from $H_{\chi} \setminus \tilde{K}$ to \hat{N} . If a **T**-valued Borel function A on \tilde{K} satisfies the condition that, for each $k \in K$ and $h \in H_{\chi}$,

 $A(k+t+h) = A(k+t)A(t)^*A(t+h)$ for α -a.a. $t\in \widetilde{K}$,

we call it an α -cocycle of $(K; \tilde{K}; H_{\chi})$. For an α -cocycle A, we can also define a unitary representation $U^{(\chi, A)}$ of G in the same way as above. Such $U^{(\chi, A)}$ is an irreducible representation of G restricting to $C(\tilde{\mu})$ with multiplicity one if it satisfies the assumption of Theorem 3.2.7. Conversely, we have the following proposition.

Proposition 3.2.9. If U is an irreducible representation of G restricting to $C(\tilde{\mu})$ with multiplicity one, then there exists an α -cocycle A of $(K;\tilde{K};H_{\chi})$ such that U is unitarily equivalent to $U^{(\chi,A)}$.

Proof. By general results [33] about semi-direct product groups, there exists a T-valued Borel function $C(k,\omega)$ on $K \times N$ such that

(1) for $f(\omega) \in L^2(\hat{N}, \tilde{\mu})$, $(U_{(z,k)}f)(\omega) = C(k,\omega) \langle z, \omega \rangle f(k \cdot \omega)$ for $(z,k) \in G$, (2) for each $k_1, k_2 \in K$,

$$C(k_1+k_2,\omega) = C(k_1,\omega)C(k_2,k_1-\omega)$$
 for $\tilde{\mu}-a.a. \omega \in \mathbb{N}$.

Take a Borel function D(k,t) on $K_{\mathbf{x}}\widetilde{K}$ such that $D(k,t) = C(k,\psi(t))$. Then D(k,t) is H_{χ} -invariant with respect to $t\in\widetilde{K}$ and satisfies, for each k_1,k_2 K,

$$D(k_1+k_2,t) = D(k_1,t)D(k_2,k_1+t)$$
 for α -a.a. $t\in \widetilde{K}$.

We choose a Borel cross section $c_1: K\setminus \widetilde{K} \longrightarrow \widetilde{K}$ and replace t by $k+c_1(x)$ ($k\in K, x\in K\setminus \widetilde{K}$) in the above expression. Then, using Fubini's theorem, we see the existence of an element k_0 in K for which, rewriting c(x) instead of $k_0+c_1(x)$, we can claim for almost all $x\in K\setminus \widetilde{K}$,

$$D(k_1+k_2,c(x)) = D(k_1,c(x))D(k_2,c(x)+k_1) \quad (3.2.1)$$

for almost all $k_1, k_2 \in K$.

Put b(t) = c(t) and a(t) = t-b(t) for $t \in K$. We define a **T**-valued Borl function A(t) on K by

$$A(t) = D(a(t), b(t))$$

Let $\widetilde{D}(k,t)$ be a cocycle of $(K;\widetilde{K})$ defined by $\widetilde{D}(k,t) = A(t)^*A(t+k)$. Then, (3.2.1) implies that

$$\widetilde{D}(k,t) = D(k,t)$$
 a.a. $(k,t) \in K \times \widetilde{K}$.

However, by the fact that D and \widetilde{D} define the same "continuous" unitary representation of K on $L^2(\widetilde{K})$, we get, for each keK

$$\widetilde{D}(k,t) = D(k,t) \quad \alpha - a.a. t \in \widetilde{K}.$$

Therefore we see that, for each $k \in K$ and each $h \in H_{\gamma}$

$$\widetilde{D}(k,t+h) = D(k,t) \quad \alpha-a.a. te\widetilde{K}.$$

This implies that A is an α -cocycle of $(K;\tilde{K};H_{\chi})$ and it is verified with no trouble that U is unitarily equivalent to $U^{(X,A)}$. [Q.E.D.]

Remark 3.2.10. By applying general considerations about cohomology [43], we may get a stronger result. Under the same assumption of Theorem 3.2.9, we can take a "cocycle" A in $Z^{C}(K;\tilde{K};H_{v})$ instead of a " α -cocycle" A as in Theorem 3.2.9.

3.3. Applications and Examples

In this section, following section 3.2 and using the results in section 2.5, we shall give new families of irreducible representations, which are non-Mackey representations, of the discrete Mautner group and the Mautner group.

Case (a) ; the discrete Mautner group.

Let G be the discrete Mautner group defined to be the semi-direct product $C \times_S Z$ of the additive group C of complex numbers with the additive group Z of integers, where the multiplication is given by

$$(z,n)(z',n') = (z+e^{in}z',n+n').$$

- 71 -

Corresponding to this group, we take the universal covering group \tilde{G} of the motion group, which is defined to be the semi-direct product $C \times_{S} R$, where R is the additive group of real numbers and the multiplication is given by

$$(z,t)(z',t') = (z+e^{it}z',t+t')$$

We regard G as a closed subgroup of $\widetilde{\mathsf{G}}$.

At first, we take $\chi^r \epsilon \hat{c}$ such that $r \epsilon R^+$ (positive real numbers) and

$$\langle z, \chi^r \rangle = e^{i(r,z)}$$
 for all $z \in C$,

where (,) means the real inner product in C. Then the stability group of R at X^{r} is equal to $2\pi Z$ for all $r \epsilon R^{+}$. Let G₀ denote the semi-direct product $C \times_{S} 2\pi Z$, and we take a unitary character L^{r} of G₀ defined by

$$L^{r}(z,2\pi n) = \langle z, \chi^{r} \rangle$$
 for all $(z,2\pi n) \epsilon G_{0}$.

Next, we take $A^{(\lambda,q)} \in \mathbb{Z}^{\mathbb{C}}(\mathbb{Z};\mathbb{R};2\pi\mathbb{Z})$ where $\lambda \in \mathbb{R}$, $q \in \mathbb{Q}$ according to (a-3) of section 2.5.

Under these preparations, we get a unitary representation $U^{(r,\lambda,q)}$ of G by section 3.2 as

$$U^{(r,\lambda,q)} = \operatorname{Ind}_{G_0}^{G}(A^{(\lambda,q)};L^r) \quad (r \in \mathbb{R}, \lambda \in \mathbb{R}, q \in \mathbb{Q})$$

As $\mathbf{Z}+2\pi\mathbf{Z}$ is dense in R, $U^{(\mathbf{r},\lambda,\mathbf{q})}$ are irreducible by Theorem 3.2.7 or Remark 3.2.8. Moreover we see that $U^{(\mathbf{r},\lambda,\mathbf{q})}$ is unitarily equivalent to $U^{(\mathbf{r}',\lambda',\mathbf{q}')}$ if and only if $\mathbf{r}=\mathbf{r}'$, $\mathbf{q}=\mathbf{q}'$, and $\lambda-\lambda' \epsilon \mathbf{Z}+2\pi\mathbf{Z}$ by Theorem 3.2.6 and (a-3) of 2.5.6. It is clear that these $U^{(r,\lambda,q)}$ are non-Mackey representations. When $q \in \mathbb{Z}$, they are reduced to those obtained by L.Baggett [4]. However, when $q \notin \mathbb{Z}$, they are "new".

Case (b) ; the Mautner group

This time, Let G denote the Mautner group defined as a semi-direct product $c^2 \times_s R$ of the two-dimensional vecter group c^2 on complex numbers with R, where the multiplication is given by

Corresponding to this group G, we pick up the connected and simply connected 6-dimensional algebraic solvable Lie group \widetilde{G} , which is defined as the semi-direct product $C^2 \times_{s} R^2$, where the multiplication is given by

We imbed G in $\widetilde{\mathsf{G}}$ as a closed subgroup by the following injection.

As in the case (a), we take at first $\chi^{(r,s)} \epsilon c^{2}$ such that r, $s \epsilon R^{+}$ and

$$\langle (z,w) \ \chi^{(r,s)} \rangle = e^{i(r,z)}e^{i(s,w)}$$
 for all $(z,w) \epsilon \mathfrak{c}^2$.

Then the stability group of \mathbb{R}^2 at $\chi^{(r,s)}$ is equal to $(2\pi \mathbf{Z})^2$ for any r, $s \in \mathbb{R}^+$. Let \mathbb{G}_0 be the semi-direct product $\mathbb{C}^2 \times_s (2\pi \mathbf{Z})^2$ of \mathbb{C}^2 with $(2\pi \mathbf{Z})^2$ and $L^{(r,s)}$ be a unitary character of \mathbb{G}_0 defined by

$$L_{(z,w,2\pi m,2\pi n)}^{(r,s)} = \langle (z,w), \chi^{(r,s)} \rangle$$
 for $(z,w,2\pi m,2\pi n) \in G_0$.

- 73 -

Next, we take $A^{(\lambda,q)} \in \mathbb{Z}^{\mathbb{C}}(\mathbb{R};\mathbb{R}^2;(2\pi\mathbb{Z})^2)$ by (b-3) of 2.5.6 where $\lambda \in \mathbb{R}$ and $q \in \mathbb{Q}$.

Under these preparations, we get the following unitary representation of G (see section 3.2).

$$U^{(r,s,\lambda,q)} = \operatorname{Ind}_{G_0}^G(A^{(\lambda,q)};L^{(r,s)})$$

where $r, s \in \mathbb{R}^+$, $\lambda \in \mathbb{R}$, $q \in \mathbb{Q}$.

The $U^{(r,s,\lambda,q)}$ are irreducible by Theorem 3.2.7 and they are non-Mackey representations. Moreover, $U^{(r,s,\lambda,q)}$ is unitarily equivalent to $U^{(r',s',\lambda',q')}$ if and only if r=r', s=s' q=q', and $\lambda'\lambda\in\mathbb{Z}+2\pi\mathbb{Z}$ by Theorem 3.2.6 and (b-3) of 2.5.6. In case $q\in\mathbb{Z}$, we can easily write them in a concrete form as follows. For $f(x,y)\in L^2([0,2\pi)\times[0,2\pi))$

$$(U_{z,w,t}^{(r,s,\lambda,q)}f)(x,y) = e^{i(q/2\pi)x[y+2\pi t]}e^{i(q/8\pi^{2})([y+2\pi t]-2y[y+2\pi t]^{2})}$$
$$e^{i\lambda t}e^{i(re^{-ix},z)}e^{i(se^{-iy},w)}f(\overline{x+t},\overline{y+2\pi t}) \quad \text{for } (z,w,t) \in \mathbb{C}$$

It is verified that these are unitarily equivalent to those found by L. Baggett[4]. In other cases, namely, $q \notin \mathbf{Z}$, the $U^{(r,s,\lambda,q)}$ (r,s $\in \mathbb{R}^+, \lambda \in \mathbb{R}, q \in \mathbb{Q}$) are "new" representations. According to 3.2.3 and section 2.5, it is not hard to write down these representations but they are much complicated than those shown in the above in the case $q \in \mathbf{Z}$.

- 74 -

Chapter IV. Decompositions of some factor representations

In this chapter, we consider the irreducible decompositions of type II factor representations of some non-regular semi-direct product groups. Taking a certain factor representation of such a group, we show it can be decomposed in many different ways into direct integrals of irreducible representations, while the diagonal algebras are spatially isomorphic each other. The explicit form of the diagonal algebra is also given.

It is well-known that irreducible decompositions of a non-type I representation are not unique in general. There are some examples which demonstrate this fact. There are about regular representations of certain groups, due to H. Yoshizawa [46], G.W. Mackey [29], A.A. Kirillov [28], and M. Saito [46], as introduced in chapter I. Moreover, M. Takesaki [50] and S. Funakoshi [14] studied decompositions of representations, related to ergodic measures.

The theory of irreducible decompositions is based on the following general result of F.I. Mautner [36]. Let G be a locally compact group and π be a unitary representation of G on a separable Hilbert sapce f_{μ} . Suppose that \Re is an abelian von Neumann subalgebra in $\pi(G)$ '. Then, there exists a standard measure space (Y, ν) such that \Re is algebraically isomorphic with $L^{\infty}(Y, \nu)$ and π is decomposed as follows.

$$\pi \cong \int_{Y}^{\oplus} \pi^{\eta} d\nu(\eta) .$$

- 75 -

Moreover, π^{η} is irreducible for ν -almost all $\eta \in Y$ if and only if (] is maximal abelian in $\pi(G)$ '.

In the chapter, we consider the irreducible decompositions of type II factor representations of some non-regular semidirect product group. In Theorem 4.1.3, a certain representation π^{χ} of such a group G will be decomposed in an explicit way to a direct integral of irreducible representations, each component having a definite form. The corresponding to maximal abelian von Neumann subalgebra in $\pi^{\chi}(G)$ ' is also described in a concrete form.

It is known that the non-type I'ness of a locally compact group or a C^* -algebra is closely related to the nonsmoothness of topological transformation groups [12], [15], [16]. In non-smooth topological transformation groups, there are various kinds of quasi-orbits and the cohomology group for each non-transitive quasi-orbit seems to be huge, at least it is known to be non-trivial under some conditions [38]. The non-uniqueness of decompositions of a non-type I representation seems to depend deeply on these two facts. The results in [14] and [50] are certainly connected with the former and the examples in [28], [29], and section 1.3 also seem to be so intrincically. The present chapter is an attempt to describe the relation of the non-uniqueness of decompositions with the latter phenomenon.

In chapter II, we studied the cohomology groups of double transformation groups, related with irreducible representations of some non-regular semi-direct product groups. The decomposition in Theorem 4.1.3 is done by using

- 76 -

a cocycle of this cohomology theory, and it is shown in Proposition 4.2.1 that two decompositions are completely different when the used cocycles are not weakly cohomologous, whereas the diagonal algebras are spatially isomorphic each other. Thus we may get a large number of different decompositions of a given representation into irreducible components, an observation which will be new. To illustrate various possibilities, we give two examples in section 4.3.

4.1. Decompositions of π^{χ}

Let G be a semi-direct product group N \times_{s}^{K} , where K acts on N as an automorphism group. We assume that N and K are locally compact abelian groups which satisfy the axiom of second countability. The action is denoted by $N \ni z \longrightarrow k \cdot z \in N$ for $k \in K$. The element of G is written as (z, k) $(z \in N, k \in K)$ and the multiplication is given by $(z, k)(z', k') = (z+k\cdot z', k+k')$. Then, an action of K on the topological space $\stackrel{\frown}{\mathrm{N}}$ (the dual of N) is defined, for k \in K and $\omega \in \widehat{N}$, by $\langle z, k \cdot \omega \rangle = \langle k \cdot z, \omega \rangle$ for all $z \in \mathbb{N}$. So we get a topological transformation group (K; \hat{N}) which satisfy $k_2 \cdot (k_1 \cdot \omega) = (k_1 k_2) \cdot \omega$ for $k_1, k_2 \in K$ and $\omega \in \mathbb{N}$. $G = \mathbb{N} \times K$ is called a "regular" semi-direct product group when (K; \hat{N}) is smooth, namely, when each orbit of $\overset{\mathbf{A}}{\mathrm{N}}$ under the action of K is locally closed (see [12]). We treat mainly the case where G is not a regular semi-direct product group. However, we assume the following conditon (*).

(*) There is a locally compact abelian group $\overset{\sim}{\mathrm{K}}$ contain-

- 77 -

ing K as a closed subgroup. The group \tilde{K} acts on N as an automorphism group and, as such, it is an extension of K. \tilde{G} = N ×_s \tilde{K} is a regular semi-direct product group.

For a unitary character X of N, we get a unitary representation π^{χ} of G, defined by $\pi^{\chi} = \iint_{G} \operatorname{Ind}_{N}^{\widetilde{G}} \chi$ which is the restriction to G of the representation of \widetilde{G} induced by χ from N. Let H_{χ} denote the stability group of \widetilde{K} at χ and let T be the one-dimensional torus group. When a T-valued Borel function A on \widetilde{K} satisfies $A(k+t+h) = A(k+t)\overline{A(t)}A(t+h)$ for all $k \in K$, $t \in \widetilde{K}$, and $h \in H_{\chi}$, we call it a cocycle of the double transformation group (K; \widetilde{K} ; H_{χ}) (see section 2.2). Using this cocycle A, we get a cocycle C^{A} of $(H_{\chi}; \widetilde{K})$ and a cocycle D^{A} of (K; \widetilde{K}) by

 $C^{A}(h, t) = \overline{A(t)}A(t+h)$ $D^{A}(k, t) = A(t)\overline{A(t+k)}$

for $h \in H_{\chi}$, $k \in K$, and $t \in \tilde{K}$. We note that $C^{A}(h,t)$ is Kinvariant and $D^{A}(k,t)$ is H₄-nvariant with respect to $t \in \tilde{K}$. In this chapter, we consider the representation π^{χ} of G and, corresponding to each cocycle A, we give decompositions of π^{χ} and the abelian von Neumann algebra in $\pi^{\chi}(G)'$. According to the Mackey's theory of induced represen-

tations [30], we get a canonical decomposition of π^{χ} as follows.

$$\pi^{\chi} = {}_{G} \left| \operatorname{Ind}_{N}^{\widetilde{G}} \chi \right|$$

$$\stackrel{\simeq}{=} {}_{G} \left| \operatorname{Ind}_{G}^{\widetilde{G}} \chi^{(\operatorname{Ind}_{N}^{G_{\chi}} \chi)} \right|, \text{ where } {}_{G} \chi = N \times_{S}^{H_{\chi}},$$

$$7^{9}$$

$$\stackrel{\simeq}{=} \frac{1}{G} \int_{H_{\chi}}^{\Phi} \int_{G_{\chi}}^{\Phi} \int_{H_{\chi}}^{\Phi} \int_{H_{\chi}}^{(\chi,\eta)} d\mu(\eta)$$
$$\stackrel{\simeq}{=} \int_{H_{\chi}}^{\Phi} \int_{G_{\chi}}^{G} \int_{G_{\chi}}^{G} \int_{L}^{(\chi,\eta)} d\mu(\eta) d\mu(\eta)$$

where μ is a Haar measure of \hat{H}_{χ} and $L^{(\chi,\eta)}$ ($\eta \in \hat{H}_{\chi}$) is a unitary character of G_{χ} , defined by $L_{(z,h)}^{(\chi,\eta)} = \langle z, \chi \rangle \langle h, \eta \rangle$ for $(z,h) \in G_{\chi}$.

Now, we consider the following problems.

(a) How do we get decompositions of π^{χ} which are completely different from the above one ?

(b) What is the abelian von Neumann algebra which gives rise to each decomposition ?

At first, we will have a realization of π^{χ} in the following way.

Lemma 4.1.1. $\pi^{\chi} \text{ is realized on } L^2(\widetilde{K})$ by

 $(\pi^{\chi}(z,k)\xi)(t) = \langle z, t \cdot X \rangle \xi(t+k)$

for $\xi(t) \in L^2(\widetilde{K})$ and $(z,k) \in G$.

Proof. This follows from simple calculations (see [30]).

Next, for an arbitrary cocycle A of the double transformation group (K; \tilde{K} ; H_{χ}), a unitary operator λ_{h}^{A} on $L^{2}(\tilde{K})$ for $h \in H_{\chi}$ is defined as follows. For $\xi(t) \in L^{2}(\tilde{K})$, $(\lambda_{h}^{A} \xi)(t) =$ $C^{A}(h,t)\xi(t+h)$. λ^{A} is a unitary representation of H_{χ} . We denote by Ω^{A} the von Neumann algebra generated by λ_{h}^{A} for

- 79 -

all $h \in H_{\chi}$.

Lemma 4.1.2. Π^A is an abelian von Neumann algebra in ${}_{\pi}^{\chi}(G)'$ on $L^2(\widetilde{K}).$

Proof. Since H_{χ} is an abelian subgroup of \widetilde{K} , it is clear that Π^A is abelian. Therefore, it is sufficient to show that

$$\pi^{\chi}(z,k)\lambda_{h}^{A} = \lambda_{h}^{A}\pi^{\chi}(z,k)$$

for all $(z,k) \in G$ and all $h \in H_{\chi}$. For $\xi(t) \in L^2(\widetilde{K})$, we have, on the one hand,

$$(\pi^{\chi}_{(z,k)}\lambda^{A}_{h}\xi)(t)$$

$$= \langle z, t \cdot \chi \rangle (\lambda^{A}_{h}\xi)(t+k)$$

$$= \langle z, t \cdot \chi \rangle C^{A}(h, t+k)\xi(t+k+h)$$

$$= \langle z, t \cdot \chi \rangle C^{A}(h,t)\xi(t+h+k),$$

and on the other hand,

$$(\lambda_{h}^{A}\pi_{(z,k)}^{\chi}\xi)(t) = C^{A}(h,t)(\pi_{(z,k)}^{\chi}\xi)(t+h) = C^{A}(h,t) \langle z, (t+h)\cdot X \rangle \xi(t+h+k) = C^{A}(h,t) \langle z, t\cdot \chi \rangle \xi(t+h+k).$$
 [Q.E.D.]

Now, to practice a decomposition of π^{χ} according to \mathfrak{A}^{A} , we prepare a family of non-Mackey representations $U^{(\chi,A,\eta)}$ $(\eta \in \widehat{H}_{\chi})$ of G = N ×_SK following the way in chapter III. Let $L^{(\chi,\eta)}$ ($\eta \in \widehat{H}_{\chi}$) be unitary characters of G_{χ} = N×_SH_{χ}, defined by

$$L(z,n) = \langle z,\chi \rangle \langle h,\eta \rangle$$

- 80 -

for $(z,h) \in G_{\chi}$. Then we can construct a unitary representation $U^{(\chi,A,\eta)}$ of G by

$$U^{(\chi,A,\eta)} = \operatorname{Ind}_{G_{\chi}}^{G}(A;L^{(\chi,\eta)})$$

When A is trivial,

$$U^{(\chi,A,\eta)} = _{G} \left| \operatorname{Ind}_{G_{\chi}}^{\widetilde{G}} L^{(\chi,\eta)} \right|$$

Our main theorem is the following. It shows an explicit decomposition of π^{χ} corresponding to π^{A} in $\pi^{\chi}(G)$, which will offer one answer of (a) and (b).

Theorem 4.1.3. Corresponding to π^A , the unitary representation π^X of G=N×_SK is decomposed as follows.

$$\pi^{X} \cong \int_{\hat{H}_{X}}^{\bigoplus} U^{(X, A, n)} d\mu(n)$$

where μ is a Haar measure of H_{χ} .

Before going into the proof of the theorem, we will say about $U^{(\chi, A, n)}$ in more detail. Define the action of K on \widetilde{K}/H_{χ} by $k \cdot x = \overline{k+t}$ for $k \in K$ and $x = \overline{t} \in \widetilde{K}/H_{\chi}$. Then, $t \cdot \chi$ in \widehat{N} may be written $x \cdot X$ for $x = \overline{t} \in \widetilde{K}/H_{\chi}$ because H_{χ} is the stability group of \widetilde{K} at $X \in \widehat{N}$. Since a cocycle $D^{A}(k,t) = A(t)\overline{A(t+k)}$ of $(K; \widetilde{K})$ is H_{χ} -invariant, $\widetilde{D}^{A}(k,\overline{t}) = D^{A}(k, t)$ is a cocycle of $(K; \widetilde{K}/H_{\chi})$. We note that an element of \widetilde{K} can be regarded as a cocycle of $(K; \widetilde{K}; H_{\chi})$.

Lemma 4.1.4. $U^{(\chi,A,\eta)}$ is unitarily equivalent to $U^{(\chi,A+\tilde{\eta},0)}$. It can be realized on $L^2(\tilde{K}/H_{\chi})$ as follows.

- 81 -

For $\xi(x) \in L^2(\tilde{K}/H_{\chi})$,

$$(U_{(z,k)}^{(\chi,A,\eta)}\xi)(x) = \langle z, x \cdot \chi \rangle \langle k, \eta \rangle \tilde{D}^{A}(k,x)\xi(k,x)$$
 for

 $(z,k) \in G$, where $\hat{\eta}$ is an extension of $\eta \in \hat{H}_{\chi}$ to \tilde{K} .

Proof. First we identify the representation space of $U^{(\chi,A,\eta)}$ with the space of $U^{(\chi,A+\tilde{\eta},0)}$. To do this, we have only to check that the conditions of a complex valued Borel function f on \tilde{G} to belong to the sapce of $U^{(\chi,A,\eta)}$ and the space of $U^{(\chi,A+\tilde{\eta},0)}$ are the same (see section 3.2). Indeed, for $(z',h)\in G_{\chi}$ and $(z,t)\in \tilde{G}$,

$$f((z',h)(z,t)) = L_{(z',h)}^{(\chi,\eta)*}A(t)\overline{A(t+h)}f((z,t)) = \overline{\langle z', \chi \rangle \langle h, \eta \rangle} A(t)\overline{A(t+h)}f((z,t)) = L_{(z',h)}^{(\chi,0)*}(A+\tilde{\eta})(t)\overline{(A+\tilde{\eta})(t+h)}f((z,t)).$$

By section 3.1, it is easy to see that $U^{(\chi,A+\tilde{\eta},0)}$ is realized on $L^2(\tilde{K}/H_{\chi})$ as the above form. [Q.E.D]

Proof of Theorem 4.1.3. In order to prove Theorem 4.1.3, we shall take four unitary operators W_1 , W_2 , W_3 , and W_4 in order and transform the representation space $L^2(\tilde{K})$ to the suitable one.

In the first place, we put $(W_1\xi)(t) = A(t)\xi(t)$ for $\xi(t) \in L^2(\tilde{K})$.

Next, take a Borel cross section c from \tilde{K}/H_{χ} to \tilde{K} . Then, by choosing suitable Haar measures of H_{χ} and \tilde{K}/H_{χ} , we can define a unitary operator W_2 from $L^2(\tilde{K})$ to $L^2(H_{\chi}) \otimes$

- 82 -

 $\text{L}^2(\tilde{\textbf{K}}/\text{H}_\chi)$ satisfying, for $h\in\text{H}_\chi$ and $x\in\widetilde{\textbf{K}}/\text{H}_\chi,$

 $(W_2\xi)(h, x) = \xi(h + c(x))$

for $\xi(t) \in L^2(\widetilde{K})$.

Further, we denote by W₃ a unitary operator F \otimes I from $L^{2}(H_{\chi}) \otimes L^{2}(\widehat{K}/H_{\chi})$ to $L^{2}(\widehat{H}_{\chi}) \otimes L^{2}(\widehat{K}/H_{\chi})$ where F is the Fourier transformation from $L^{2}(H_{\chi})$ to $L^{2}(\widehat{H}_{\chi})$ and I is the identity operator on $L^{2}(\widehat{K}/H_{\chi})$.

Let $\operatorname{H}^{1}_{\chi}$ denote the annihilator of H_{χ} in \widetilde{K} . Then, by Pontrjiagin's duality, $\operatorname{H}^{1}_{\chi}$ is isomorphic with $\widehat{\widetilde{K}}/\operatorname{H}^{1}_{\chi}$. Via a Borel cross section from $\widehat{\widetilde{K}}/\operatorname{H}^{1}_{\chi}$ to \widetilde{K} , we can define a Borel extension \widetilde{n} ($\eta \in \widehat{\operatorname{H}}_{\chi}$) from $\operatorname{H}^{1}_{\chi}$ to $\widehat{\widetilde{K}}$. So, we can define a unitary operator W_{4} on $\operatorname{L}^{2}(\widehat{\operatorname{H}}_{\chi}) \otimes \operatorname{L}^{2}(\widetilde{K}/\operatorname{H}_{\chi})$ by

$$(W_{l_{1}}\xi)(n,x) = \langle c(x), \hat{n} \rangle \xi(n,x)$$

for $\xi(\eta, x) \in L^2(\widehat{H}_{\chi}) \otimes L^2(\widetilde{K}/H_{\chi})$.

Now, we denote by W the operator $W_4 \circ W_3 \circ W_2 \circ W_1$, which is a unitary operator from $L^2(\tilde{K})$ to $L^2(\hat{H}_{\chi}) \otimes L^2(\tilde{K}/H_{\chi})$. After some calculations, we get, for $\tau(\eta) \otimes \xi(x) \in L^2(\hat{H}_{\chi})$ $\otimes L^2(\tilde{K}/H_{\chi})$,

$$W\lambda_{h}^{A}W^{*}; \tau(\eta)\otimes\xi(x)$$

 \longrightarrow <h, η > $\tau(\eta)\otimes\xi(x)$

and

$$W^{\chi}_{(z,k)} W^{*} : \tau(\eta) \otimes \xi(x)$$

$$\longrightarrow \tau(\eta) \otimes \langle k, \tilde{\eta} \rangle \langle z, x \cdot \chi \rangle \widetilde{D}^{A}(k,x) \xi(k,x)$$

$$= \tau(\eta) \otimes (U^{(\chi,A,\eta)}_{(z,k)} \xi)(x) \quad (\text{see Lemma 4.1.4}).$$

- 83 -

It is clear that $\{W\lambda_h^A W^*; h \in H_{\chi}\}$ generates an abelian von Neumann algebra $L^{\infty}(\hat{H}_{\chi}) \otimes \mathbb{C}_{L^2}(\tilde{K}/H_{\chi})$ which is spatially isomorphic with \mathfrak{A}^A . Hence, corresponding to this algebra, we can decompose the Hilbert space $L^2(\hat{H}_{\chi}) \otimes L^2(\tilde{K}/H_{\chi})$ to $\int_{\hat{H}_{\chi}}^{\mathfrak{G}} \ell_{\lambda}^{\eta} d\mu(\eta)$ where μ is the Haar measure of \hat{H}_{χ} and ℓ_{λ}^{η} is constantly equal to $L^2(\tilde{K}/H_{\chi})$. By this decomposition, we get,

$$\lambda_{h}^{A} \cong \int_{\hat{H}_{\chi}}^{\Phi} \langle h, n \rangle d\mu(n)$$

and

$$\pi^{\chi}(z,k) \cong \int_{\widehat{H}_{\chi}}^{\oplus} U^{(\chi,A,\eta)}_{(z,k)} d\mu(\eta) . \qquad [Q.E.D.]$$

4.2. Properties of π^{χ} and $U^{(\chi,A,\eta)}$

In this section, we investigate the properties of the representations π^{χ} and $U^{(\chi,A,\eta)}$ of $G = N \times_{S} K$. In chapter II, we studied the cohomology group of double transformation groups. For the double transformation group (K; \tilde{K} ; H_{χ}), we denote the abelian group of all T-valued cocycles of (K; \tilde{K} ; H_{χ}) by Z(K; \tilde{K} ; H_{χ}). A cocycle A is called a coboundary if there exist an H_{χ} -invariant cocycle E and K-invariant cocycle F such that A(t)=E(t)F(t) for almost all $t \in \tilde{K}$. The subgroup of all coboundaries in Z(K; \tilde{K} ; H_{χ}) is denoted by B(K; \tilde{K} ; H_{χ}). Then, we define the cohomology group H(K; \tilde{K} ; H_{χ}), as the quotient group of Z(K; \tilde{K} ; H_{χ}) by B(K; \tilde{K} ; H_{χ}). Since the group of unitary characters of \tilde{K} , namely \hat{K} , is considered as a subgroup of Z(K; \tilde{K} ; H_{χ}), we may make the factor group $\hat{K}/(\hat{R} \wedge B(K; \tilde{K}; H_{\chi})$), denoyed by $H_0(K; \tilde{K}; H_{\chi})$. We denote by

 $\tilde{H}(K; \tilde{K}; H_{\chi})$ the T-valued weak cohomology group of (K; $\tilde{K}; H_{\chi}$) which is defined by H(K; $\tilde{K}; H_{\chi})/H_0(K; \tilde{K}; H_{\chi})$. For the detail, see chapter II.

By Theorem 3.2.6 and Lemma 4.1.4, we get immediately the following proposition.

Proposition 4.2.1. If a cocycle A is not weakly cohomologous to a cocycle A', then, for any choice of η and η' in \widehat{H}_{χ} , $U^{(\chi,A,\eta)}$ is never unitarily equivalent to $U^{(\chi,A',\eta')}$.

Thus, we see that Theorem 4.1.3. gives at least as many completely different decompositions of π^{χ} as the elements of $\widetilde{H}(K; \widetilde{K}; H_{\chi})$. Note that the abelian von Neumann algebras π^{A} are mutually spatially isomorphic. The emphasis may be put on this fact, as this possibility has never been pointed out before.

In section 3.2, we studied the irreducibility of $U^{(\chi,A,\eta)}$.

Proposition 4.2.2. If $K + H_{\chi}$ is dense in \hat{K} , $U^{(\chi,A,\eta)}$ $(\eta \in \hat{H}_{\chi})$ are irreducible representations of G.

Therefore, we get the following proposition.

Proposition 4.2.3. If $K + H_{\chi}$ is dense in \tilde{K} , \Re^{A} is a maximal abelian subalgebra in $\pi^{\chi}(G)$ ' for each $A \in Z(K; \tilde{K}; H_{\chi})$.

When we take a unitary character ζ of \widetilde{K} as a cocycle of

- 85 -

(K; \tilde{K} ; H_{χ}) and we consider the representation $U^{(\chi,\zeta,0)}$, simply denoted by $U^{(\chi,\zeta)}$, we see the following lemma by Theorem 3.2.6 and Proposition 2.4.6.

Lemma 4.2.4. $U^{(\chi,\zeta)}$ is unitarily equivalent to $U^{(\chi,\zeta')}$ if and only if $\zeta' - \zeta \in K^{\perp} + H_{\chi}^{\perp}$.

Note that $U^{(\chi,\tilde{\eta})} = U^{(\chi,0,\eta)}$ for $\eta \in \widehat{H}_{\chi}$ where $\widehat{H}_{\chi} \ge \eta \rightarrow \tilde{\eta} \in \widehat{K}$ is a Borel extension map. Then, Theorem 4.1.3 asserts that

$$\begin{array}{c} \chi \\ \pi \cong \int_{\widehat{H}_{\chi}}^{\bigoplus} U^{(\chi, \widehat{\eta})} d\mu(\eta) \end{array}$$

by considering the case that a cocycle A is trivial.

Next, we define unitary representations $\tilde{\pi}^{\chi}$ and $\tilde{\tilde{\pi}}^{\chi}$ of G = N x_sK as follows.

$$\tilde{T}^{\chi} = \int_{\widehat{K}}^{\Phi} U^{(\chi, \widehat{\sigma})} d\nu(\sigma)$$

where ν is a Haar measure of \hat{K} and $\hat{K} \circ \sigma \rightarrow \hat{\sigma} \in \hat{K}$ is a Borel extension map.

$$\widehat{\pi}^{\chi} = \int_{\widehat{K}}^{\oplus} U^{(\chi,\zeta)} d\gamma(\zeta)$$

where γ is a Haar measure of $\widetilde{K}.$

Proposition 4.2.5. π^{χ} , $\tilde{\pi}^{\chi}$, and $\tilde{\pi}^{\chi}$ are mutually quasi-equivalent. Therefore, $\pi^{\chi}(G)$ ", $\tilde{\pi}^{\chi}(G)$ ", and $\tilde{\pi}^{\chi}(G)$ " are algebraically isomorphic each other.

Proof. This follows immedictely from Proposition 2.4,6 [Q.E.D] We denote by $L^{\infty}(\tilde{K}/H_{\chi}) \times_{\alpha} K$ a von Neumann algebra obtained as a crossed product of $L^{\infty}(\tilde{K}/H_{\chi})$ with K by the canonical action α of K on $L^{\infty}(\tilde{K}/H_{\chi})$. Then, we have the following.

Lemma 4.2.6. $\tilde{\pi}^{\chi}(G)''$ is spatially isomorphic with $L^{\infty}(\tilde{K}/H_{\chi}) \times_{\alpha} K$.

Proof. π^{χ} is realized on $L^{2}(K) = L^{2}(K/H_{\chi})$ as follows. For $\tau(\sigma) \otimes \xi(x) \in L^{2}(\widehat{K}) \otimes L^{2}(\widetilde{K}/H_{\chi})$,

 $\widehat{\pi}_{(z,k)}^{\chi} : \tau(\sigma) \otimes \xi(x)$ $\longrightarrow \langle k, \sigma \rangle \tau(\sigma) \otimes \langle z, x.\chi \rangle \xi(k.x)$

for $(z,k) \in G$. Let W be a unitary operator $F \otimes I$ from $L^2(\hat{K}) \otimes L^2(\tilde{K}/H_{\chi})$ to $L^2(K) \otimes L^2(\tilde{K}/H_{\chi})$ where F is the Fourier transformation from $L^2(\hat{K})$ to $L^2(K)$. Then, by simple calculations, we get, for $\rho(s) \otimes \xi(x) \in L^2(K) \otimes L^2(\tilde{K}/H_{\chi})$,

$$\begin{split} & \widetilde{\pi}^{\chi}_{(z,k)} \ \forall^{*} : \rho(s) \otimes \xi(x) \\ & \longrightarrow \rho(s+k) \otimes \langle z, x \cdot \chi \rangle \xi(k \cdot x). \end{split}$$

It is clear that the set of Ψ_z : $x \longrightarrow \langle z, x \cdot \chi \rangle (z \in N)$ generates an abelian von Neumann algebra $L^{\infty}(\tilde{K}/H_{\chi})$ on $L^2(K/H_{\chi})$. Therefore, we see that W $\tilde{\pi}^{\chi}(G)W^*$ generates $L^{\infty}(\tilde{K}/H_{\chi}) \times_{\alpha} K$.

[Q.E.D.]

Lemma 4.2.7. If K + H_{\chi} is dense in \tilde{K} and $K \wedge H_{\chi} = \{0\}$, $L^{\infty}(\tilde{K}/H_{\chi}) \times_{\alpha} K$ is a factor. Under these assumptions, $L^{\infty}(\tilde{K}/H_{\chi}) \times_{\alpha} K$ is an injective type II factor if and only if $K + H_{\chi} \neq \widetilde{K}.$

Proof. The assumptions that $K + H_{\chi}$ is dense in \tilde{K} and $K \cap H_{\chi} = \{0\}$ means that the action $\propto \text{ of } K \text{ on } \tilde{K}/H_{\chi}$ is ergodic and free. Then, it is clear that $L^{\infty}(\tilde{K}/H_{\chi}) \times_{\alpha} K$ is a factor under these assumptions [18]. Further, $K + H_{\chi} = \tilde{K}$ if and only if the action α of K on \tilde{K}/H_{χ} is transitive. Therefore, $K + H_{\chi} \neq \tilde{K}$ if and only if $L^{\infty}(\tilde{K}/H_{\chi}) \times_{\alpha} K$ is a non-type I factor. Since the measure on \tilde{K}/H_{χ} is K-invariant, $L^{\infty}(\tilde{K}/H_{\chi}) \times_{\alpha} K$ must be an injective type II factor (see [6] and [18]). [Q.E.D.]

Combining Proposition 4.2.5 with Lemma 4.2.6 and Lemma 4.2.7, we get the following theorem.

Theorem 4.2.8. $\pi^{\chi}(G)$ " is algebraically isomorphic with $L^{\infty}(\tilde{K}/H_{\chi}) \times_{\alpha} K$. If $K + H_{\chi}$ is dense in \tilde{K} and $K \cap H_{\chi} = \{0\}$, π^{χ} is a factor representation. Under these assumptions, π^{χ} is an injective type II factor representation if and only if $K + H_{\chi} \neq \tilde{K}$.

Remark 4.2.9. We are interested in the case that $K + H_{\chi}$ is dense in \tilde{K} and not equal to \tilde{K} . Under this situation, $U^{(\chi,A,\eta)}$ is a non-Mackey irreducible representation of G and π^{χ} is a non-type I representation so that G is not a type I group. Moreover, we have got the result $H(K; \tilde{K}; H_{\chi}) \supset$ $H_0(K; \tilde{K}; H_{\chi}) \cong \hat{\tilde{K}}/(K^1 + H_{\chi}^1)$ in Proposition 2.4.6, which is stronger than the general result in [38] that $H(K; \tilde{K}; H_{\chi}) \neq$ $\{0\}$. In some cases, we know that $\tilde{H}(K; \tilde{K}; H_{\chi}) \supset \mathbb{Q}$ where Q is

- 88 -

the set of rational numbers (see (a-4) and (b-4) in section 2.5).

Remark 4.2.10. By using Theorem 12.1 in [30], we can get another decomposition of π^{χ} . Let ψ be the Borel map from \tilde{K} onto $\operatorname{Orb}_{\tilde{K}}(\chi)$ in \hat{N} , defined by $\psi(t) = t \cdot \chi$. Put $V^{(\chi,t)} = \operatorname{Ind}_{N}^{G}\psi(t)$ $(t \in \tilde{K})$. Then, it is clear that $V^{(\chi,t)}$ $V^{(\chi,t')}$ if and only if $t'-t \in K + H_{\chi}$. Take a Borel cross section c from \tilde{K}/K to \tilde{K} . Then, $\tilde{K}/K \ni y \longrightarrow V^{(\chi,c(y))}$ is measurable and it is seen that

$$\chi \cong \int_{\widetilde{K}/K}^{\bigoplus} V^{(\chi,c(\lambda))} d\lambda(\lambda)$$

where v is a Haar measure of \tilde{K}/K . If $K \cap H_{\chi} = \{0\}, V^{(\chi,t)} = Ind_{N}^{G}\psi(t)$ is Mackey irreducible representation of G. If $K + H_{\chi}$ is dense in \tilde{K} and $K + H_{\chi} \neq \tilde{K}$, $V^{(\chi,t)}$ ($t \in \tilde{K}$) are never unitarily equivalent to $U^{(\chi,A,\eta)}$ for whatever $A \in Z(K; \tilde{K}; H_{\chi})$ and $\eta \in \hat{H}_{\chi}$. When, we observe these phenomena from the view point of group C^{*} - algbera of G, it is easier to understand. We will show it in the subsequent chapter.

4.3. Applications and examples

π

Here, we consider the discrete Mautner group and the Mautner group, both of which are non-regular semi-direct product groups and satisfy the condition (*) in section 4.1. We denote the additive groups of complex numbers, real numbers, rational numbers, and integers by C, R, Q, and Z, respectively. Example 4.3.1: the discrete Mautner group

Let \tilde{G} be the semi-direct product $C \times_{s} \mathbb{R}$, where the multiplications is defined by $(z,t)(z',t') = (z+e^{it}z', t+t')$ for z, z' $\in C$ and t, t' $\in \mathbb{R}$. This group \tilde{G} is the universal covering group of 3-dimensional motion group and a regular semi-direct product group. We take a closed subgroup G of \tilde{G} , given by

 $G = \{(z,n) \in \widetilde{G}; z \in \mathbb{C} \text{ and } n \in \mathbf{Z}\}.$

 $G = C \times_{S} Z$ is the discrete Mautner group.

We take a unitary character $\chi^{\mathbf{r}}$ of \mathbf{C} $(\mathbf{r} \in \mathbf{R}^{+})$, defined by $\langle \mathbf{z}, \chi^{\mathbf{r}} \rangle = e^{\mathbf{i}(\mathbf{r},\mathbf{z})}$ for $\mathbf{z} \in \mathbf{C}$, where (,) means the real inner product in \mathbf{C} . The stability group $\mathbf{H}_{\mathbf{r}}$ at $\chi^{\mathbf{r}}$ is $2\pi \mathbf{Z}$ for all $\mathbf{r} \in \mathbf{R}^{+}$. Put $\mathbf{G}_{0} = \mathbf{C} \times_{\mathbf{S}} 2\pi \mathbf{Z} = \mathbf{C} \times 2\pi \mathbf{Z}$. Then, the unitary representations $\pi^{\mathbf{r}}(\mathbf{r} \in \mathbf{R}^{+})$ and $\mathbf{U}^{(\mathbf{r},\mathbf{q},\lambda)}$ $(\mathbf{r} \in \mathbf{R}^{+}, \mathbf{q} \in \mathbf{Q}, \lambda \in [0,1))$ are as follows.

$$\pi^{\mathbf{r}} = \int_{\mathbf{G}} \operatorname{Ind}_{\mathbf{G}}^{\widetilde{\mathbf{G}}} \chi^{\mathbf{r}}$$
$$U^{(\mathbf{r},\mathbf{q},\lambda)} = \operatorname{Ind}_{\mathbf{G}_{0}}^{\mathbf{G}} (\mathbf{A}^{\mathbf{q}}; \chi^{\mathbf{r}_{\mathbf{x}}} \eta^{\lambda}) \quad (\text{See chapter III})$$

where $\eta^{\lambda}(\lambda \in [0, 1))$ are unitary characters of $2\pi \mathbf{Z}$, defined by $\eta^{\lambda}(2\pi n) = e^{2\pi i \lambda n}$ and A^{q} ($q \in \mathbb{Q}$) are cocycles of ($\mathbf{Z}; \mathbf{R}; 2\pi \mathbf{Z}$) given in 2.5.6. We know in section 2.5 that A^{q} is weakly cohomologous to $A^{q'}$ if and only if q = q'.

Now, we get, by Theorem 4.1.3,

$$\pi^{\mathbf{r}} \cong \int_{0}^{1} \mathbb{U}^{(\mathbf{r},\mathbf{q},\lambda)} d\mu(\lambda) \quad \text{for each } \mathbf{q} \in \mathbb{Q},$$

where μ is the Lebesgue measure of [0, 1). Since $\mathbf{Z} + 2\pi \mathbf{Z}$

- 90 -

is dense in R, $U^{(r,q,\lambda)}$ are irreducible representations of G by Theorem 3.2.6. If $q \neq q'$, $U^{(r,q,\lambda)}$ is never unitarily equivalent to $U^{(r,q',\lambda')}$ for any λ , $\lambda' \in [0, 1)$ by Proposition 4.2.1. Moreover, $\pi^{r}(G)$ " is algebraically isomorphic with $L^{\infty}(\mathbf{T}) \times_{\alpha} \mathbf{Z}$, so that π^{r} is an injective type II₁ factor representation by Theorem 4.2.8.

Example 4.3.2: the Mautner group

Let G denote the Mautner group which is the semi-direct product group ${\rm C}^2\times_{\rm s}R$ with the multiplication

for $z, z', w, w' \in C$ and $t, t' \in R$.

Associated with this G, we take a 6-dimensional algebraic solvable Lie group \tilde{G} which is the semi-direct product $C^2 \times_{s} R^2$ with the multiplication

 $(z,w,u,v)(z',w',u',v') = (z+e^{iu}z', w+e^{iv}w, u+u', v+v').$ G is regarded as a closed subgroup of \tilde{G} by the imbedding

$$G \ni (z, w, t) \longrightarrow (z, w, t, 2\pi t) \in \widetilde{G}.$$

We take a unitary character $\chi^{(r,s)}$ of \mathbb{C}^2 $(r,s \in \mathbb{R}^+)$, defined by

$$\langle (z,w), \chi^{(r,s)} \rangle = e^{i(r,z)} e^{i(s,w)}$$

for $(z,w) \in \mathbb{C}^2$. The stability group $H^{(r,s)}$ of \mathbb{R}^2 at $\chi^{(r,s)}$ is $(2\pi \mathbf{Z})^2$ for all r, $s \in \mathbb{R}^+$. Put $G_0 = \mathbb{C}^2 \times (2\pi \mathbf{Z})^2 = \mathbb{C}^2 \times (2\pi \mathbf{Z})^2$. Then the unitary representations $\pi^{(r,s)}$ and $U^{(r,s,q,\lambda,\omega)}$ $(\mathbf{r}, \mathbf{s} \in \mathbb{R}^{+}, \mathbf{q} \in \mathbb{Q}, \lambda, \omega \in \mathbb{R}) \text{ are as follows.}$ $\pi^{(\mathbf{r}, \mathbf{s})} = \int_{G} |\operatorname{Ind}_{\mathbb{C}^{2}}^{\widetilde{G}} \chi^{(\mathbf{r}, \mathbf{s})}$ $U^{(\mathbf{r}, \mathbf{s}, \mathbf{q}, \lambda, \omega)} = \operatorname{Ind}_{G_{0}}^{G} (\mathbb{A}^{q}; \chi^{(\mathbf{r}, \mathbf{s})} \times \eta^{(\lambda, \omega)})$

where unitary characters $\eta^{(\lambda,\omega)}$ of $(2\pi Z)^2$ $(\lambda,\omega \in \mathbb{R})$ are defined by

$$\langle (2\pi m, 2\pi n), \eta^{(\lambda,\omega)} \rangle = e^{2\pi i \lambda m} e^{2\pi i \omega n}$$

for $(2\pi m, 2\pi n) \in (2\pi \mathbf{Z})^2$ and A^q $(q \in \mathbb{Q})$ are cocycles of (R, $\mathbb{R}^2; (2\pi \mathbf{Z})^2$) constructed in section 2.5. We note that A^q is weakly cohomologous to A^q' if and only if q=q'.

Theorem 4.1.3 asserts that

$$\pi^{(\mathbf{r},\mathbf{s})} \cong \int_{0}^{1} \int_{0}^{1} \oplus_{0} U^{(\mathbf{r},\mathbf{s},\mathbf{q},\lambda,\omega)} \mathbf{d}\mu(\lambda) \, d\mu(\omega)$$

for each $q \in \mathbb{Q}$, where μ is the Lebesgue measure of the interval [0, 1]. Since $\mathbb{R} + (2\pi \mathbb{Z})^2$ is dense in \mathbb{R}^2 , identifying the subgroup $\{(t, 2\pi t) \in \mathbb{R}^2; t \in \mathbb{R}\}$ with \mathbb{R} , $U^{(r,s,q,\lambda,\omega)}$ are irreducible representations of G by Theorem 3.2.6. If $q \neq q'$, $U^{(r,s,q,\lambda,\omega)}$ are never unitarily equivalent to $U^{(r,s,q',\lambda',\omega')}$ for arbitrary choice of $\lambda, \omega, \lambda'$, and $\omega' \in \mathbb{R}$ by Proposition 4.2.1. Therefore, the above decomposition of $\pi^{(r,s)}$ means that there are at least as many completely different irreducible decompositions of $\pi^{(r,s)}$ as the elements of \mathbb{Q} . Moreover, $\pi^{(r,s)}(G)$ " is algebraically isomorphic with $L^{\infty}(\mathbb{T}^2) \times_{\beta} \mathbb{R}$ and so $\pi^{(r,s)}$ is an injective type II_{∞} factor representation of G by Theorem 4.2.8.

Chapter V. Representations of C^{*}-crossed products

This chapter is devoted to study representations of certain C^{*}-crossed products. In the previous chapters, we investigated decompositions of representations of some non-regular semi-direct product groups, related with a kind of cohomology group. In this chapter, we extend those results from a view point of operator algebras and show that the variety of decompositions of a non-type I representation is connected not only with ergodic measures but also with the cohomology group.

For two closed subgroups H and K of a locally compact abelian group G, we get a C*-crossed product A = $C_0(G/H) \times_{\gamma} K$. We investigate decompositions of a certain representation π^0 of A. In section 5.1, we find abelian von Neumann subalgebras π^{a} and α^{b} in the commuting algebra $\pi^{0}(A)$ associated with the automorphisms α^a and β^b of $\pi^0(A)'$ where $a \in \mathbb{Z}(K; G; H)$ and $b \in Z(H^{\perp}; \hat{G}; K^{\perp})$. We will have also some necessary and sufficient conditions of the maximality of π^{a} and β^{b} in $\pi^{0}(A)'$. In section 5.2, we study decompositions of π^0 corresponding to π^a . To do this, we study generalized induced representations of C^{*}-crossed products following to chapter III. In section 5.3, using the concept of Heisenberg representations, we write down decompositions of π^0 corresponding to the above \mathbb{A}^a and \mathbb{B}^b in explicit forms. In section 5.4, we show some examples and an application to unitary representations of a certain locally compact group.

5.1. Abelian subalgebras in $\pi^0(A)$

Let G be a locally compact second countable "abelian" group. For a closed subgroup H of G, we can consider two von Neumann algebras $\mathfrak{N}(H)$ and $\mathfrak{G}(G/H)$ on the Hilbert space $L^2(G)$ of all square summable measurable functions with respect to a Haar measure of G. $\mathfrak{N}(H)$ is the von Neumann algebra generated by $\{U_h; h \in H\}$ where U is the regular representation of G. $\mathfrak{G}(G/H)$ is the von Neumann algebra generated by the multiplication operators P(f) on $L^2(G)$ defined by functions of $L^{\infty}(G)^H$, the set of all H-invariant essentially bounded measurable functions of G.

For a family of closed subgroups $\{H_i ; i \in I\}$ of G, we denote by $\bigvee H_i$ the closed subgroup of G generated by $i \in I$ $\bigcup H_i$. Similarly, for a family of von Neumann algebras $\{m_i ; i \in I\}$ on $L^2(G)$, we denote by $\bigvee \mathcal{M}_i$ the von Neumann $i \in I$ algebra generated by $\bigcup \mathcal{M}_i$. The following fact of lattice $i \in I$ correspondence about $\mathcal{N}(H)$ and $\mathcal{B}(G/H)$ was obtained by M. Takesaki and N. Tatsuuma.

Lemma 5.1.1. (Theorem 4 in [53])

If {H_i; i \in I} is a family of closed subgroups of G, then

(i) $\bigvee \mathfrak{N}(H_{i}) = \mathfrak{N}(\bigvee H_{i})$ i $\in I$ i $\in I$ i i $\in I$ (ii) $\bigcap \mathfrak{N}(H_{i}) = \mathfrak{N}(\bigcap H_{i})$ i $\in I$ i $\in I$ i $\in I$ (i)' $\bigvee \mathfrak{B}(G/H_{i}) = \mathfrak{B}(G/\bigcap H_{i})$ i $\in I$ i $\in I$ i $\in I$ (ii)' $\bigcap \mathfrak{B}(G/H_{i}) = \mathfrak{B}(G/\bigvee H_{i})$.

- 94 -

For two arbitrary closed subgroups H and K of G, we denote by $\mathcal{M}(G/K, H)$ the von Neumann algebra generated by $\mathcal{N}(H)$ and $\mathcal{B}(G/K)$. Then, the following generalized commutation relation holds.

Lemma 5.1.2. (M. Takesaki [52])
$$\mathcal{M}(G/K, H)' = \mathcal{M}(G/H, K).$$

We can generalize the result of Lemma 5.1.1 as follows.

Proposition 5.1.3. Let $\{H_i; i \in I\}$ and $\{K_i; i \in I\}$ be families of closed subgroups of G, then

(i) $\bigvee \mathcal{M}(G/K_{i},H_{i}) = \mathcal{M}(G/\bigcap K_{i}, \bigvee H_{i})$ $i \in \mathbb{I}$ $i \in \mathbb{I}$ $i \in \mathbb{I}$

(ii)
$$\bigcap_{i \in I} \mathfrak{M}(G/K_i, H_i) = \mathfrak{M}(G/\bigvee K_i, \bigcap H_i)$$
$$i \in I \qquad i \in I \qquad i \in I$$

Proof. This follows immediately from Lemma 5.1.1 and Lemma 5.1.2 by simple calculations. [Q.E.D.]

Corollary 5.1.4. Let H and K be closed subgroups of G. Then,

(i) $\mathfrak{M}(G/K, H)$ is a factor if and only if $K \cap H = \{0\}$ and K + H is dense in G.

(ii) $\mathcal{M}(H)$ is a maximal abelian von Neumann subalgebra in $\mathcal{M}(G/K, H)$ if and only if K + H is dense in G.

(iii) $\mathcal{B}(G/K)$ is a maximal abelian von Neumann subalgebra in $\mathcal{R}(G/K, H)$ if and only if $K \wedge H = \{0\}$. Proof.

(i) $\mathcal{M}(G/K, H)$ is a factor. $\iff \mathfrak{m}(G/K,H) \cap \mathfrak{m}(G/K,H)' = \mathfrak{c}_{L^2(G)}$ $\iff \mathfrak{M}(G/K,H) \cap \mathfrak{M}(G/H,K) = \mathfrak{C}_{L^2(G)}$ $\iff \mathfrak{M}(G/H \vee K, H \cap K) = \mathfrak{C}_{L^2(G)}$ $\iff H \lor K = G \text{ and } H \land K = \{0\}.$ (ii) $\mathcal{O}(H)$ is maximal abelian in $\mathcal{M}(G/K, H)$ $\iff \pi(H) \cap \mathcal{M}(G/K, H) = \theta(H)$ $\iff \mathfrak{M}(G/G,H) \cap \mathfrak{M}(G/K,H) = \mathfrak{M}(G/G,H)$ $\iff \mathfrak{m}(G/H,G) \cap \mathfrak{m}(G/K,H) = \mathfrak{m}(G/G,H)$ $\iff \mathcal{M}(G/H \lor K, H) = \mathcal{M}(G/G, H)$ \iff HVK = G (iii) $\mathcal{O}(G/K)$ is maximal abelian in $\mathcal{M}(G/K, H)$ $\iff \mathcal{B}(G/K) \cap \mathcal{R}(G/K,H) = \mathcal{B}(G/K)$ $\iff m_2(G/K, \{0\}) \cap m_2(G/K, H) = m_2(G/K, \{0\})$ $\iff \mathfrak{M}(G/\{0\}, K) \land \mathfrak{M}(G/K, H) = \mathfrak{M}(G/K, \{0\})$ $\iff \mathfrak{M}(G/K, K \cap H) = \mathfrak{M}(G/K, \{0\})$ \iff K \cap H = {0} [Q.E.D.]

For two closed subgroups H and K of G, we are interested in the following case (*).

(*) $H \cap K = \{0\}$ and H + K is dense in G.

Under the condition (*), we see that $\mathcal{M}(G/K,H)$ is a factor and that $\mathfrak{N}(H)$ and $\mathfrak{G}(G/K)$ are maximal abelian von Neumann subalgebras in $\mathfrak{M}(G/K,H)$. Moreover, under the condition (*), we note that $\mathfrak{M}(G/K,H)$ is an injective type II factor if and only if K + H \neq G. The next examples satisfy such conditions.

Example 5.1.5. Let Z and R denote the additive groups of integers and real numbers respectively and let $^{\theta}$ be a positive irrational number.

(i) G = R, $H = \theta Z$, and K = Z. (ii) $G = R^2$, $H = Z^2$, and $K = \{(x, y) \in R^2 ; y = \theta x\}$.

For two closed subgroups H and K of G, we get a double transformation group (K;G;H) as defined in section 2.2. Let T be the torus group of the complex numbers with absolute value 1. When a T-valued Borel function a(g) on G satisfies the cocycle condition;

 $a(k + g + h) = a(k + g)\overline{a(g)} a(g + h)$

for each $k \in K$, $g \in G$, and $h \in H$, a(g) is called a cocycle of (K; G; H) and the abelian group of all such cocycles is denoted by Z(K; G; H). A cocycle a_1 is said to be cohomologous to a cocycle a_2 if there exist a K-invariant cocycle e_1 and an H-invariant cocycle e_2 such that $a_1(g)\overline{a_2(g)}$ = $e_1(g)e_2(g)$ for almost all $g \in G$. Further, a_1 is said to be weakly cohomologous to a_2 if $a_1\overline{a_2}$ is cohomologous to some unitary character of G. We denote by H(K; G; H) the cohomology group of all cohomologous classes of cocycles of (K;G;H) and by $\widetilde{H}(K; G; H)$ the factor group of all weakly cohomologous classes of such cocycles.

For a cocycle a in Z(K; G; H), we define an operator $\rho(a)$ on $L^2(G)$ by

 $(\rho(a)\xi)(g) = a(g)\xi(g)$ for $\xi(g) \in L^2(G)$.

- 97 -

For a bounded operator S on $L^{2}(G)$, we put

$$\alpha^{a}(S) = \rho(a) * S \rho(a)$$

Then, α^{a} is an automorphism of the full operator algebra $\mathcal{L}(L^{2}(G))$. For an operator $\rho(f)$ of $\mathcal{B}(G/K)$, we get

$$\alpha^{a}(\rho(f)) = \rho(a)*\rho(f)\rho(a)$$
$$= \rho(\overline{a} f a)$$
$$= \rho(f).$$

Thus α^{a} leaves each operator of $\mathcal{B}(G/K)$ invariant and $\alpha^{a}(\mathcal{B}(G/K)) \subset \mathcal{M}(G/K,H)$.

Next, we show $\alpha^{a}(\mathcal{N}(H)) \subset \mathcal{M}(G/K,H)$. Let U_{h} (h \in H) be a generator of $\mathcal{N}(H)$. Then, $\alpha^{a}(U_{h}) \in \mathcal{B}(G/H)'$ and $\alpha^{a}(U_{h}) \in \mathcal{N}(K)'$. Indeed, on the one hand, for an operator $\rho(f) \in \mathcal{B}(G/H)$ and $\xi(g) \in L^{2}(G)$,

$$(\alpha^{a}(U_{h})_{\rho}(f)\xi)(g) = (\rho(a)^{*}U_{h}^{\rho}(a)\rho(f)\xi)(g) = \overline{a(g)}a(g + h)f(g + h)\xi(g + h) = f(g)\overline{a(g)}a(g + h)\xi(g + h) = (\rho(f)\rho(a)^{*}U_{h}^{\rho}(a)\xi)(g) = (\rho(f)\alpha^{a}(U_{h}^{\rho})\xi)(g).$$

On the other hand, for a generator U_k (k \in K) of $\mathcal{O}(K)$ and an arbitrary $\xi(g) \in L^2(G)$,

$$(U_{k}\alpha^{a}(U_{h})\xi)(g) = \overline{a(g+k)}a(g + k + h)\xi(g + k + h)$$

- 98 -

$$= \overline{a(g)}a(g + h)\xi(g + k + h)$$
$$= (\alpha^{a}(U_{h})U_{k}\xi)(g).$$

Therefore, we get

$$\alpha^{a}(U_{h}) \in \mathcal{B}(G/H) \land \mathcal{M}(K)'$$

$$= (\mathcal{B}(G/H) \lor \mathcal{M}(K))$$

$$= \mathcal{M}(G/H, K)'$$

$$= \mathcal{M}(G/K, H).$$

Hence, $\alpha^{a}(\mathfrak{N}(H)) \subset \mathfrak{M}(G/K, H)$. We see now $\mathfrak{M}(G/K,H)$ is α^{a} -invariant and so α^{a} gives an automorphism of $\mathfrak{M}(G/K,H)$.

Now, we define an automorphism β^{b} of $\mathfrak{M}(G/K,H)$ for $b \in \mathbb{Z}(H^{\perp}; \widehat{G}; K^{\perp})$ as follows, where \widehat{G} is the dual group of G and H^{\perp} and K^{\perp} are the annihilators of H and K in \widehat{G} . Take the Fourier transformation F from $L^{2}(G)$ onto $L^{2}(\widehat{G})$ and, for a bounded operator S on $L^{2}(G)$, put

 $\tilde{F}(S) = F \circ S \circ F$.

Then, $\tilde{F}(S)$ is a bounded operator on $L^2(\hat{G})$. Through this \tilde{F} , $\mathfrak{N}(H)$ is transformed onto $\mathcal{B}(\hat{G}/H^{\perp})$ and $\mathcal{B}(G/K)$ is transformed onto $\mathfrak{N}(K^{\perp})$ so that $\mathfrak{M}(G/K,H)$ is transformed onto $\mathfrak{M}(\hat{G}/H^{\perp},K^{\perp})$. Similarly as in the case α^a , a cocycle b in $Z(H^{\perp}; \hat{G}; K^{\perp})$ gives rise to an automorphism $\tilde{\beta}^b$ of $\mathfrak{M}(\hat{G}/H^{\perp}, K^{\perp})$ which leaves each operator of $\mathcal{B}(\hat{G}/H^{\perp})$ invariant. Put

$$\beta^{b} = \tilde{F}^{-1} \circ \tilde{\beta}^{b} \circ \tilde{F}.$$

Then, β^{b} is an automorphism of $\mathcal{M}(G/K,H)$ which leaves each operator of $\mathfrak{N}(H)$ invariant. Then we get the following theorem.

Theorem 5.1.6. For a cocycle a in Z(K; G; H), α^{a} is an automorphism of $\mathcal{M}(G/K, H)$ which leaves each operator of $\mathcal{B}(G/K)$ invariant. A cocycle b in Z(H[⊥]; \widehat{G} ; K[⊥]) also gives rise to an automorphism β^{b} of $\mathcal{M}(G/K, H)$ which leaves each operator of $\mathfrak{A}(H)$ invariant. The correspondences $a \rightarrow \alpha^{a}$ and $b \rightarrow \beta^{b}$ are isomorphisms from Z(K; G; H) and Z(H[⊥]; \widehat{G} ; K[⊥]) into Aut $\mathcal{M}(G/K, H)$ of all automorphisms of $\mathcal{M}(G/K, H)$ and satisfy $\alpha^{a} \circ \beta^{b} = \beta^{b} \circ \alpha^{a}$.

Proof. The latter properties about α^a and β^b follow immediately from their definitions. [Q.E.D.]

Put $alpha^{a} = \alpha^{a}(\mathfrak{N}(H))$ and $alpha^{b} = \beta^{b}(\mathfrak{B}(G/K))$ for the automorphisms α^{a} and β^{b} ($a \in Z(K; G; H)$) and $b \in Z(H^{\perp}; \widehat{G}; K^{\perp})$) of $\mathfrak{M}(G/K, H)$. Then, we get abelian von Neumann subalgebras \mathfrak{R}^{a} and \mathfrak{B}^{b} in $\mathfrak{M}(G/K, H)$ and if $a_{1} \neq a_{2}$ ($b_{1} \neq b_{2}$), $\mathfrak{A}^{a_{1}} \neq \mathfrak{A}^{a_{2}}$ ($\mathfrak{B}^{b_{1}} \neq \mathfrak{B}^{b_{2}}$).

Corollary 5.1.7. If K + H is dense in G, each \mathcal{A}^{a} (a $\in Z(K; G; H)$) is a maximal abelian subalgebra in $\mathcal{M}(G/K, H)$. If $K \cap H = \{0\}$, each \mathcal{B}^{b} (b $\in Z(H^{i}; G; K^{i})$) is a maximal abelian subalgebra in $\mathcal{M}(G/K, H)$.

Proof. This follows immediately from Corollary 5.1.4 and Theorem 5.1.6. [Q.E.D.]

For a locally compact abelian group G, take and fix two closed subgroups H and K of G. Let $C_0(G/H)$ denote the abelian C*-algebra of all continuous functions on the locally compact homogeneous space G/H vanishing at infinity. Then,
the canonical action of K on G/H induces an action of K on the C*-algebra $C_0(G/H)$ as automorphisms and we denote such an action by Y. From the C*-dynamical system ($C_0(G/H), K, Y$), we get the C*-crossed product $C_0(G/H) \times_{\gamma} K$ (see [40]) which we will denote by A(G/H,K) or abbreviatedly by A, hereafter.

Let ρ^0 be a representation of $C_0(G/H)$ on $L^2(G)$ defined by

$$(\rho^{0}(f)\xi)(g) = f(g)\xi(g)$$

for $f \in C_0(G/H)$ and $\xi \in L^2(G)$, where $G \ni g \longrightarrow g \in G/H$ is the canonical projection. Next, we define a unitary representation U^0 of K on $L^2(G)$ by

$$(U_{l_{r}}^{U}\xi)(g) = \xi(g + k)$$

for k K.

Then, it is clear that (ρ^0, U^0) is a convariant representation of the C^{*}-dynamical system $(C_0(G/H), K, \gamma)$. We denote by π^0 the representation of the C^{*}-algebra A naturally defined from this convariant representation (ρ^0, U^0) [11]. Then, we get the following.

Proposition 5.1.8.

(i) $\pi^{0}(A)' = \mathcal{M}(G/K, H)$ and $\pi^{0}(A)'' = \mathcal{M}(G/H, K)$

(ii) Under the condition (*), π^0 is a factor representation of A and there exist maximal abelian subalgebras \Re^a and \Re^b in $\pi^0(A)'$ parametrized by $a \in Z(K;G;H)$ and $b \in Z(H^+;\hat{G};k^+)$.

(iii) \int_{Π}^{a} are spatially isomorphic with each other and algebraically isomorphic with $L^{\infty}(\hat{H}, \mu)$, where μ is a Haar measure of \hat{H} . \mathcal{B}^{b} are spatially isomorphic with each other

and algebraically isomorphic with $L^{\infty}(G/K, v)$, where v is a Haar measure of the factor group G/K.

Due to Proposition 5.1.8, we know that under the condition (*) the representation π^0 of A is decomposed into a direct integral of irreducible representations corresponding to each maximal abelian subalgebra \Re^a and \mathscr{G}^b in $\pi^0(A)'$. In section 5.2, we shall describe decompositions of π^0 corresponding to \Re^a . In section 5. 3, we shall give an explicit form of the irreducible representations of A which appear as components of decompositions of π^0 corresponding to \mathscr{I}^a and \mathscr{B}^b .

5.2. Decompositions of π^0

In this section, we study generalized induced representations of C^{*}-crossed products following to chapter III, in order to state decompositions of π^0 . The results described here were obtained by suggestions of T. Kajiwara [27].

Let G be a locally compact second countable abelian group and (B, G, α) be a C*-dynamical system. We take two closed subgroups K and H of G and denote by α^{K} and α^{H} of the automorphism α considered for the elements of K and H. Let (λ , V) be a convariant representation on $\frac{1}{K}$ of the C*-dynamical system (B, H, α^{H}). Put $\mathcal{E} = \lambda(B)' \cap V(H)'$ and we denote by \mathcal{C}^{u} the unitary group of \mathcal{E} . Let a(g) be a \mathcal{C}^{u} valued cocycle of the double transformation group (K; G; H). The definition of cocycles and the cohomologous relation among them are quite similar to that we have given for Tvalued case. For the precise definition, see chapter II. We define a covariant representation of the C*-dynamical system (B, K, α^{K}) starting from (λ , V) and a(g). This definition is analogous to the group case described in chapter III.

We denote by ξ the set of all ξ 's satisfying the following conditions.

(1) ξ is an f_0 -valued Borel function on G

(2) $\xi(g + h) = V_h^*\xi(g)$ for every $g \in G$ and $h \in H$.

(3) $\int_{G/H} ||\xi(g)||^2 d_{\nu}(\dot{g})_{<\infty}$ where ν is a Haar measure of the factor group G/H.

Then, $\oint_{\mathcal{T}}$ has a Hilbert space structure with the norm defined by

$$\|\xi\| = \left(\int_{G/H} \|\xi(g)\|^2 d\nu(\dot{g}) \right)^{1/2}$$

For each $k \in K$, we define an operator U_k on β by

$$(U_{k}\xi)(g) = a(g)a(g + k)*\xi(g + k)$$

for each $\xi \in f_{\xi}$ and $g \in G$. By the fact that

$$(U_{k}\xi)(g + h) = a(g + h)a(g + k + h)*\xi(g + k + h)$$

= a(g)a(g + k)*V_{h}^{*}\xi(g + k)
= V_{h}^{*}a(g)a(g + k)*\xi(g + k)
= V_{h}^{*}(U_{k}\xi)(g)

for every $\xi \in \xi$ and $h \in H$, $U_k \xi$ belongs to ξ . It is clear that

- 103 -

 U_k is a unitary operator on ζ . Next, for each $y \in B$, we define an operator $\rho(y)$ on ζ by

$$(\rho(y)\xi)(g) = \lambda(\alpha_{\sigma}(y))\xi(g)$$

for each $\xi \in \mathcal{L}_{z}$ and $g \in G$. By the fact that

$$(\rho(y)\xi)(g + h) = \lambda(\alpha_{g+h}(y))\xi(g + h)$$
$$= V_h^*\lambda(\alpha_g(y)) V_h V_h^*\xi(g)$$
$$= V_h^* (\rho(y)\xi)(g)$$

for every $\xi \in \xi$ and $h \in H$, $\rho(y)\xi$ belongs to ξ . Clearly, $\rho(y)$ is a bounded operator on ξ .

Proposition 5.2.1. (ρ ,U) is a convariant representation on ξ of the C*-dynamical system (B, K, α^k).

Proof. This follows immediately by routine arguments and simple calculations. [Q.E.D.]

This construction is a generalization of Takesaki's induced convariant representation studied in [51]. We denote the above (ρ , U) by $\operatorname{Ind}_{H}^{K}$ (a; λ ,V). When K = G and a is trivial, $\operatorname{Ind}_{H}^{K}$ (a; λ , V) coincides with the ordinary induced representation $\operatorname{Ind}_{H}^{G}$ (λ , V).

Remark 5.2.2. Our construction is closely related to Ramsay's induced representation [44] but in some cases ours affords finer informations.

We shall develop some general theory of these representations.

Lemma 5.2.3. For two cocycles a_1 and a_2 of (K; G; H), if a_1 is cohomologous to a_2 , $Ind_H^K(a_1; \lambda, V) \cong Ind_H^K(a_2; \lambda, V)$. Proof. This follows immediately from the definition. [Q.E.D]

Let H_1 and H_2 be two closed subgroups of H such that $H_1 C H_2$. We assume henceforth that cocycles of (K; G; H) are all **T**-valued. Take a **T**-valued cocycle a, then we can regard this a as a cocycle of (K; G; H_1) (i = 1, 2). Let (λ ,V) be a covariant representation of (B, H_1 , α^{H_1}). Then, we get the following stage theorem.

Proposition 5.2.4. (Stage Theorem)

 $\operatorname{Ind}_{H_{1}}^{K}(a; \lambda, V) \cong \operatorname{Ind}_{H_{2}}^{K}(a; \operatorname{Ind}_{H_{1}}^{H_{2}}(\lambda, V)).$

Proof. The proof is carried out by an analogy of the proof of the stage theorem about ordinary induced representations by G.W. Mackey (Theorem 4.1 in [30]). We omit the detail. [Q.E.D.]

Remark 5.2.5. This is not the same with the stage theorem of A. Ramsay [44], because H, H_1 , and H_2 are not necessarily subobjects of K.

Next, we shall describe the subgroup theorem. Let K_1 be a closed subgroup of K. We denote by G_1 the closed subgroup generated by K_1 + H in G. A **T**-valued cocycle a of (K; G; H) can be regarded as a cocycle of (K_1 ; G_1 ; H) by the restriction to G_1 . Let (λ , V) be a covariant representation

- 105 -

of (B, H, α^{H}). There exists a canonical action of G on Z(K₁; G₁; H) of all **T**-valued cocycles of (K₁; G₁; H) and on Rep B of all representations of B defined, for each g ϵ G, by

$$(g \cdot a)(g') = a(g + g')$$
 for all $g' \in G_1$

and

$$(g \cdot \lambda)(y) = \lambda(\alpha_{\sigma}(y))$$
 for all $y \in B$.

Let $G/G_1 \ni t \longrightarrow \tilde{t} \in G$ be a Borel cross section from G/G_1 to G and v be a Haar measure of G/G_1 . Then, we get the following.

Proposition 5.2.6. (Subgroup Theorem)

$$(\mathbf{B},\mathbf{K}_{1},\boldsymbol{\alpha}^{\mathbf{K}_{1}})\left|\operatorname{Ind}_{\mathbf{H}}^{\mathbf{K}}(\mathbf{a};\boldsymbol{\lambda},\mathbf{V})\right| \cong \int_{\mathbf{G}/\mathbf{G}_{1}}^{\bigoplus} \operatorname{Ind}_{\mathbf{H}}^{\mathbf{K}_{1}}(\hat{\mathbf{t}}\cdot\mathbf{a};\hat{\mathbf{t}}\cdot\boldsymbol{\lambda},\mathbf{V})d\mathbf{v}(\mathbf{t})\right|$$

Proof. This also follows by the modifications of the proof of the subgroup theorem of G.W. Mackey in [30]. [Q.E.D.]

Let a be a cocycle of (K; G; H) again. Suppose that a covariant representation (λ , V) of (B, H, α^{H}) is decomposed as follows.

$$(\lambda, V) \cong \int_{Z}^{\oplus} (\lambda^{\zeta}, V^{\zeta}) d_{\nu}(\zeta)$$

on some measure space (Z, \vee) . Then, we have the following proposition by routine arguments.

Proposition 5.2.7. The field $\{Z \ni \zeta \longrightarrow \operatorname{Ind}_{H}^{K}(a; \lambda^{\zeta}, V^{\zeta})\}$

- 106 -

is v-measurable and

$$\operatorname{Ind}_{\mathrm{H}}^{\mathrm{K}}(\mathrm{a}; \lambda, \mathbb{V}) \cong \int_{\mathrm{Z}}^{\bigoplus} \operatorname{Ind}_{\mathrm{H}}^{\mathrm{K}}(\mathrm{a}; \lambda^{\zeta}, \mathbb{V}^{\zeta}) d_{\mathrm{V}}(\zeta).$$

Now, we shall consider the decompositions of the representation π^0 given in section 5.1. For a locally compact second countable abelian group G, take and fix two closed subgroups H and K of G. Recall that we have denoted by A the C^{*}-crossed product C₀(G/H)_× K for the C^{*}-dynamical system (C₀(G/H), K, Y).

Let (1, I) be the covariant representation of the C*dynamical system (C₀(G/H), H, γ^{H}) given by 1(f) = f(0), I(h) = 1, for every f in C₀(G/H) and every h in H. For a Z(K; G; H), We get the covariant representation (ρ^{a} , U^a) of (C₀(G/H), K, γ) by

 $(\rho^{a}, U^{a}) = Ind_{\{0\}}^{K}(a; \iota, I)$

where we regarded a as a cocycle of (K; G; $\{0\}$). It is clear that all cocycles of (K; G; $\{0\}$) are cohomologous to the trivial one. Therefore, by Lemma 5.2.3, we see that

$$(\rho^{a}, U^{a}) \cong (\rho^{0}, U^{0})$$

for all a \in Z(K; G; H), here the superscript 0 stands for the trivial cocycle. These covariant representations define the class of representations of π^0 of A which was just defined in section 5.1.

Let (ι, ζ) be the one-dimensional covariant representation of $(C_0(G/H), H, \gamma^H)$ where ζ is in \widehat{H} (the dual group of H). For a $\in Z(K; G; H)$, we have a covariant representaion

- 107 -

 $(\rho^{(a,\zeta)}, U^{(a,\zeta)}) = \operatorname{Ind}_{H}^{K}(a; \iota, \zeta) \text{ of } (C_{0}^{(G/H)}, K, \gamma).$ This $(\rho^{(a,\zeta)}, U^{(a,\zeta)})$ defines the representation $\pi^{(a,\zeta)}$ of A [11].

For a $\in Z(K; G; H)$, we have got the abelian subalgebra $\mathfrak{o}_{\mathbf{C}}^{\mathbf{A}}$ in $\pi^{\mathbf{O}}(\mathbf{A})'$ described in section 5.1. Let μ be a Haar measure of $\widehat{\mathbf{H}}$. Under these preparations, we get the following main theorem.

Theorem 5.2.8. Corresponding to each Π^a , the representation π^0 of A is decomposed as follows.

$$\pi^{0} \cong \int_{\widehat{H}}^{\bigoplus} (a, \zeta) d\mu(\zeta).$$

Proof. We consider the covariant representations (ρ^0, U^0) and $(\rho^{(a, \zeta)}, U^{(a, \zeta)})$ of $(C_0(G/H), K, \gamma)$ which defined π^0 and $\pi^{(a, \zeta)}$ respectively.

$$(\rho^{0}, U^{0})$$

$$\cong (\rho^{a}, U^{a})$$

$$= \operatorname{Ind}_{\{0\}}^{K}(a; 1, I)$$

$$\cong \operatorname{Ind}_{H}^{K}(a; \operatorname{Ind}_{\{0\}}^{H}(1, I)) \text{ by Proposition 5.2.4}$$

$$\cong \operatorname{Ind}_{H}^{K}(a; \int_{\widehat{H}}^{\bigoplus}(1, \zeta) d\mu(\zeta))$$

$$\triangleq \int_{\widehat{H}}^{\bigoplus} \operatorname{Ind}_{H}^{K}(a; 1, \zeta) d\mu(\zeta) \text{ by Proposition 5.2.7}$$

$$= \int_{H}^{\bigoplus} (\rho^{(a, \zeta)}, U^{(a, \zeta)}) d\mu(\zeta).$$

Hence we get

$$\pi^{0} \cong \int_{H}^{\bigoplus} \pi^{(a,\zeta)} d\mu(\zeta)$$

It is not difficult to verify that the subalgebra σc^{a} is

- 108 -

transformed to $\mathcal{M}(H)$ by the unitary operator which gives the unitary equivalence between (ρ^0 , U^0) and (ρ^a , U^a) so that $\mathcal{M}(H)$ becomes the diagonal algebra of the decomposition. [Q.E.D.]

Now, we shall describe the properties of the family of $\{\pi^{(a,\zeta)}; a \in Z(K; G; H) \text{ and } \zeta \in \widehat{H}\}$. We note that a unitary character of G is a cocycle of (K; G; H). Let $\widehat{\zeta}$ denote a character which extends ζ to G.

Proposition 5.2.9. The representations $\pi^{(a,\zeta)}(a\in\mathbb{Z}(K;G;H))$ and $\zeta\in\widehat{H}$ of A have the following properties.

(i) $\pi^{(a,\zeta)}$ is unitarily equivalent to $\pi^{(a+\zeta,0)}$.

(ii) $\pi^{(a,0)}$ is unitarily equivalent to $\pi^{(a',0)}$ if and only if a is cohomologous to a'.

(iii) If a cocycle a is not weakly cohomologous to a cocycle a', $\pi^{(a,\zeta)}$ is never unitarily equivalent to $\pi^{(a',\zeta')}$ for arbitrary choice of ζ , $\zeta' \in \hat{H}$.

(iv) $\pi^{(a,\zeta)}$ is irreducible if and only if H + K is dense in G.

Proof. The proof goes on by a modification of the techniques described in section 3.2. So we omit the detail. [Q.E.D.]

We see that the representation π^0 of A has been decomposed into a direct integral of $\pi^{(a,\zeta)}$ in as many ways as the cardinal number of the weak cohomology group $\widetilde{H}(K; G; H)$. We note that this fact does not depend on whether the repre-

- 109 -

sentation π^0 is type I or not. When π^0 is a factor representation of A, all $\pi^{(a, \zeta)}$ are irreducible representations of A (see Corollary 5.1.4).

5.3. Further discussions on the decomposition of π^0 We denote by Rep G the set of all unitary representations of a locally compact group G and by Rep A the set of all non-degenerate *-representations of a C^{*}-algebra A.

Suppose G be a locally compact second countable abelian group. Let K be a closed subgroup of G and H' be a closed subgroup of the dual \widehat{G} of G.

Definition 5.3.1 If a unitary representation U of K on $f_{\rm A}$ and a unitary representation V of H on the same space are so related that the Heisenberg commutation relation

$$U_k V_\omega = \langle k, \omega \rangle V_\omega U_k$$

holds for each kéK and wéH', we call the pair (V, U) a Heisenberg representation of (H', K). We denote by H-Rep(H', K) the set of all Heisenberg representations of (H', K). For two Heisenberg representations (V¹, U¹) and (V², U²) of (H', K), we say that (V¹, U¹) is unitarily equivalent to (V², U²) if there exists a unitary operator W from the representation space of (V¹, U¹) to the space of (V², U²) such that V² = W V¹ W^{*} and U² = W U¹ W^{*}.

Take and fix two closed subgroups H and K of G. We construct the C^{*}-dynamical system (C₀(G/H), K, γ) and the C^{*}-algebra A = C₀(G/H)×_vK, as before.

- 110 -

Let H^{\perp} be the annihilator of H in \widehat{G} and $C^{*}(H^{\perp})$ be the group C^{*} -algebra of H^{\perp} . Then, it is well-known that there is a bijective correspondence $V \longrightarrow \widetilde{V}$ from Rep H^{\perp} to Rep $C^{*}(H^{\perp})$ such that

$$\widetilde{V}(f) = \int_{H^{\perp}} f(\omega) V(\omega) d\nu(\omega)$$

for $f \in L^{1}(H^{\perp}, v)$ where v is a Haar measure of H^{\perp} [10]. Since we can regard the dual of H^{\perp} as G/H by Pontryagin's duality, we see that $C^{*}(H^{\perp}) \geq C_{0}(G/H)$ by the Gelfand transformation. Hence we get a bijection ϕ ; $V \rightarrow \phi(V)$ from Rep H^{\perp} to Rep $C_{0}(G/H)$. For a Heisenberg representation (V, U) of $(H^{\perp}, K), (\phi(V), U)$ becomes a covariant representation of $(C_{0}(G/H), K, \gamma)$. It is clear that the correspondence (V, U) $(\phi(V), U)$ is bijective from H-Rep (H^{\perp}, K) to C-Rep $(C_{0}(G/H), K, \gamma)$ of all covariant representations of $(C_{0}(G/H), K, \gamma)$. Further, we know that there exists a canonical correspondence between C-Rep $(C_{0}(G/H), K, \gamma)$ and Rep A [40]. Then, we get the following lemma.

Lemma 5.3.2. There is a bijective correspondence $(V, U) \longrightarrow \pi^{(V,U)}$ from H-Rep(H[⊥], K) to Rep A such that $(\phi(V),U)$ is the covariant representation of $(C_0(G/H), K, \gamma)$ which defines $\pi^{(V,U)}$. The correspondence $(V, U) \longrightarrow \pi^{(V,U)}$ is a one to one map from the set of equivalence classes of H-Rep(H[⊥],K) onto the set of equivalence classes of Rep A.

We shall say that a Heisenberg representation (V, U) of (H^{\perp}, K) is associated with a representation π of A if π ^(V,U)

- 111 -

is unitarily equivalent to π .

Example 5.3.3.

(i) A Heisenberg representation (V^0, U^0) of (H^{\perp}, K) associated with the representation π^0 of A described in section 5.1 and 5.2 is given as follows.

For $\xi(g) \in L^2(G)$,

 $(\nabla^{0} \xi)(g) = \langle g, \omega \rangle \xi(g)$ for each $\omega \in H^{\perp}$

and

$$(U_{k}^{0}\xi)(g) = \xi(g + k)$$
 for each $k \in K$.

(ii) The following Heisenberg representation (\hat{v}^0, \hat{u}^0) of (H^{\perp}, K) is also associated with the same representation π^0 . For $n(\chi) \in L^2(\hat{G})$,

 $(\hat{V}_{\omega}^{0}, \eta)(\chi) = \eta(\chi + \omega)$ for each $\omega \in H^{4}$

and

$$(\hat{U}_{k}^{0} \eta)(\chi) = \langle k, \chi \rangle \eta(\chi)$$
 for each $k \in K$.

In section 5.1, we found abelian subalgebras \mathfrak{A}^{a} (a $\in \mathbb{Z}(K; G; H)$) and \mathfrak{B}^{b} (b $\in \mathbb{Z}(H^{\perp}; G; K^{\perp})$) in $\pi^{0}(A)$ '. In section 5.2, we got the decomposition of π^{0} corresponding to \mathfrak{A}^{a} as

$$\pi^{0} \cong \int_{\widehat{H}}^{\bigoplus} (a, \zeta) d\mu(\zeta).$$

 $\pi^{(a,\zeta)}$ in here was defined as a covariant representation

$$(\rho^{(a,\zeta)}, {}^{0}U^{(a,\zeta)}) = \text{Ind}_{H}^{K}(a; \iota, \zeta).$$

Now we will give an explicit form for this component representation using Heisenberg representations. Let $\tilde{\zeta}$ be a character of G which is an extension of ζ . The canonical action of K on the space G/H is denoted by G/H $\ni x \longrightarrow k \cdot x \epsilon$ G/H for each k \in K. For a cocycle a(g) of (K; G; H), we have a cocycle c^a(k, x) of (K; G/H) by

$$c^{a}(k, g) = a(g)\overline{a(g + k)}$$

for every $k \in K$ and $g \in G$.

Proposition 5.3.4. The Heisenberg representation $(V^{(a,\zeta)}, U^{(a,\zeta)})$ of (H^{\perp}, K) associated with $\pi^{(a,\zeta)}$ for $a \in Z(K; G; H)$ and $\zeta \in H$ is given as follows.

For $\xi(x) \in L^2(G/H)$,

 $(V_{\omega}^{(a,\zeta)}\xi)(x) = \langle x, \omega \rangle \xi(x)$ for each $\omega \in \mathbb{H}^{\perp}$

and

$$(U_k^{(a,\zeta)}\xi)(x) = \langle k, \zeta \rangle c^a(k, x)\xi(k \cdot x)$$
 for each $k \in K$.

Proof. Let $g^{(a,\zeta)}$ be the representation space of the covariant representation $(\rho^{(a,\zeta)}, {}^{0}{}_{U}^{(a,\zeta)})$ which defines $\pi^{(a,\zeta)}$. Then, for each $\xi \in g^{(a,\zeta)}$, $\tilde{\zeta}\xi$ becomes H-invariant and square summable on G/H. Therfore, the correspondence $\xi \longrightarrow \tilde{\zeta}\xi$ defines a unitary operator W from $g^{(a,\zeta)}$ onto $L^{2}(G/H)$. By simple calculations it is easy to check that

$$W^{\rho}(a,\zeta)W^{*} = \phi(V^{(a,\zeta)})$$

and

- 113 -

$$W^{0}U^{(a,\zeta)}W^{*} = U^{(a,\zeta)}.$$
 [Q.E.D]

Next, we will see the decomposition of π^0 corresponding to \mathcal{B}^b . Let $G/K \ni z \longrightarrow \tilde{z} \in G$ be a Borel cross section. We denote the canonical action of H^{\perp} on the space \widehat{G}/K^{\perp} by $\widehat{G}/K^{\perp} \ni \sigma \longrightarrow \omega \cdot \sigma \in \widehat{G}/K^{\perp}$ for each $\omega \in H^{\perp}$. A cocycle $b(\chi)$ of $(H^{\perp}; \widehat{G}; K^{\perp})$ defines a cocycle $d^b(\omega, \sigma)$ of $(H^{\perp}; \widehat{G}/K^{\perp})$ by

$$d^{b}(\omega, \chi) = b(\chi)\overline{b(\chi + \omega)}$$

for all $\omega \in H^{\perp}$ and $\chi \in G$. Then, we get the following theorem.

Theorem 5.3.5. Corresponding to each of the abelian subalgebras \mathfrak{G}^{b} in $\pi^{0}(A)'$ (b $\in \mathbb{Z}(\operatorname{H}^{\perp}; \widehat{G}; \operatorname{K}^{\perp})$), the representation π^{0} of A is decomposed as follows.

$$\pi^{0} \cong \int_{G/K}^{\bigoplus} \hat{\pi}^{(b,z)} d\nu(z)$$

and

where v is a Haar measure of the factor group G/K and the Heisenberg representations $(\hat{V}^{(b,z)}, \hat{U}^{(b,z)})$ of (H^{\perp}, K) associated with the above $\hat{\pi}^{(b,z)}$ ($z \in G/K$) are given for $\eta(\sigma) \in L^2(\hat{G}/K^{\perp})$ by

 $(\hat{\mathbb{V}}_{\boldsymbol{\omega}}^{(b,z)},\eta)(\sigma) = \langle \hat{z}, \omega \rangle d^{b}(\omega,\sigma)\eta(\omega\cdot\sigma)$ for each $\omega \in \mathbb{H}^{\perp}$

$$(U_k^{(b,z)}, \eta)(\sigma) = \langle k, \sigma \rangle \eta(\sigma)$$
 for each $k \in K$

Proof. Note that the Heisenberg representation (V^0, U^0) associated with π^0 in (i) of Example 5.3.3 was decomposed into a direct integral of the Heisenberg representations as was shown in Proposition 5.3.4 corresponding to π^a . If we

- 114 -

go over to the dual \hat{G} of G and take the Heisenberg representation (\hat{V}^0 , \hat{U}^0) in (ii) of Example 5.3.3 instead of (V^0 , U^0), the desired assertions are immediately obtained. [Q.E.D.]

Now, we shall describe the properties of $\hat{\pi}^{(b,z)}$. We note that we can regard an element of G as a cocycle of $(H^{\perp}; \hat{G}; K^{\perp})$ by the fact that $\hat{G} \cong G$ as locally compact groups.

Proposition 5.3.6. The representations $\hat{\pi}^{(b,z)}$ of A ($b \in Z(H^{1}; \hat{G}; K^{1})$ and $z \in G/K$) have the following properties. (i) $\hat{\pi}^{(b,z)} \cong \hat{\pi}^{(b+\widetilde{z},0)}$, where $G/K \ni z \longrightarrow \widetilde{z} \in G$ is a Borel cross section.

(ii) $\hat{\pi}^{(b,0)}$ is unitarily equivalent to $\hat{\pi}^{(b',0)}$ if and only if b is cohomologous to b'.

(iii) Suppose a cocycle b is not weakly cohomologous to a cocycle b'. Then, $\pi^{(b,z)}$ is never unitarily equivalent to $\hat{\pi}^{(b',z')}$ for arbitrary choice of z and $z' \in G/K$.

(iv) $\pi^{(b,z)}$ is irreducible if and only if $K \cap H = \{0\}$.

Proof. Take the Heisenberg representations $(\hat{V}^{(b,z)}, \hat{U}^{(b,z)})$ associated with $\hat{\pi}^{(b, z)}$ and compare them with $(V^{(a,\zeta)}, U^{(a,\zeta)})$ in Proposition 5.3.4. Then, we see that the former becomes the latter if we exchange the role of \hat{G} and G. Hence, the above statements about $\hat{\pi}^{(b,z)}$ are immediate from the properties of $\pi^{(a,\zeta)}$ and the duality of G. [Q.E.D]

Thus, we see that Theorem 5.3.5 gives as many completely different decompositions of π^0 as the cardinal number of

- 115 -

 $\widetilde{H}(\operatorname{H}^{\perp}; \widehat{G}; \operatorname{K}^{\perp})$. However, it is difficult to find a general relation between the family of $\pi^{(a,\zeta)}$ and the family of $\widehat{\pi}^{(b,z)}$. In some concrete cases, we can show that almost all $\widehat{\pi}^{(b,z)}$ are not unitarily equivalent to any $\pi^{(a,\zeta)}$. This will be noticed in the later part of this chapter.

When π^0 is a non-type I factor representation of A, π^0 must be an injective type II factor representation. In this case, by Corollary 5.1.4, H + K is not equal to G but is dense in G and H \wedge K = {0}. Therefore, all $\pi^{(a,\zeta)}$ and $\hat{\pi}^{(b,z)}$ are irreducible representations. Moreover, it will be seen easily that $\pi^{(0,\zeta)}$ (resp. $\hat{\pi}^{(0,z)}$) are never unitarily equivalent to any $\hat{\pi}^{(b,z)}$ (resp. $\pi^{(a,\zeta)}$).

When π^0 is a type I factor representation, all $\pi^{(a,\zeta)}$ and $\hat{\pi}^{(b,z)}$ are still irreducible representations but in this case H(K; G; H) = {0} and H(H[⊥]; G; K[⊥]) = {0} as we have seen in chapter II. Therefore, it is no trouble to verify that all component representations are unitarily equivalent and that the decompositions of π^0 as given in Theorem 5.2.8 and Theorem 5.3.5 become all same.

When π^0 is a type I but not factor representation, H + K is not dense or H \cap K = {0}. In the first case, none of $\pi^{(a,\zeta)}$ are irreducible representations and in the second case, none of $\hat{\pi}^{(b,z)}$ are irreducible. $\hat{H}(K;G;H)$ and $\hat{H}(H^{\perp}; \hat{G}; K^{\perp})$ may or may not be trivial.

Remark 5.3.7. When a cocycle b of $(H^{\perp}; \hat{G}; K^{\perp})$ is the trivial one, the assertion of Theorem 5.3.5 coincides with the result obtained by applying the subgroup theorem

- 116 -

(Proposition 5.2.6) to the covariant representation which defines $\pi^0\,.$

5.4. Applications and examples

First we argue the decomposition of representations of the pair of groups in Example 5.1.5.

Example 5.4.1. The case of G = R, $H = {}^{\theta}Z$, and K = Z. The C^{*}-crossed product $A = C(R/\theta Z) \times_{\gamma} Z$ is the irrational rotation algebra. Let χ^{u} ($u \in R$) be unitary characters of R defined by $\chi^{u}(t) = e^{2\pi i u t}$ for each $t \in R$. Then, the dual group $\hat{G} = \{\chi^{u}; u \in R\}$ can be identified with R as a locally compact abelian group. Through this identification we can regard the annihilator K¹ of K as Z and the annihilator H¹ of H as $(1/\theta)Z$.

For p, $q \in Z$, put

 $a^{p}(t) = e^{(i\pi p/\theta)t^{2}}$ for each $t \in R = G$

and

$$b^{q}(u) = e^{i\pi\theta q u^{2}}$$
 for each $u \in \mathbb{R} = \hat{G}$.

Then, it is not difficult to check that $a^p \in Z(K; G; H)$ and $b^q \in Z(H^+; \hat{G}; K^+)$ for every p and $q \in Z$. Further, by similar techniques as in chapter II, we see that $a^p(\text{resp. } b^q)$ is weakly cohomologous to $a^{p'}$ (resp. $b^{q'}$) if and only if p = p' (resp. q = q').

Let π^0 be the representation of A on $L^2(\mathbb{R})$ as defined in section 5.1. Then, π^0 is decomposed as follows.

- 117 -

$$\pi^{0} \cong \int_{0}^{1/\theta} \pi^{(p,u)} du \qquad (p \in \mathbf{Z}) \qquad (\mathbf{I})$$
$$\cong \int_{0}^{1} \hat{\pi}^{(q,s)} ds \qquad (q \in \mathbf{Z}) \qquad (\mathbf{II})$$

(I) The Heisenberg representations $(V^{(p,u)}, U^{(p,u)})$ of $(H^{\perp}, K) = ((1/\theta)\mathbf{Z}, \mathbf{Z})$ associated with $\pi^{(p,u)}$ are given by, for $\xi(x) \in L^2([0, \theta))$,

$$(V_{\omega}^{(p,u)}\xi)(x) = e^{2\pi i \omega x}\xi(x)$$
 for each $\omega \in (1/\theta)Z$

and

$$(U_{m}^{(p,u)}\xi)(x) = e^{2\pi i u m} e^{(2\pi i p/\theta)mx} e^{(\pi i p/\theta)m^{2}}\xi(\overline{x+m})$$

for each $m \in \mathbb{Z}$, where $x = [x] + \overline{x}([x] \in \Theta \mathbb{Z}, \overline{x} [0, \Theta])$.

(II) The Heisenberg representations of $(\hat{V}^{(q, s)}, \hat{U}^{(q, s)})$ of $(H^{\perp}, K) = ((1/\theta)\mathbf{Z}, \mathbf{Z})$ associated with $\hat{\pi}^{(q, s)}$ are given by, for $n(y) \in L^2([0, 1))$,

$$(\hat{v}_{\omega}^{(q,s)}_{n})(y) = e^{2\pi i s \omega} e^{2\pi i \theta q \omega y} e^{\pi i \theta q \omega^{2}} \eta(\overline{y + \omega})$$

for each $\omega \in (1/\theta)\mathbf{Z}$, and

$$(\bigcup_{m}^{(q,s)} \eta)(y) = e^{2\pi i m y} \eta(y)$$
 for each $m \in \mathbb{Z}$.

By simple calculations, we can see the following facts. Let χ^{V} ($v \in [0, \theta)$) be unitary characters of $(1/\theta)\mathbf{Z}$ given by $\langle \omega, \chi^{V} \rangle = e^{2\pi i v \omega}$ for $\omega \in (1/\theta)\mathbf{Z}$. Then,

(a)
$$V^{(p,u)} \cong \int_{0}^{\theta} \chi^{v} dv.$$

(b) $\hat{V}^{(0,s)} \cong \sum_{n \in \mathbb{Z}}^{\gamma \oplus} \chi^{\overline{s+n}}.$
(c) $\hat{V}^{(q,s)} \cong |q| \int_{0}^{\theta} \chi^{v} dv$ if $q \neq 0$,

|q| appears here as the multiplicity of the representation.

Therefore, we can conclude that $\hat{\pi}^{(q,s)}$ is never unitarily equivalent to $\pi^{(p,u)}$ for every $p \in \mathbb{Z}$, $u \in [0,1/\theta)$ and $s \in [0, 1)$ if $q \neq \pm 1$. For q = 1 or -1, we know that $\hat{\pi}^{(q,s)}$ is unitarily equivalent to $\pi^{(a,u)}$ for some $a \in \mathbb{Z}(H;G;K)$ and $u \in [0,1/\theta)$. Note that π^0 is an injective type II₁ factor representation of A and all $\pi^{(p,u)}$ and $\hat{\pi}^{(q,s)}$ are irreducible representations of A.

Example 5.4.2. The case of $G = R^2$, $H = Z^2$, and $K = \{(x,y)\in R^2; y=0x\}$.

In this case, $A = C_0(G/H) \times_{\gamma} K = C(T^2) \times_{\delta} R$ where the action δ of R is defined by

$$S(t) \cdot (x, y) = (\overline{x + t}, \overline{y + \theta t})$$

for teR and $(x, y) \in [0, 1) \times [0, 1) \cong \mathbf{T}^2$.

Let $\chi^{(u,v)}$ (u, $v \in \mathbb{R}$) be unitary characters of \mathbb{R}^2 given by

$$\langle (x, y), \chi^{(u,v)} \rangle = e^{2\pi i (ux+vy)}$$

for $(x, y) \in \mathbb{R}^2$. Then, $\widehat{G} = \{\chi^{(u,v)}; u, v \in \mathbb{R}\}$ can be indentified with \mathbb{R}^2 as a locally compact group. Through this identification,

$$\mathbf{K}^{\perp} = \{(\mathbf{u}, \mathbf{v}) \in \mathbf{R}^{2}; \mathbf{v} = -(1/\theta)\mathbf{u}\} \cong \mathbf{R}$$
$$\mathbf{H}^{\perp} = \{(\mathbf{u}, \mathbf{v}) \in \mathbf{R}^{2}; \mathbf{u}, \mathbf{v} \in \mathbf{Z}\} \cong \mathbf{Z}^{2}.$$

For p and q $\in \mathbb{Z}$, put

- 119 -

$$a^{p}((x, y)) = e^{-\pi i p(\theta y - \overline{x})^{2}}$$

for $(x, y) \in \mathbb{R}^2$ where $0 \leq \overline{x} < 1$, and

$$b^{q}((x, y)) = e^{-\pi i q(\theta y + \overline{x})^{2}}$$

for $(x, y) \in \mathbb{R}^2$. Then, it is no trouble to verify that $a^p \in \mathbb{Z}(K; G; H)$ and $b^q \in \mathbb{Z}(H^1: \hat{G}; K^1)$ for every p and $q \in \mathbb{Z}$.

Let π^0 be the representation of A as defined in section 5.1. Then, π^0 is decomposed as follows.

$$\pi^{0} \cong \int_{0}^{1} \int_{0}^{1} \oplus \pi^{(p,u,v)} du dv \qquad (I)$$
$$\cong \int_{-\infty}^{\infty} \bigwedge_{\pi}^{(q,s)} ds \qquad (II)$$

The diagonal algebra of the decomposition (I) is π^{a^p} and the diagonal algebra of (II) is β^{b^q} .

(I) The Heisenberg representation $(V^{(p,u,v)}, U^{(p,u,v)})$ of $(H^{\perp}, K) = (\mathbb{Z}^2, \mathbb{R})$ associated with $\pi^{(p,u,v)}$ is given as follows.

For
$$\xi(x, y) \in L^2([0, 1) \times [0, 1))$$
,

$$(V_{(m,n)}^{(p,u,v)}\xi)(x, y) = e^{2\pi i\theta m x} e^{2\pi i n y} \xi(x, y)$$

for $(m, n) \in \mathbb{Z}^2$, and

$$(U_{t}^{(p,u,v)}\xi)(x,y) = e^{2\pi i ut} e^{2\pi i v\theta t} e^{ip\theta \{2[x+t](y-\theta t)+\theta[x+t]^{2}\}}$$
$$\xi(\overline{x+t}, \overline{y+\theta t})$$

for t ϵ R where x + t = $\overline{x+t}$ + [x+t], $0 \le \overline{x+t} < 1$ and [x+t] ϵZ .

(II) The Heisenberg representation $(\widehat{V}^{(q,s)}, \widehat{U}^{(q,s)})$ of $(H^{\perp}, K) = (\mathbb{Z}^{2}, \mathbb{R})$ associated with $\widehat{\pi}^{(q,s)}$ is given as follows. For $\eta(z) \in L^{2}(\mathbb{R})$,

$$(\widehat{\mathbf{V}}_{(m,n)}^{(q,s)}\eta)(z) = e^{2\pi i m s} e^{\pi i \theta q (2nz+n^2)} \eta(z+n+(1/\theta)m)$$

for (m, n) $\in \mathbb{Z}^2$ and

$$(\hat{U}_t^{(q,s)}\eta)(z) = e^{2\pi i z t} \eta(z)$$

for $t \in R$.

Take the representation $U^{(p,u)}$ (p ϵZ and $u \epsilon [0,1/\theta)$) of Z in Example 5.4.1. Then we get

$$\mathbb{U}^{(p,u,v)} \cong \operatorname{Ind}_{\mathbf{Z}}^{\mathbb{R}} \mathbb{U}^{(p,\overline{u+(1/_{\theta})v})}$$

where $0 \leq \overline{u+(1/\theta)v} < 1/\theta$. Hence, we see the following facts. Let χ^{u} ($u \in \mathbb{R}$) be unitary characters of \mathbb{R} given by $\chi^{u}(t)$ = $e^{2\pi i n t}$ for $t \in \mathbb{R}$.

(a)
$$\widehat{U}^{(q,s)} \cong \int_{-\infty}^{\infty} \chi^{u} du$$

(b) $U^{(0,u,v)} \cong \sum_{m,n\in\mathbb{Z}}^{\gamma \oplus} \chi^{u+n+\theta(v+m)}$
(c) $U^{(p,u,v)} \cong |p| \int_{-\infty}^{\infty} \chi^{u} du$ if $p \neq 0$

where |p|appears as the multiplicity of the representation.

Therefore, we get the fact that for each u, $v \in [0, 1)$, $\pi^{(p,u,v)}$ is never unitarily equivalent to $\hat{\pi}^{(q,s)}$ for arbitrary choice of $q \in \mathbf{Z}$ and $s \in \mathbb{R}$ if $p \neq \pm 1$.

We note that the representation π^0 of A is an injective type II_{∞} factor representation and all $\pi^{(p,u,v)}$ and $\hat{\pi}^{(q,s)}$ are irreducible.

Now, we show an application of our results to unitary representations of certain locally compact groups.

Let $\widetilde{G} = N \times_{s} \widetilde{K}$ be a regular semi-direct product of N

with \tilde{K} where N and \tilde{K} are locally compact second countable abelian groups and \tilde{K} acts on N as an automorphism group. The action of \tilde{K} on N induces the action of \tilde{K} on \hat{N} (the dual of N) as a topological transformation group and the regularity of \tilde{G} means that the topological transformation group (\tilde{K} ; \hat{N}) is smooth ([12], [16], [32]). Taking a closed subgroup K of \tilde{K} , we shall consider the closed subgroup G = N ×_SK of \tilde{G} = N×_S \tilde{K} . For a unitary character χ of N, we get a unitary representation π^{χ} of G given by

$$\pi^{\chi} = \frac{1}{G} \left| \operatorname{Ind}_{N}^{\widetilde{G}} \chi \right|.$$

We denote by \mathfrak{m}^{χ} the representation of the group C*-algebra C^{*}(G) of G corresponding to the above representation π^{χ} of G [40]. Let H_{χ} denote the stability group of K at $\chi \in \widehat{N}$. Then, we have a C^{*}-crossed product $A^{\chi} = C_0(\widetilde{K}/H_{\chi})X_{\gamma}K$ where γ is the canonical action of K on $C_0(\widetilde{K}/H_{\chi})$. Let π^0 be the representation of A^{χ} as defined in section 5.1. Then, we get the following proposition.

Proposition 5.4.3. There exists a homomorphism ψ^{χ} from C^{*}(G) onto A^X such that Ker ψ^{χ} = Ker $\tilde{\pi}^{\chi}$ and $\tilde{\pi}^{\chi}$ = $\pi^{0}\circ\psi^{\chi}$. This ψ^{χ} induces a natural correspondence Ψ^{χ} from Rep A^X to Rep G which has the following properties.

(i) For each $\pi \in \operatorname{Rep} A^{X}$, $\pi(A^{X})'' = \Psi^{X}(\pi)(G)''$

(ii) For π , $\pi' \in \operatorname{Rep} A^{X}$, $\pi \cong \pi'$ if and only if $\Psi^{X}(\pi) \cong \Psi^{X}(\pi')$.

(iii) Suppose $\pi^{0} \cong \int_{Z}^{\bigoplus} \pi^{\zeta} d\mu(\zeta)$.

- 122 -

Then, we get $\pi^{\chi} \cong \int_{-\pi}^{\oplus} \Psi^{\chi}(\pi^{\zeta}) d\mu(\zeta).$

Proof. It is not difficult to see that $C^*(G)/\operatorname{Ker} \tilde{\pi}^X$ is isomorphic with the C^* -crossed product $C_0(\operatorname{Orb}_{\widetilde{K}}(X)) \times_{\delta} K$ where δ is the action of \widetilde{K} on N and $\operatorname{Orb}_{\widetilde{K}}(X) = \{\delta(t) \cdot X; t \in \widetilde{K}\}$. Put $\phi^X(t) = \delta(t) \cdot X$ for every $t \in \widetilde{K}$. Then, ϕ^X induces a map $\widetilde{\phi}^X$ from $\widetilde{K}/\operatorname{H}_X$ to $\operatorname{Orb}_{\widetilde{K}}(X)$ which is a homeomorphism due to the regularity of \widetilde{G} [12]. Through this $\widetilde{\phi}^X$, the action δ of K is transformed into the action γ of K and $C_0(\operatorname{Orb}_{\widetilde{K}}(X)) \times_{\delta} K$ is isomorphically transformed to $C_0(\widetilde{K}/\operatorname{H}_X) \times_{\gamma} K = A^X$. Then, we get a homomorphism ψ^X from $C^*(G)$ onto A^X such that $\operatorname{Ker} \psi^X$ = Ker $\widetilde{\pi}^X$ by composing these maps.

Since there exists a canonical correspondence between Rep G and Rep C^{*}(G), the correspondence $\pi \to \pi \circ \psi^{\chi}$ from Rep A^{χ} to Rep C^{*}(G) induces the correspondence ψ^{χ} from Rep A^{χ} into Rep G.

Other properties follow immediately from the above definitions of ψ^{χ} and ψ^{χ} . We omit the detail. [Q.E.D.]

The following corollary is easily obtained.

Corollary 5.4.4. The unitary representation π^{χ} of G have the following properties.

(i) $\pi^{\chi}(G)'' = \mathfrak{M}(\tilde{K}/H_{\chi}, K)$ and $\pi^{\chi}(G)' = \mathfrak{M}(\tilde{K}/K, H_{\chi}).$

(ii) There exist abelian von Neumann subalgebras σ^{a} and σ^{b} in $\pi^{\chi}(G)'$ parametrized with $a \in Z(K; \tilde{K}; H_{\chi})$ and $b \in Z(H_{\chi}^{\perp}; \tilde{K}; K^{\perp})$.

(iii) Corresponding to σ^a , π^{χ} is decomposed as follows.

- 123 -

$$\pi^{\chi} \cong \int_{\widehat{H}_{\chi}}^{\bigoplus} \Psi^{\chi}(\pi^{(a,\zeta)}) d\mu(\zeta).$$

If K + H_X is dense in \tilde{K} , \mathfrak{A}^{a} is maximal abelian in $\pi^{\chi}(G)'$ and $\Psi^{\chi}(\pi^{(a,\zeta)})$ are all irreducible representations of G. (iv) Corresponding to \mathfrak{B}^{b} , π^{χ} is decomposed as follows.

$$\pi^{\chi} \cong \int_{\widetilde{K}/K}^{\bigoplus} \Psi^{\chi}(\widehat{\pi}^{(b,z)}) d\nu(z).$$

If $K \cap H_{\chi} = \{0\}, \mathfrak{G}^{b}$ is maximal abelian in $\pi^{\chi}(\mathfrak{G})'$ and $\Psi^{\chi}(\hat{\pi}^{(b,z)})$ are all irreducible representations of G.

We note that the assertion (iii) coincides with the result described in chapter III and that the decomposition in the case of b = 0 at (iv) coincides with the result obtained from the subgroup theorem of induced representations by G.W. Mackey [30]. Corollary 5.4.4 shows that there are other possibilities of decompositions of the representation π^{χ} of G.

Example 5.4.5. The discrete Mautner group

Let \tilde{G} be the 3-dimensional solvable Lie group given as a semi-direct product $C \times_{s}^{R}$ of the additive group C of all complex numbers with R with the multiplication;

 $(z, t)(z', t') = (z + e^{it}z', t + t')$

for z, z' \in C and t, t' \in R. Take the closed subgroup G = C \times_{s} Z of $\tilde{G} = C \times_{s}$ R. This group G is the discrete Mautner group.

Let χ^{ζ} ($\zeta \in \mathbb{C}$) be unitary characters of \mathbb{C} given by $\chi^{\zeta}(z) = e^{i(z,\zeta)}$, where (,) means the real inner product

of C. For $r \in \mathbb{R}^+$, put

$$\pi^{\mathbf{r}} = \frac{1}{G} | \operatorname{Ind}_{\mathbf{C}}^{\widetilde{\mathbf{G}}} \chi^{\mathbf{r}}.$$

Then, π^r is an injective type II₁ factor representation of G for each r $\in R^+$.

By Example 5.4.1 and Corollary 5.4.4, we get the following decompositions of π^{r} .

$$\pi^{\mathbf{r}} \cong \int_{0}^{1/2} \mathfrak{m}^{(\mathbf{r},\mathbf{p},\mathbf{u})} d\mathbf{u} \quad (\mathbf{p} \in \mathbf{Z}) \tag{I}$$
$$\cong \int_{0}^{1} \mathfrak{m}^{(\mathbf{r},\mathbf{q},\mathbf{s})} d\mathbf{s} \quad (\mathbf{q} \in \mathbf{Z}) \tag{II}$$

Observe that the $\pi^{(r,p,u)}$ and $\pi^{(r,q,s)}$ are irreducible representations of G. We have moreover the following properties.

(i) If $p \neq p'$ (resp. $q \neq q'$), each $\pi^{(r,p,u)}$ (resp. $\hat{\pi}^{(r,q,s)}$) is never unitarily equivalent to $\pi^{(r,p',s)}$ (resp. $\hat{\pi}^{(r,q',s')}$) for arbitrary choice of $u' \in [0,1/2\pi)$ (resp. $s' \in [0,1)$).

(ii) (a)
$$\mathbf{c} | \pi^{(\mathbf{r},\mathbf{p},\mathbf{u})} \cong \int_{|\zeta|=\mathbf{r}}^{\bigoplus} \chi^{\zeta} d\zeta.$$

(b) $\mathbf{c} | \hat{\pi}^{(\mathbf{r},0,s)} \cong \sum_{\mathbf{n}\in\mathbf{Z}}^{\bigoplus} \chi^{\mathbf{r}e^{\mathbf{i}(\mathbf{n}+2\pi s)}}.$
(c) $\mathbf{c} | \hat{\pi}^{(\mathbf{r},q,s)} \cong |q| \int_{|\zeta|=\mathbf{r}}^{\bigoplus} \chi^{\zeta} d\zeta$ if $q \neq 0$.

Hence we see that every $\hat{\pi}^{(r,q,s)}$ except the case of $q = \pm 1$ is never unitarily equivalent to each $\pi^{(r,p,u)}$.

We note that the decomposition (II) of π^r is a newly obtained one. It was not described in chapter III and it

- 125 -

shows that G has the irreducible representations such as $\hat{\pi}^{(r,q,s)}$ with the property (c) of (ii). We couldn't give an explicit form of $\hat{\pi}^{(r,q,s)}$, but we know the existence of such irreducible representations.

Example 5.4.6. the discrete Heisenberg group

Next, let G be the discrete Heisenberg group, defined as a group of matrices :

$$G = \left\{ \begin{pmatrix} 1 & k & u \\ 0 & 1 & \ell \\ 0 & 0 & 1 \end{pmatrix}; k, \ell \in \mathbb{Z} \text{ and } u \in \mathbb{R} \right\}.$$

We take a larger group G given as

$$\widetilde{G} = \left\{ \begin{pmatrix} 1 \ t \ u \\ 0 \ 1 \ \ell \\ 0 \ 0 \ 1 \end{pmatrix}; \quad \ell \in \mathbb{Z} \text{ and } t, u \in \mathbb{R} \right\}$$

Then, G is a closed subgroup of \tilde{G} which is a type I group. We denote an element $\begin{pmatrix} 1 & t & u \\ 0 & 1 & \ell \\ 0 & 0 & 1 \end{pmatrix}$ of \tilde{G} as (ℓ , u, t).

Put

$$N = \{(\ell, u, 0) ; \ell \in \mathbb{Z} \text{ and } u \in \mathbb{R} \}.$$

Then, N is a normal subgroup of \widehat{G} . Let χ^{σ} ($\sigma \in \mathbb{R}$) be a unitary character of N given by

$$\chi^{\sigma}((\ell,u,0)) = e^{i\sigma u}$$
.

We define a unitary representation π^{σ} of G by

$$\pi^{\sigma} = \int_{G} \left| \text{Ind}_{N}^{\widetilde{G}} \chi^{\sigma} \right|.$$

Then, by Corollary 5.4.4, we get the following decompositions of π^{σ} .

$$\pi^{\sigma} \cong \int_{0}^{\sigma} U^{(\sigma, p, \lambda)} d\lambda \qquad (p \in \mathbb{Z}) \quad (\mathbb{I})$$
$$\cong \int_{0}^{l} V^{(\sigma, q, r)} dr \qquad (q \in \mathbb{Z}) \quad (\mathbb{II})$$

(i) If $\sigma \in \mathbb{R} \setminus (1/2\pi) \mathbb{Q}$, π^{σ} is an injective type II₁ factor representation of G and $U^{(\sigma,p,\lambda)}$ and $V^{(\sigma,q,r)}$ are irreducible representations of G.

(ii) $U^{(\sigma,p,\lambda)}$ is written in an explicit form as follows. For $\xi(x) \in L^2([0,2\pi/\sigma))$,

 $(U_{(l,u,k)}^{(\sigma,p,\lambda)}\xi)(x) = e^{i\sigma u} e^{i\ell x} e^{i\lambda k} e^{i\sigma pkx} e^{i\sigma pk^2/2} \xi(\overline{x+k}),$

where $0 \leq \overline{x+k} < 2\pi/\sigma$. If $p \neq p'$, $U^{(\sigma,p,\lambda)}$ is never unitarily equivalent to $U^{(\sigma',p',\lambda')}$ for arbitrary choice of $\sigma' \in \mathbb{R}$, $\lambda' \in [0,\sigma)$.

(iii) $V^{\left(\sigma,q,r\right)}$ is also written in an explicit form as follows.

For $\zeta(y) \in L^2([0, 2\pi))$,

 $(\mathbb{V}_{(\ell,u,k)}^{(\sigma,q,r)}(y) = e^{i\sigma u} e^{i\gamma k} e^{ir\sigma} e^{iq \ell y} e^{i\sigma q \ell^2/2} \zeta(\overline{y+\sigma \ell}),$

where $0 \leq \overline{y+\sigma} \leq 2\pi$. If $q \neq q'$, $V^{(\sigma,q,r)}$ is never unitarily equivalent to $V^{(\sigma',q',r')}$ for arbitrary choice of $\sigma' \in \mathbb{R}$, $r' \in [0, 1)$.

(iv) If $p \neq \pm 1$ (resp. $q \neq \pm 1$), $U^{(\sigma,p,\lambda)}$ (resp. $V^{(\sigma,q,r)}$) is never unitarily equivalent to $V^{(\sigma',q',r')}$ (resp. $U^{(\sigma',p',\lambda')}$ for any $\sigma' \in \mathbb{R}$, $q' \in \mathbb{Z}$, and $r' \in [0, 1)$ (resp. $\sigma' \in \mathbb{R}$, $p' \in \mathbb{Z}$, and $\lambda' \in [0, \sigma)$).

- 127 -

For the detail, see [22].

Example 5.4.7. A subgroup of the discrete Heisenberg group

Let H denote the subgroup of the discrete Heisenberg group, defined by

$$H = \left\{ \begin{pmatrix} 1 & k & m \\ 0 & 1 & \varrho \\ 0 & 0 & 1 \end{pmatrix} ; k, \ell, m \in \mathbb{Z} \right\}.$$

Put

$$\begin{aligned} \mathbf{1}_{\pi}^{\sigma} &= \prod_{\mathbf{H}} | \pi^{\sigma} & (\sigma \in \mathbf{R}), \\ \mathbf{1}_{\mathbf{U}}^{(\sigma,\mathbf{p},\lambda)} &= \prod_{\mathbf{H}} | \mathbf{U}^{(\sigma,\mathbf{p},\lambda)} & (\mathbf{p} \in \mathbf{Z}, \lambda \in \mathbf{R}), \\ \mathbf{1}_{\mathbf{V}}^{(\sigma,\mathbf{q},\mathbf{r})} &= \prod_{\mathbf{H}} | \mathbf{V}^{(\sigma,\mathbf{q},\mathbf{r})} & (\mathbf{q} \in \mathbf{Z}, \mathbf{r} \in \mathbf{R}). \end{aligned}$$

Then, we get the following results about the unitary representations of H.

$$\overset{L}{\pi}^{\sigma} \cong \int_{0}^{\sigma} \overset{L}{U}^{(\sigma, p, \lambda)} d\lambda \quad \text{for each } p \in \mathbb{Z}$$
$$\overset{\cong}{=} \int_{0}^{1} \overset{L}{V}^{(\sigma, q, r)} dr \quad \text{for each } q \in \mathbb{Z}.$$

(i) If $\sigma \in \mathbb{R} \setminus (1/2\pi) \mathbb{Q}$, $^{1}\pi^{\sigma}$ is an injective type II₁ factor representation of H and $^{1}U^{(\sigma,p,\lambda)}$ and $^{1}V^{(\sigma,q,r)}$ are irreducible representations of H.

(ii) ${}^{l}U^{(\sigma,p,\lambda)} \cong {}^{l}U^{(\sigma',p',\lambda')}$ if and only if $\sigma' - \sigma \in 2\pi Z$, p' = p, and $\lambda' - \lambda \in 2\pi Z + \sigma Z$. ${}^{l}V^{(\sigma,q,r)} \cong {}^{l}V^{(\sigma',q',r')}$ if and only if $\sigma' - \sigma \in 2\pi Z$, q' = q, and $r' - r \in Z + (2\pi/\sigma)Z$. Moreover, if $q \neq \pm l$, each ${}^{l}V^{(\sigma,q,r)}$ is never unitarily equivalent to any of ${}^{l}U^{(\sigma,p,\lambda)}$. Example 5.4.8. The free group F_2 on two generators Let F_2 be the free group generated by two elements a and b. Then, there exists a homomorphism ψ from F_2 onto H such that

$$\psi(a) = \begin{pmatrix} 1 & 1 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{pmatrix}$$

$$\Psi(b) = \begin{pmatrix} 1 & 0 & 0 \\ 0 & 1 & 1 \\ 0 & 0 & 1 \end{pmatrix}$$

Put

and

$${}^{2}\pi^{\sigma} = \pi^{\sigma} \circ \psi \qquad (\sigma \in \mathbb{R})$$

$${}^{2}U^{(\sigma,p,\lambda)} = {}^{1}U^{(\sigma,p,\lambda)} \circ \psi \qquad (p \in \mathbb{Z}, \lambda \in \mathbb{R})$$

$${}^{2}V^{(\sigma,q,r)} = {}^{1}V^{(\sigma,q,r)} \circ \psi \qquad (q \in \mathbb{Z}, r \in \mathbb{R}).$$

Then, we can interpret the results in Example 5.4.7 as they concern the representations of $\rm F_{2}$

H. Yoshizawa [54] showed, the non-uniqueness of irreducible decompositions of a representation, and it was about the regular representation of F_2 (see section 1.2). Our results are entirely different from his case. We note that $2\pi^{\sigma}$ is an "injective" type II₁ factor representation of F_2 , while the regular representation of F_2 is not injective but type II₁ factor representation.

5.4.9. The Dixmier group D

Let D be a 7-dimensional simply connected solvable Lie

group with the multiplication

for u, $v \in C$ and x, y, $z \in R$. D is known to be a non-type I group among simply connected solvable Lie groupS as pointed out by J. Dixmier [8] and so we shall call D the Dixmier group. Using the Mackey's machine and the results concerning the discrete Heisenberg group G, we shall show that we obtain a large number of irreducible representations of D. Put

$$M = \{ (u, v, 0, 0, 0), u, v \in \mathbb{C} \}$$

and

$$H = \{(0, 0, x, y, z) ; x, y, z \in \mathbb{R}\}.$$

Then, M is an abelian group isomorphic with c^2 and H is the Heisenberg group. H acts on M by the inner automorphism and D can be regarded as M x_s H which is a regular semi-direct product of M with H.

Let $\omega^{(s,t)}$ (s, t $\in \mathbb{R}^+$) be unitary characters of M given by

$$\langle (u, v), \omega^{(s,t)} \rangle = e^{i(s,u)}e^{i(t,v)}$$

for $(u,v) \in \mathbb{C}^2 = \mathbb{N}$ where (,) means the real inner product of \mathbb{C} . Then, the stability group of H at $\omega^{(s,t)}$ in $\widehat{\mathbb{M}}$ (the dual of M) is just the discrete Heisenberg group G. Put $\mathbb{D}_0 = \mathbb{M} \times_s \mathbb{G}$ and define a representation ${}^0\mathbb{W}^{(s,t,L)}$ of \mathbb{D}_0 for $\omega^{(s,t)} \in \widehat{M}$ and some representation L of G by

$$O_{W}(s,t,L) = \omega(s,t)L_{g}$$

Then, we get a unitary representation $W^{(s,t,L)}$ of \mathbb{P} by

$$W^{(s,t,L)} = \operatorname{Ind}_{D_0}^{D_0} W^{(s,t,L)}$$

For s, $t \in \mathbb{R}^+$ and $\sigma \in \mathbb{R}$, Put

$$\begin{aligned} & 3_{\pi}(s,t,\sigma) = W^{(s,t,L)} & \text{When } L = \pi^{\sigma}, \\ & 3_{U}(s,t,\sigma,p,\lambda) = W^{(s,t,L)} & \text{When } L = U^{(\sigma,p,\lambda)}, \\ & 3_{V}(s,t,\sigma,q,r) = W^{(s,t,L)} & \text{When } L = V^{(\sigma,q,r)}, \end{aligned}$$

where $p \in \mathbb{Z}$, $\lambda \in \mathbb{R}$, $q \in \mathbb{Z}$, and $r \in \mathbb{R}$.

Then, we get the following results immediately.

$$\begin{split} & 3_{\pi}(\mathbf{s}, \mathbf{t}, \sigma) \\ & \cong \int_{0}^{\sigma} 3_{U}(\mathbf{s}, \mathbf{t}, \sigma, \mathbf{p}, \lambda) \, d\lambda \qquad \text{for each } \mathbf{p} \in \mathbb{Z} \\ & \cong \int_{0}^{1} 3_{V}(\mathbf{s}, \mathbf{t}, \sigma, \mathbf{q}, \mathbf{r}) \, d\mathbf{r} \qquad \text{for each } \mathbf{q} \in \mathbb{Z}. \end{split}$$

(i) If $\sigma \in \mathbb{R} \setminus (1/2\pi)\mathbb{Q}$, $3_{\pi}(s,t,\sigma)$ is an injective type II₆ factor representation of D and $3_{U}(s,t,\sigma,p,\lambda)$ and $3_{V}(s,t,\sigma,p,r)$ are irreducible representations of D.

(ii) ${}^{3}_{U}(s,t,\sigma,p,\lambda) \cong {}^{3}_{U}(s',t',\sigma',p',\lambda')$ if and only if $s'=s, t'=t,\sigma'=\sigma, p'=p$, and $\lambda'-\lambda \in 2\pi \mathbb{Z} + \sigma \mathbb{Z}$. ${}^{3}_{V}(s,t,\sigma,q,r) \cong$ ${}^{3}_{V}(s',t',\sigma',q',r')$ if and only if $s'=s, t'=t, \sigma'=\sigma, q'=q$, and $r'-r \in \mathbb{Z} + (2\pi/\sigma)\mathbb{Z}$. Moreover, if $q \neq \pm 1$, ${}^{3}_{V}(s,t,\sigma,q,r)$ is never unitarily equivalent to each ${}^{3}_{U}(s',t',\sigma',p,\lambda)$.

- 131 -

Here, we note that ${}^{3}U^{(s,t,\sigma,p,\lambda)}$ (s,t $\in \mathbb{R}^{+}, \sigma \in \mathbb{R} \setminus (1/2\pi)Q$, $p \in \mathbb{Z}$, $\lambda \in \mathbb{R}$) and $V^{(s,t,\sigma,q,r)}$ (s,t $\in \mathbb{R}^{+}, \sigma \in \mathbb{R} \setminus (1/2\pi)Q$, $q \in \mathbb{Z} \setminus \{0, \pm 1\}$, $r \in \mathbb{R}$) are new parametrized families of irreducible representations of the Dixmier group D so far as we know.

BIBRIOGRAPHY

- [1] W.B.Arveson; An invitation to C^{*}-algebras, Springer Verlag, (1976).
- [2] L.Auslander & C.C.Moore; Unitary representations of solvableLie groups, Mem. Amer. Math. Soc., 62, (1966).
- [3] S.C.Bagchi, J.Mathew, & N.G.Nadkarni; On systems of imprimitivity on locally compact abelian groups with dense action, Acta Math., 133, (1974), 287-304.
- [4] L.Baggett; Representations of the Mautner group I, Pacific J.Math., 45, (1978), 7-22.
- [5] I.D.Brown; Representations of finitely generated nilpotent groups, Pacific J. Math., 45, (1973), 13-26.
- [6] A.Connes; Classification of injective factors, Ann. Math., 104, (1976), 73-116.
- [7] J.Dixmier; Algèbre quasi-unitaires, Comm. Math. Helv., 26, (1952), 275-322.
- [8] J.Dixmier; Sur le revetement universel d'un groupe de Lie de type I, C. R. Acad. Sci. Paris, 252, (1961), 2805-2806.
- [9] J.Dixmier; Les algèbres d'operateurs dans l'espace hilbertien (Algèbres de von Neumann), Gauthier-villars, Paris, (1969).
- [10] J.Dixmier; Les C^{*}-algèbres et leurs représentations, Gauthier-Villars, Paris, (1968).
- [11] S.Doplicher, D.Kastler, & C.W.Robinson; Covariance algebras in field theory and statistical mechanics, Comm. Math. Phys., 3, (1966), 1-28.
- [12] E.G.Effros; Transformation groups and C^{*}-algebras, Ann. of Math., 81, (1965), 38-55.

- [13] J.Feldman & C.C.Moore; Ergodic equivalence relations, cohomology and von Neumann algegras I,II, Bull. AMS, 81, (1975), 921-924;
 Trans. AMS, 234, (1977), 289-361.
- [14] S.Funakoshi; On representations of non-type I groups, Tohoku Math. J., 31, (1979), 139-150.
- [15] J.Glimm; Type I C^{*}-algebras, Ann. of Math., 73, (1961), 572-612.
- [16] J.Glimm; Locally compact transformation groups, Trans. AMS, 101, (1961), 124-138.
- [17] A.Guichardet; Sur la cohomologie des groupes topologiques I,II,
 III, Bull. Sc. Math., 95, (1971), 161-175; 96, (1972), 305-332;
 98, (1974), 201-208.
- [18] P.Hahn; On regular representations of measure groupoids, Trans. AMS, 242, (1978), 35-72.
- [19] S.Kawakami; On representations of Heisenberg-type groups, unpublished, (1976).
- [20] S.Kawakami; Irreducible representations of certain semi-direct product groups, Math. Jap., 26, No.6, (1981), 667-693.
- [21] S.Kawakami; On decompositions of some factor representations, (to appear in Math. Jap., 27).
- [22] S.Kawakami; Representations of the discrete Heisenberg group, (to appear in Math. Jap. 27).
- [23] S.Kawakami; On decompositions of regular representation of F_2 , (in preparation).
- [24] S.Kawakami; On decompositions of regular representation of a certain semi-direct product group, (in preparation).
- [25] S.Kawakami; Disintegration of Polish groups, (in preparation).
- [26] S.Kawakami & T.Kajiwara; Representations of certain non-type I C*-crossed products, (to appear in Math. Jap.).

- [27] S.Kawakami & T.Kajiwara; Weak cohomology related with the variaty of decompositions of representations, (in preparation).
- [28] A.A.Kirillov; Elements of the theory of representations, Moscow, (1972).
- [29] G.W.Mackey; On induced representations of groups, Amer. J. Math., 73, (1951), 576-592.
- [30] G.W.Mackey; Induced representations of locally compact groups I,II, Ann. of Math., 55, (1952), 101-139; 58, (1953), 193-221.
- [31] G.W.Mackey; The theory of group representations, Lecture Notes, Chicago Univ., (1955).
- [32] G.W.Mackey; Borel structures in groups and their duals, Trans. AMS, 83, (1957), 134-165.
- [33] G.W.Mackey; Unitary representations of group extentions I, Acta Math., 99, (1958), 265-311.
- [34] G.W.Mackey; Ergodic theory and virtual groups, Math.Ann., 166, (1966), 187-207.
- [35] G.W.Mackey; Induced representations of groups and quantum mechanics, W.A. Benjamin, (1968).
- [36] F.I.Mautner; Unitary representations of locally compact groups I, II, Ann. of Math., 51, (1950), 1-25; 52, (1950), 528-556.
- [37] C.C.Moore; Extensions and low-dimensional cohomology theory of locally compact groups I,II,III,IV, Trans. AMS, 113, (1964), 40-63; 113, (1964), 64-86; 221, (1976), 1-33; 221, (1976), 35-38.
- [38] C.C.Moore and K.Schmidt; Coboundaries and homomorphism for nonsingular actions and a problem of H.Helson, Proc. London Math. Soc., 40, (1980), 443-475.
- [39] M.Nakamura & H.Umegaki; Heisenberg commutation relations and the plancherel theorem, Proc. Jap. Acad., 37, (1961), 239-242.

- 135 -

- [40] G.K.Pedersen; C^{*}-algebras and their automorphism groups, London Math. Soc. Mono., No.14, Academic Press, (1978).
- [41] L.Pukanszky; Unitary representations of solvable Lie groups, Ann. Sci. École Norm. Sup., 4, (1971), 457-608.
- [42] L.Pukanszky; Characters of connected Lie groups, Acta Math., 133, (1974), 81-137.
- [43] A.Ramsay; Virtual groups and group action, Adv. Math., 6, (1971), 253-322.
- [44] A.Ramsay; Nontransitive quasiorbits in Mackey's analysis of group extensions, Acta Math., 137, (1976), 17-48.
- [45] K.Schmidt; Lectures on cocycles of ergodic transformation groups, Warwick, (1976).
- [46] M.Saito; Représentation unitaires monomiales d'un groupe discret, en particulier du groupe modulaire, J. Math. Soc. Japan, Vol. 26, No.3, (1974), 464-482.
- [47] S.Sakai; C*-algebras and W*-algebras, Springer-Verlag, (1971).
- [48] O.Takenouchi; Families of unitary operators defined on groups, Math. J. Okayama Univ., 6, (1957), 171-179.
- [49] O.Takenouchi; Sur la facteur-représentation d'un groupe de Lie de type (E), Math. J. Okayama Univ., 7, (1957),151-161.
- [50] M.Takesaki; On some representations of C^{*}-algebras, Tohoku Math. J., 15, (1963), 79-95.
- [51] M.Takesaki; Covariant representations of C⁻-algebras and their locally compact automorphism groups, Acta Math., 119, (1967), 273-303.
- [52] M.Takesaki; A generalized commutation relation for the regular representation, Bull. Soc. Math. France, 97, (1969), 287-297.
- [53] M.Takesaki & N.Tatsuuma; Duality and Subgroups II, J. Functional Analysis, 11, (1972), 184-190.
- [54] H.Yoshizawa; Some remarks on unitary representations of free group, Osaka Math. J., Vol. 3, (1951), 55-63.
- [55] G.Zeller-Meier; Produits croisés d'une C*-algèbre par un groupe d'automorphisms, J. Math. Pures Appl., 47, (1968), 101-239.

Satoshi KAWAKAMI Dep. of Applied Math. Faculty of Engineering Science Osaka University Toyonaka, Osaka, 560, Japan